Printed at the Mathematical Centre, 49, 2e Boerhaavestraat, Amsterdam.

The Mathematical Centre, founded the 11-th of February 1946, is a nonprofit institution aiming at the promotion of pure mathematics and its applications. It is sponsored by the Netherlands Government through the Netherlands Organization for the Advancement of Pure Research (Z.W.O).

MATHEMATICAL CENTRE TRACTS 107

THE NUMERICAL SOLUTION OF NONLINEAR OPERATOR EQUATIONS BY IMBEDDING METHODS

C. DEN HEIJER

MATHEMATISCH CENTRUM AMSTERDAM 1979

AMS(MOS) subject classification scheme (1970): 65H10, 65J05 ISBN 90 6196 175 0

ACKNOWLEDGEMENTS

First of all I wish to thank Prof.dr. M.N. Spijker, who introduced me to the subject of this monograph. His interest and stimulating remarks were very much appreciated.

I am also grateful to Prof.dr. A. van der Sluis and Prof.dr.

W.W.E. Wetterling for their useful comments on an early draft of the paper. Furthermore, I wish to thank Mr. M. Bakker and Mr. F.J. Burger

for their programming assistance in the computational experiments.

Finally, I am indebted to the Mathematical Centre for the opportunity to publish this monograph in their series Mathematical Centre Tracts, and all those at the Mathematical Centre who have contributed to its technical realization. -

CONTENTS

	1
CHAPTER I. INTRODUCTION	1
1.1 Imbedding methods: a desirer	4
1.2 Impedating methods. Historical Salary	7
1.3 Scope of the study	
A A A A A A A A A A A A A A A A A A A	9
CHAPTER 2. PRELIMINARIES	9
2.1 Conventions and notations	11
2.2 The spectrum of a finear operator	13
2.3 Iterative methods	14
2.4 The radius of convergence	15
2.5 Local convergence behaviour of iterative processes	19
2.6 The imbedding and the differential equation	
	25
CHAPTER 3. RUNGE-KUTTA METHODS	25
3.1 One-step methods	26
3.2 Description of the Runge-Kutta methods	26
3.2.1 Runge-Kutta methods with scalar coefficients	20
3.2.2 Runge-Kutta methods with operator coefficients	27
3.3 Nonconvergence of Newton's method	28
3.3.1 Strong stability of discretization methods	28
3.3.2 Class of strongly stable Runge-Kutta methods	30
	33
CHAPTER 4. CONSTRUCTION OF ITERATIVE METHODS	
4.1 Iterative methods based on Runge-Kutta methods	33
with scalar coefficients	
4.2 Iterative methods based on Runge-Kutta methods	25
with operator coefficients	35
4.2.1 Preliminaries	35
4.2.2 Formula for iterative methods based on Runge-Kutta	
methods with operator coefficients	38
	43
CHAPTER 5. LOCAL CONVERGENCE	43
5.1 Necessary conditions and sufficient conditions	46
5.2 Further conditions on the iterative methods	16
5.2.1 Simpler sufficient conditions	40

	5.2.2 Iterative methods based on Runge-Kutta methods	
	with scalar coefficients	48
	5.2.3 Iterative methods based on Runge-Kutta methods	
	with operator coefficients	49
5.3	Proof of the theorems	50
	5.3.1 Preliminary lemmata	50
	5.3.2 Proof of Theorem 5.1.1	66
	5.3.3 Proof of the Theorems 5.2.1 and 5.2.2	67
	5.3.4 Proof of the Theorems 5.2.3-5	67
	5.3.5 Proof of the Theorems 5.2.6 - 8	69
CHAPTER 6. RADI	US OF CONVERGENCE	71
	PART I	
6.1	The class $F < \sigma, \beta, \gamma >$	72
6.2	The radius of convergence of Newton's method	
	with respect to $F < \sigma, \beta, \gamma >$	72
	6.2.1 Proof of Theorem 6.2.1	73
	6.2.2 Proof of Lemma 6.2.5 where $E = IR$	76
	6.2.3 The E-extension of a real function	80
6.3	Iterative methods with greater radii of convergence	86
	6.3.1 Proof of the Theorems 6.3.1 and 6.3.2	89
	PART II	
6.4	The class $F < \sigma, \alpha >$	97
6.5	The radius of convergence of Newton's method	
	with respect to $F < \sigma, \alpha >$	101
	6.5.1 Proof of Theorem 6.5.1	102
6.6	The radius of convergence of $M_{\overline{\alpha}}$ with respect to $F < \sigma, \alpha >$	105
	6.6.1 Proof of Theorem 6.6.1	111
	6.6.2 Proof of the Lemmata 6.5.5 and 6.6.8	114
CHAPTER 7. NUME	RICAL EXPERIMENTS	121
7.1	Iterative methods to be tested	121
7.2	Numerical results with the methods described in	
	section 7.1	125

CONTINUATION ALGORITHMS	131
CHAPTER 8. PREDICTOR-CORRECTOR CONTINUED TO THE	131
8.1 Introduction	134
8.2 Determination of Uk	136
8.3 Estimation of $\ U_k(t) - U(t)\ $	140
8.4 Local processes; estimation of the radii of convergence	1 4 2
8.4.1 Newton's method	142
8.4.2 A mixed method	145
8.5 Numerical results with the PCC algorithms	148
	157
REFERENCES	
	163

.

SUMMARY

CHAPTER 1

1

INTRODUCTION

1.1. IMBEDDING METHODS: A DESCRIPTION

Let E be a (real) Hilbert space, $D \subset E$ and let F: $D \rightarrow E$ be a nonlinear operator. In this monograph we shall be concerned with the numerical solution of the equation

(1.1.1) F(x) = 0.

Until further notice we assume that D = E.

Suppose that $x^* \in E$ is a solution of (1.1.1). A well-known method for approximating x^* is *Newton's method*. It consists of the calculation of a sequence of approximations $\{x_k^{}\}$ where

(1.1.2)
$$x_{k+1} = x_k - \Gamma(x_k)F(x_k)$$
 (k = 0,1,2,...).

In (1.1.2) $x_0 \in E$ is a given approximation to x^* and $\Gamma(x) \equiv [F'(x)]^{-1}$ where F'(x) denotes the Fréchet-derivative of F at x. Newton's method has the drawback that when the starting point x_0 is remote from x^* , the sequence $\{x_k\}$ defined in (1.1.2) generally will not converge to x^* . In many such cases *imbedding methods* - or *continuation methods*, as they are also called - appear to be able to generate a sequence $\{x_k\}$ that converges to x^* . In these methods a mapping H: $[\tau_0, \tau_1] \times E \to E$ is introduced, for which H(t, x) depends continuously on t, such that

 $H(\tau_0, x) = 0$ is easily solvable

(1.1.3) and

 $H(\tau_1, x) \equiv F(x).$

 $[\tau_0, \tau_1]$ is a closed interval in **R**. Thus the operator F is imbedded in the family of operators {H(t, \cdot) | t $\in [\tau_0, \tau_1]$ }.

We observe that if F does not depend naturally on a suitable parameter t, it is still always possible to define an H, satisfying (1.1.3), in several ways. For example, let $x_0 \in E$ be given, then

H:
$$[0,1] \times E \rightarrow E$$
,

(1.1.4)

$$H(t,x) = (1-t) \{F(x) - F(x_0)\} + tF(x)$$
 (for all $t \in [0,1]$ and $x \in E$)

satisfies the conditions (1.1.3). More generally, let K: $E \times E \rightarrow E$, where K may depend on F, satisfy K(x,x) = 0 (for all $x \in E$). Then, for any $x_0 \in E$, the operator

H:
$$[0,1] \times E \rightarrow E$$

(1.1.5)

 $H(t,x) = (1-t)K(x,x_0) + tF(x) \quad (for all t \in [0,1] and x \in E)$

meets the conditions (1.1.3).

We return to our original problem and suppose that a mapping H: $[\tau_0, \tau_1] \times E \rightarrow E$ has been introduced satisfying (1.1.3). Instead of the single problem (1.1.1), the entire family of problems

(1.1.6)
$$H(t,x) = 0$$
 (t $\in [\tau_0, \tau_1]$)

is considered. Let $u_0 \in E$ be the solution of $H(\tau_0, x) = 0$. Suppose that (1.1.6) has, for each $t \in [\tau_0, \tau_1]$, a unique solution x = U(t), which depends continuously on t. (See e.g. [MEYER, 1968] for the restrictions on H which ensure that such a curve U exists. We shall not go into this type of problem.) We have

(1.1.7) H(t,U(t)) = 0 (for all $t \in [\tau_0, \tau_1]$)

and

(1.1.8)
$$U(\tau_0) = u_0, \quad U(\tau_1) = x^*.$$

Thus U defines a curve in E with starting point u_0 and with end point equal to the solution \mathbf{x}^{\star} of (1.1.1). We shall describe some ways in which the imbedding can be used in the numerical solution of equation (1.1.1).

a. Discrete imbedding

As a first possibility for approximating $x^* = U(\tau_1)$ one may approximate successively the solutions of

(1.1.9)
$$H(t_i, x) = 0$$
 (i = 0,1,...,N)

by some numerical process (e.g. Newton's method). Here N is an integer and $\{t_0, t_1, \dots, t_N\}$ is a partition of $[\tau_0, \tau_1]$, that is, $\tau_0 = t_0 < t_1 < \dots < t_N = \tau_1$ holds. As starting point for the iterative process for approximating $U(t_i)$ the last iterate of the iterative process for approximating $U(t_{i-1})$ is often used (i \ge 1). If this approximation is close to U(t_{i-1}) and if $t_i - t_{i-1}$ is sufficiently small then hopefully a sequence may be generated that converges to U(t_i).

This way of approximating x^* is called *discrete imbedding*.

b. Transformation to an initial value problem

We next consider a somewhat different way of approximating $\mathbf{x}^{\star} = U(\tau_1)$. Assume that the mapping U, satisfying (1.1.7) is continuously differentiable on $[\tau_0, \tau_1]$ and that H has continuous partial Fréchet-derivatives. Differentiating the identity (1.1.7) with respect to t, we obtain

(1.1.10)
$$\partial_1^{H(t,U(t))} + \partial_2^{H(t,U(t))}\dot{U}(t) = 0$$
 (for all $t \in [\tau_0, \tau_1]$),

where $\partial_1 H$ and $\partial_2 H$ are the partial Fréchet-derivatives of H with respect to t and x respectively and $\dot{U}(t)$ denotes $\frac{d}{dt}U(t)$. If we assume that $\partial_2 H(t,U(t))$ is invertible (for all $t \in [\tau_0, \tau_1]$) then (cf. (1.1.10)) U satisfies the following initial value problem:

(1.1.11a)
$$\dot{U}(t) = -[\partial_2 H(t, U(t))]^{-1} \{\partial_1 H(t, U(t))\}$$
 (t $\epsilon [\tau_0, \tau_1]$),

$$(1.1.11b)$$
 $U(0) = u_0$.

Thus $x^* = U(\tau_1)$ may be approximated by applying a numerical integration procedure to (1.1.11). The approximation of \mathbf{x}^{\star} thus obtained can then be used as starting point for an iterative process for approximating \mathbf{x}^{\star} more closely (e.g. Newton's method).

Transforming (1.1.1) into the initial value problem (1.1.11) is often called *Davidenko's method*.

c. Iterative imbedding

We finally notice that imbeddings of type (1.1.5) may also be used to construct iterative methods for approximating x^* . To that end, let K: E × E \rightarrow E, such that K(x,x) = 0 (for all x ϵ E), be given. Let $x_0 \epsilon$ E and let H: [0,1] × E \rightarrow E be of type (1.1.5). In this case the initial value problem (1.1.11) reduces to

$$\dot{\mathbf{X}}(t) = -[(1-t)\partial_1 \mathbf{K}(\mathbf{X}(t), \mathbf{x}_0) + t\mathbf{F}'(\mathbf{X}(t))]^{-1} \{-\mathbf{K}(\mathbf{X}(t), \mathbf{x}_0) + \mathbf{F}(\mathbf{X}(t))\}$$
(1.1.12)

$$(t \in [0, 1]),$$

$$\mathbf{X}(0) = \mathbf{x}_0.$$

Computing the solution X(t) of (1.1.12) at t = 1 be means of a given numerical integration procedure, we obtain an approximation, say $x_1 \approx X(1)$, which is uniquely determined by x_0 . We thus have $x_1 = G(x_0)$, where the operator G depends only on F, K and the given numerical integration procedure. Solving (1.1.12) once more by the same numerical integration procedure, with x_0 replaced by x_1 , we obtain an approximation $x_2 \approx X(1)$, which is related to x_1 by $x_2 = G(x_1)$. In this way we arrive at the iterative process

$$(1.1.13) \qquad x_{k+1} = G(x_k) \qquad (k = 0, 1, 2, ...).$$

We call this way of constructing iterative methods, *iterative imbedding*. Hereafter we shall use for iterative imbedding initial value problems that are of a rather more general type than problem (1.1.12) (cf. section 2.6).

1.2. IMBEDDING METHODS: HISTORICAL SURVEY AND CURRENT STATUS

Originally, imbedding methods were only used as tools to demonstrate the existence of solutions to operator equations. In [FICKEN, 1951] a review is given of such applications, which date back at least to the last century.

The idea of using the *discrete imbedding method* for the numerical solution of nonlinear equations seems to date back to [LAHAYE, 1934, 1935] (see also [ORTEGA & RHEINBOLDT, 1970; pp. 234-235] for a bibliography on early

numerical applications). More recent investigations have been performed by e.g. [AVILA, 1974], [LAASONEN, 1970], [POMENTALE, 1974], [WACKER, 1971, 1974, 1977(a,b)] and [RIBARIČ & SELIŠKAR, 1974]. In [LEDER, 1974] and [RHEINBOLDT, 1975, 1976] adaptive methods are proposed for determining the partition $\{t_0, t_1, \dots, t_N\}$.

The idea of transforming problem (1.1.1) into the initial value problem (1.1.11) is usually attributed to Davidenko ([DAVIDENKO, 1953]). Davidenko applied this method to a variety of problems including integral equations and matrix inversion (e.g. [DAVIDENKO, 1965(a,b), 1975]). In [RALL, 1968] an exposition of Davidenko's work is given (it also contains some translations and a bibliography). A review of still more applications (among which two-point boundary value problems) is given in [WASSERSTROM, 1973]. In [MEYER, 1968] and [BOSARGE, 1971] the method is considered, with the initial value problem (1.1.11) solved by Runge-Kutta methods.

In [GAVURIN, 1958] a somewhat different way of transforming problem (1.1.1) into an initial value problem is used. Any iterative process of type

(1.2.1)
$$x_{k+1} = x_k + P(x_k)$$
 (k = 0,1,2,...),

where $x_0 \in E$ is given, can be conceived as an application of Euler's method (see e.g. [LAMBERT, 1973; p. 13]) with stepsize h = 1 to the initial value problem

 $\dot{Y}(t) = P(Y(t))$ (t $\in [0, \infty)$), $Y(0) = x_0$.

The curve Y is called the *continuous analogue of the iterative process* (1.2.1) (see also [ROSENBLOOM, 1956], [BITTNER, 1967] and [UEBERHUBER, 1976]).

This approach is closely related to Davidenko's method (cf. [MEYER, 1968]). For example, let

H: $[0,\infty)$ × E → E, (1.2.3)

(1.2.2)

 $H(t,x) = F(x) - e^{-t}F(x_0) \qquad (for all t \in [0,\infty) and x \in E).$

Suppose that for all t ϵ [0, ∞) the problem H(t,x) = 0 has a unique solution x = U(t), which depends continuously on t. We have

(1.2.4)
$$F(U(t)) - e^{-t}F(x_0) = 0, \quad (t \in [0,\infty))$$

If we assume that F'(U(t)) exists and is invertible (for all $t \in [0,\infty)$) then differentiating (1.2.4) with respect to t and using the identity (1.2.4) yield

$$\dot{U}(t) = -\Gamma(U(t))F(U(t))$$
 (t $\in [0,\infty)$),
(1.2.5)
 $U(0) = x_0$.

Thus the curve Y = U, satisfying (1.2.5), is the continuous analogue of Newton's method. See also [BOGGS, 1971], [BOGGS & DENNIS, 1974], [BITTNER, 1967] and [DI LENA & TRIGIANTE, 1976].

We also mention the possibility of combining the discrete imbedding method with Davidenko's method. For example, let a partition $\{t_0, t_1, \ldots, t_N\}$ of $[\tau_0, \tau_1]$ be given. Then the first step of the k-th stage (k = 1,2,...,N) of this process consists of computing an approximation u_k of $U(t_k)$ by applying a numerical integration procedure on (1.1.11a) at $[t_{k-1}, t_k]$. The second step of the k-th stage consists of solving the equation (1.1.9), for i = k by means of an iterative process with starting point u_k . See e.g. [DEIST & SEFOR, 1967] and [BROYDEN, 1969]. More recently in [FEILMEIER, 1972], [KUBIČEK, 1976], [MENZEL & SCHWETLICK, 1976], [SCHWETLICK, 1975, 1976], [DEUFLHARD, PESCH & RENTROP, 1976] and [DEUFLHARD, 1976] these types of methods have been investigated.

The method of *iterative imbedding* has been investigated in [KIZNER, 1964] for the case E = IR. In this paper the relation is given between the order of accuracy of a numerical integration method for solving the initial value problem (1.1.12) and the order of convergence of the corresponding iterative process (1.1.13). In [DAVIDENKO, 1966], [BITTNER, 1967], [KLEIN-MICHEL, 1968] and [PETRY, 1971] the iterative imbedding method is considered for the case that E is IR^n or an arbitrary Banach space. In these papers the emphasis lies on constructing high order iterative methods of type (1.1.13). In the papers of Kleinmichel and Petry theorems on these iterative methods are given which are similar to the famous Newton-Kantorovich theorem (cf. [KANTOROWITSCH & AKILOW, 1964; Theorem 6 (1.XVIII)]).

1.3. SCOPE OF THE STUDY

In this monograph we shall be concerned mainly with *iterative imbedding*. As described in section 1.1, in this way iterative methods (1.1.13) for solving (1.1.1) can be constructed. These iterative methods are based on the operator K (determining the initial value problem (1.1.12)) and on a numerical integration procedure. We shall restrict our attention to integration procedures of (generalized) Runge-Kutta type.

In the last chapter we shall consider some *algorithms* based on *discrete imbedding*, for the solution of problem (1.1.1). In some of them *Davidenko's method* is used also.

In chapter 2 we introduce the basic concepts to be used throughout this study. In Section 2.4 we introduce the concept of a radius of convergence r(M;F) of an iterative method M with respect to some (given) class F of operators F. Let x^* denote the solution of F(x) = 0 where F belongs to F. Then r(M;F) indicates how close to x^* an initial guess x_0 should be, for the iterative process to yield a sequence of approximations x_k that converges to x^* . Section 2.5 is concerned with the local convergence behaviour of an iterative process (i.e. its convergence behaviour near the solution x^* of F(x) = 0). We introduce two types of local convergence behaviour that lie between the well-known concepts of local and quadratic convergence. We derive both necessary and sufficient conditions for these intermediate types of local convergence - generalizing the results of [OSTROWSKI, 1960; Theorems 22.1 and 22.2] and [KITCHEN, 1966]. In section 2.6 we present the imbedding on which the iterative methods with which we shall be concerned will be based.

In chapter 3 (section 3.2) we introduce the Runge-Kutta methods that will be used for constructing the iterative processes (1.1.13). In section 3.3 we shall indicate that the problem of the nonconvergence of Newton's method is closely related to a certain type of instability of Euler's method.

In chapter 4 we derive general formulae for iterative methods which are constructed by means of iterative imbedding.

In chapter 5 we investigate the local convergence behaviour of the iterative methods that were constructed in chapter 4. We derive both necessary and sufficient conditions for quadratic convergence and for the two intermediate types of convergence introduced in section 2.5. Our main results in this chapter are formulated in nine theorems, that are given in sections 5.1 and 5.2. The most obvious conditions on the operator K, for which the iterative process (1.1.13) is locally convergent, are given in the

Theorems 5.2.3-8 (cf. subsections 5.2.2-3). Some of these theorems may be viewed as generalizations of the results of [KIZNER, 1964].

In chapter 6 we determine the radii of convergence of iterative methods of the type described in section 4.1. Our main results in this chapter are formulated in four theorems (Theorems 6.2.1, 6.3.1, 6.5.1 and 6.6.1). Part I of this chapter is concerned with $F < \sigma, \beta, \gamma >$, a class of operators F which is defined in section 6.1. In section 6.2 we determine the radius of convergence of Newton's method with respect to $F < \alpha, \beta, \gamma >$. A related result is given in [RHEINBOLDT, 1975] although in that paper it was only shown that the value obtained is a lower bound of the radius of convergence. In section 6.3 we prove that a class of iterative methods that are closely related to the socalled damped Newton methods, all have a greater radius of convergence than Newton's method. Part II of chapter 6 is concerned with $F < \sigma, \alpha >$, a class of operators F which is defined in section 6.4. In section 6.5 we determine the radius of convergence of Newton's method with respect to this class. Finally, in section 6.6 we are able to give an explicit expression of the radii of convergence (with respect to $F(\sigma, \alpha)$) of the iterative methods which were considered in section 6.3.

In chapter 7 we present numerical experiments with iterative methods which were dealt with in the preceding chapters. Iterative methods based on a generalized Runge-Kutta method related to the Backward Euler method, appear to be more successful in solving the testproblems than the other methods tested (including Newton's method).

In chapter 8 we present some algorithms for solving problem (1.1.1). In section 8.1 we describe an algorithm which is proposed in [RHEINBOLDT, 1975]. This algorithm is based on discrete imbedding and has an adaptive step strategy for determining the partition $\{t_0, t_1, \ldots, t_N\}$ (cf. section 1.1). In the sections 8.2, 8.3 and 8.4 we present some variants of this algorithm, in some of which Davidenko's method is used also (cf. section 1.1). These variants also use the results of section 6.5. From the numerical results given in section 8.5, it appears that all these algorithms are very reliable, but the algorithms in which Davidenko's method is used require the least amount of work for the solution of a problem.

CHAPTER 2

PRELIMINARIES

In this chapter we introduce some well-known concepts such as the spectral radius of a linear operator (section 2.2). In section 2.3 we give a definition of an iterative method. In section 2.4 the concept of the radius of convergence of an iterative method is introduced. Let x^* be the solution of (1.1.1). Then this concept indicates how close to x^* an initial guess x_0 must be in order that the application of the iterative method to problem (1.1.1) may yield a sequence of approximations x_k that converges to x^* . Section 2.5 is concerned with the local convergence behaviour of iterative processes. We introduce two types of local convergence behaviour that lie between the well-known concepts of local and quadratic convergence. In section 2.6 we present the imbedding on which the iterative methods to be investigated in the following chapters will be based.

2.1. CONVENTIONS AND NOTATIONS

From now on the following conventions hold.

If F is an operator, then D(F) denotes its domain. Let Z_1 and Z_2 be Banach spaces. Then

$$L[z_1, z_2] = \{ C \mid C \colon z_1 \to z_2; C \text{ is linear and bounded} \}.$$

Furthermore we define the sets $L^{(n)}[z_1,z_2]$ by

$$L^{(1)}[z_1, z_2] = L[z_1, z_2]$$
 and $L^{(n)}[z_1, z_2] = L[z_1, L^{(n-1)}[z_1, z_2]]$
(n = 2,3,...).

Let ℓ and n be the positive integers with $\ell \leq n$. Let $Q \in L^{(n)}[z_1, z_2]$ and $y_j \in Z_1$ $(j = 1, 2, ..., \ell)$. We use the notation

$$Qy_1y_2\cdots y_\ell = (\dots (Qy_1)y_2)\dots)y_\ell.$$

Let F: D(F) $\rightarrow \mathbb{Z}_2$ with D(F) $\subset \mathbb{Z}_1$ and let x ϵ interior(D(F)). Then F'(x) (or F⁽¹⁾(x)) denotes the (*Fréchet-*)*derivative* of F at x (F'(x) $\epsilon L[\mathbb{Z}_1,\mathbb{Z}_2]$). F"(x) (or (F⁽²⁾(x)) denotes the *second (Fréchet-*)*derivative* of F at x (it is the Fréchet-derivative of the operator F' at x, F"(x) $\epsilon L^{(2)}[\mathbb{Z}_1,\mathbb{Z}_2]$). The n-th (*Fréchet-*)*derivative* of F at x (n = 3,4,...) is denoted by F⁽ⁿ⁾(x).

If $z_1 = z_2 = z$ we set $L[z_1, z_2] = L[z]$ and $L^{(n)}[z_1, z_2] = L^{(n)}[z]$ (n = 1,2,...).

If $C \in L[Z]$, then C is said to be *invertible* if $T \in L[Z]$ exists such that CT = TC = I, the identity (we write $T = [C]^{-1}$).

For i = 1,2,...,n+1 let Z_i be a given Banach space with norm $\|\cdot\|_i$. Let $Z_0 = Z_1 \times Z_2 \times \ldots \times Z_n$ be the product space. We shall always assume that the norm in Z_0 is $\|\cdot\|_0$ where $\|x\|_0 = \max\{\|x_i\|_i \mid 1 \le i \le n\}$ (for all $x = (x_1, x_2, \ldots, x_n) \in Z_0$). Let P: D(P) $\Rightarrow Z_{n+1}$ with D(P) $\subset Z_0$. Let $x = (x_1, x_2, \ldots, x_n) \in Z_0$. For $i = 1, 2, \ldots, n$ we define the operator $P(x_1, \ldots, x_{i-1}, \cdot, x_{i+1}, \ldots, x_n)$ by

$$P(x_{1},...,x_{i-1},\cdot,x_{i+1},...,x_{n}) = Q$$

with

$$D(Q) = \{v \mid v \in Z_{i}; (x_{1}, \dots, x_{i-1}, v, x_{i+1}, \dots, x_{n}) \in D(P)\},\$$

$$Q(v) = P(x_{1}, \dots, x_{i-1}, v, x_{i+1}, \dots, x_{n}) \quad (\text{for all } v \in D(Q)).$$

For $x_i \in interior(D(P(x_1, \dots, x_{i-1}, \cdot, x_{i+1}, \dots, x_n)))$ we denote by $\partial_i P(x)$ the partial (Fréchet-)derivative of P with respect to x_i at x (i = 1,2,\dots,n). For $x_j \in interior(D(\partial_i P(x_1, \dots, x_{j-1}, \cdot, x_{j+1}, \dots, x_n)))$ we denote be $\partial_{ij} P(x)$ the derivative $\{\partial_i [\partial_i P]\}(x)$ (i,j = 1,2,...,n).

E denotes a real Hilbert space, with inner product (•,•) and norm $\|\cdot\| = (\cdot, \cdot)^{\frac{1}{2}}$. We shall always assume that $E \neq \{0\}$. Let $x \in E$ and $\sigma \in (0, \infty]$. We set

$$B(\mathbf{x},\sigma) = \{\mathbf{y} \mid \mathbf{y} \in \mathbf{E}; \|\mathbf{y}-\mathbf{x}\| < \sigma\}.$$

Furthermore if $V \subseteq E$ is a subset of E, then \overline{V} denotes the *closure* of V.

For a detailed definition of the above concepts we refer to [KANTOROWITSCH & AKILOW, 1964].

Let F: $D(F) \rightarrow E$ with $D(F) \subset E$ and let $x \in interior(D(F))$. Suppose F is Fréchet-differentiable at x, and suppose F'(x) is invertible. Then $\Gamma(x)$ denotes $[F'(x)]^{-1}$.

Let X: $[0,1] \rightarrow E$, then $\dot{X}(t)$ denotes $\frac{d}{dt} X(t)$ (t $\in [0,1]$).

Let $\phi: D(\phi) \rightarrow \mathbb{R}$ with $D(\phi) \subset \mathbb{R}$. ϕ is *isotone* (antitone) on $D(\phi)$ if $\phi(\xi_1) \leq \phi(\xi_2)$ ($\phi(\xi_1)$) $\geq \phi(\xi)$) whenever $\xi_1, \xi_2 \in D(\phi)$ and $\xi_1 \leq \xi_2$. ϕ is strictly isotone (anitone) on $D(\phi)$ if $\phi(\xi_1) < \phi(\xi_2)$ ($\phi(\xi_1) > \phi(\xi_2)$) whenever $\xi_1, \xi_2 \in D(\phi)$ and $\xi_1 \leq \xi_2$.

Let m and n be integers. We shall always use the conventions

$$\sum_{i=m}^{n} \dots = 0 \quad \text{and} \quad \prod_{i=m}^{n} \dots = 1 \quad (\text{if } m > n).$$

IR denotes the set of real numbers, C denotes the set of complex numbers. \mathbb{R}^n and \mathbb{C}^n denote the real and the complex n-dimensional vector space, respectively. Finally, $\mathbb{N} = \{1, 2, 3, \ldots\}$.

2.2. THE SPECTRUM OF A LINEAR OPERATOR

Let $E_{\mathbb{C}}$ denote the *complex extension* of E (see [KANTOROWITSCH & AKILOW, 1964; section 2(XIII)]). $E_{\mathbb{C}}$ is a complex Hilbert space. We denote its innerproduct by $(\cdot, \cdot)_{\mathbb{C}}$. The norm in $E_{\mathbb{C}}$ is $\|\cdot\|_{\mathbb{C}} = (\cdot, \cdot)_{\mathbb{C}}^{\frac{1}{2}}$. We note that E is a subspace of $E_{\mathbb{C}}$. Any element z of $E_{\mathbb{C}}$ can be written in the form z = x + iy where $x, y \in E$ are uniquely determined. Let $z_1, z_2 \in E_{\mathbb{C}}$. Then

$$(z_1, z_2)_{\mathfrak{C}} = (x_1, x_2) + (y_1, y_2) + i\{(y_1, x_2) - (x_1, y_2)\}$$

where $x_j, y_j \in E$ and $z_j = x_j + iy_j$ (j = 1, 2). Let $T \in L(E_{\sigma})$.

DEFINITION 2.2.1. The spectrum of T is the set

(2.2.1) $sp(T) = \{ \alpha \mid \alpha \in \mathbb{C}; [T-\alpha I] \text{ is not invertible} \}.$

sp(T) is non-empty (cf. [RUDIN, 1973; Theorem 10.13(a)]).

DEFINITION 2.2.2. The number

 $(2.2.2) \qquad \operatorname{sr}(T) = \sup\{ |\alpha| \mid \alpha \in \operatorname{sp}(T) \}$

is called the *spectral radius* of T.

We have

THEOREM 2.2.1.

$$\operatorname{sr}(\mathbf{T}) = \lim_{n \to \infty} \|\mathbf{T}^{n}\|^{n} = \inf_{n \ge 1} \|\mathbf{T}^{n}\|^{n}.$$

PROOF. cf. [RUDIN, 1973; Theorem 10.13(b)].

Let C \in L(E), and let

(2.2.3)

$$C_{c}z = Cx + iCy$$
 ($z \in E_{c}$, $z = x + iy$, $x, y \in E$).

.

Then $C_{\mathbb{C}} \in L(E_{\mathbb{C}})$ and it follows that

 $C_{\mathbb{C}}: E_{\mathbb{C}} \rightarrow E_{\mathbb{C}}'$

$$(2.2.4) \| C_{\mathfrak{C}} \|_{\mathfrak{C}} = \| C \|_{\mathfrak{C}}$$

 $C_{\mathbb{C}}$ is called the *extension of* C *in* $E_{\mathbb{C}}$. The *spectrum of* C *is* defined by $sp(C) = sp(C_{\mathbb{C}})$. Analogously, the *spectral radius of* C *is* defined by $sr(C) = sr(C_{\mathbb{C}})$. From (2.2.4) and Theorem 2.2.1 it follows that

(2.2.5)
$$\operatorname{sr}(C) = \lim_{n \to \infty} \|C^n\|^{\frac{1}{n}} = \inf_{n \ge 1} \|C^n\|^{\frac{1}{n}}.$$

Let ρ be a rational function with real coefficients, i.e. $\rho\left(z\right)$ Ξ $\left[q\left(z\right)\right]^{-1}p(z)$ where p and q are polynomials with real coefficients. We define

(2.2.6)
$$D_{\mathbb{C}}(\rho) = \{z \mid z \in \mathbb{C}; q(z) \neq 0\}$$

and

 $(2.2.7) \qquad D_{E}(\rho) = \{C \mid C \in L(E); q(C) \text{ is invertible} \}.$

Let C ϵ D_E(ρ). Then ρ (C) denotes the linear operator [q(C)]⁻¹p(C).

THEOREM 2.2.2. Let $\rho: D(\rho) \rightarrow \mathbb{C}$ be a rational function with real coefficients. Let $C \in L(E)$. If $sp(C) \subset D_{\mathbb{C}}(\rho)$ then

 $C \in D_{E}(\rho)$ and $sp(\rho(C)) = \rho(sp(C))$.

<u>PROOF</u>. Let $\rho(z) \equiv [q(z)]^{-1}p(z)$.

- 1. It follows from [RUDIN, 1973; Theorem 10.28(a)] that $q(C_{\mathbb{C}})$ is invertible. Since $q(C_{\mathbb{C}}) = [q(C)]_{\mathbb{C}}$ we have $C \in D_{E}(\rho)$.
- 2. Since $q(C_{\mathfrak{g}}) = [q(C)]_{\mathfrak{g}}, p(C_{\mathfrak{g}}) = [p(C)]_{\mathfrak{g}}$ we have $\rho(C_{\mathfrak{g}}) = [\rho(C)]_{\mathfrak{g}}$. From [RUDIN, 1973; Theorem 10.28(b)] follows that

 $sp(\rho(C_{\mathbb{C}})) = \rho(sp(C_{\mathbb{C}})).$

Consequently $sp(\rho(C)) = \rho(sp(C_{C})) = \rho(sp(C))$, which completes the proof of the theorem.

2.3. ITERATIVE METHODS

In this section we introduce some concepts that will play an important role in our investigations. Throughout,

 $(2.3.1) \qquad G = \{G \mid G: D(G) \rightarrow E; D(G) \subset E\}$

and

 $(2.3.2) \qquad F = \{F \mid F: D(F) \rightarrow E; D(F) \subset E; F(x) = 0 \text{ has a unique solution} \}.$

Let $F \in F$ be given. Then x^* will always denote the solution of F(x) = 0. Let $\hat{F} \subset F$.

<u>DEFINITION 2.3.1</u>. Any operator with domain \hat{F} and range in G is called an *iterative method* for \hat{F} .

We note that this concept of an iterative method is much wider than what is intuitively understood by the term. However, for ease of presentation, we shall work with the above concept of an iterative method.

Let M be an iterative method for \hat{F} and let F ϵ \hat{F} . Set

 $(2.3.3) D(M,F) = \{x_0 \mid \text{there exists a sequence } \{x_k\} \text{ such that } x_k \in D(G) \text{, where} \\ G = M(F) \text{, and } x_{k+1} = G(x_k) \quad (k = 0, 1, 2, ...) \}.$

<u>DEFINITION 2.3.2</u>. The *iterative process*, corresponding to the iterative method M and the function $F \in \hat{F}$ will be denoted by [M,F], and consists in computing vectors $\mathbf{x}_{\mathbf{k}}$ from

(2.3.4a)
$$x_{k+1} = G(x_k)$$
 (k = 0,1,2,...)

where

$$(2.3.4b)$$
 G = M(F).

In this connection, the operator G in (2.3.4b) is called an *iteration* function. The starting point x_0 of (2.3.4a) should be an element of D(M,F) in order to prevent the iterative process from breaking off prematurely.

<u>DEFINITION 2.3.3</u>. The set D(M,F) defined in (2.3.3) is called the *domain* of the iterative process [M,F].

Let $x_0 \in D(M,F)$. Then the sequence $\{x_k\}$ generated by x_0 and the iterative process [M,F] is, of course, defined by (2.3.4).

DEFINITION 2.3.4. The set

(2.3.5)
$$S(M,F) = \{x_0 \mid x_0 \in D(M,F) \text{ and the sequence } \{x_k\} \text{ generated by } x_0 \text{ and } [M,F] \text{ converges to } x^*\}$$

is called the region of convergence of the iterative process [M,F].

2.4. THE RADIUS OF CONVERGENCE

Let $F \in F$. As pointed out in the introduction, we are interested in iterative methods M such that the related iterative processes [M,F] generate sequences $\{x_k\}$ that converge to x^* , even if the starting point x_0 is remote from x^* .

In this section we introduce the concept of the radius of convergence of an iterative method M. This concept indicates how close to x^* a starting point x_0 should be for the iterative process [M,F] to generate a sequence

 $\{x_k\}$ that converges to x^* .

Let \hat{F} be a non-empty subset of F.

DEFINITION 2.4.1. Let M be an iterative method (for \hat{F}) and F ϵ \hat{F} . We define

$$r(M,F) = \begin{cases} 0 & (\text{if } x^* \notin \text{ interior}(S(M,F))), \\ \\ \sup_{\sigma \in \sigma} \sigma > 0 \text{ and } B(x,\sigma) \subset S(M,F) \end{cases} \text{ (otherwise).}$$

r(M,F) is called radius of convergence of the iterative process [M,F].

DEFINITION 2.4.2. Let M be an iterative method (for \hat{F}) and let F_0 be a non-empty subset of \hat{F} . Then

$$\mathbf{r}(\mathbf{M};\mathbf{F}_{0}) = \inf\{\mathbf{r}(\mathbf{M},\mathbf{F}) \mid \mathbf{F} \in \mathbf{F}_{0}\}$$

is called the radius of convergence of the iterative method M with respect to ${\rm F}_0.$

It is clear that the larger $r(M;F_0)$ is for an iterative method M, the less sensitive with regard to starting points will be the iterative processes [M,F] (F ϵ F₀) generated by it.

2.5. LOCAL CONVERGENCE BEHAVIOUR OF ITERATIVE PROCESSES

Let $\hat{F} \subset F$ and let M be an iterative method for \hat{F} . Let $F \in \hat{F}$.

<u>DEFINITION 2.5.1</u>. The iterative process [M,F] is called *locally convergent* (LC) is a neighbourhood V of x^* exists such that V \subset S(M,F).

Obviously the iterative process will have a positive radius of convergence if and only if it is locally convergent. We give three other definitions concerning the convergence behaviour of [M,F] near x^* .

DEFINITION 2.5.2. The iterative process [M,F] is called *stably convergent* (SC) if a constant $\theta > 0$, an N₀ $\in \mathbb{N}$ and a neighbourhood V of x^* exist such that $V \subset S(M,F)$, and for all $x_0 \in V$, the sequence $\{x_k\}$ generated by x_0 and [M,F] satisfies

 $\|\mathbf{x}_{k} - \mathbf{x}^{*}\| \leq \theta \|\mathbf{x}_{0} - \mathbf{x}^{*}\| \qquad (\text{for all } k \geq N_{0}).$

<u>DEFINITION 2.5.3</u>. The iterative process [M,F] is called *regularly convergent* (RC) if a neighbourhood V of x^* exists such that $V \subseteq S(M,F)$ and for all $x_0 \in V$, the sequence $\{x_k\}$ generated by x_0 and [M,F] satisfies

$$\|\mathbf{x}_{k} - \mathbf{x}^{*}\| \leq \|\mathbf{x}_{0} - \mathbf{x}^{*}\|$$
 (k = 1, 2, ...).

DEFINITION 2.5.4. The iterative process [M,F] is called *quadratically convergent* (QC) if a neighbourhood V of x^* and a number $\delta > 0$ exist such that $V \subseteq S(M,F)$ and for all $x_0 \in V$, the sequence $\{x_k\}$ generated by x_0 and [M,F] satisfies

$$\|\mathbf{x}_{k+1} - \mathbf{x}^{*}\| \le \delta \|\mathbf{x}_{k} - \mathbf{x}^{*}\|^{2}$$
 (k = 0,1,2,...).

It is clear that the following relations hold.

 $QC \Rightarrow RC \Rightarrow SC \Rightarrow LC$.

If [M,F] is locally convergent and G = M(F) is continuous in x^* then obviously $G(x^*) = x^*$. This means that x^* is a *fixed point* of G.

Conversely, the following theorem holds.

<u>THEOREM 2.5.1</u>. Suppose x^* is a fixed point of G = M(F). Let $x^* \in interior(D(G))$ and let G be differentiable at x^* . The following four statements (i) - (iv) hold. (i) If $sr(G'(x^*)) < 1$ then [M,F] is stably convergent. (ii) If $sr(G'(x^*)) > 1$ then [M,F] is not stably convergent. (iii) If $\|G'(x^*)\| < 1$ then [M,F] is regularly convergent. (iv) If $\|G'(x^*)\| > 1$ then [M,F] is not regularly convergent.

In order to prove Theorem 2.5.1 we need two lemmata.

LEMMA 2.5.2. Let P: $D(P) \rightarrow E$ with $D(P) \subseteq E$. Suppose

- (a) The operator P has a fixed point y^* with $y^* \in interior(D(P))$ and P is differentiable at y^* .
- (b) $\|\mathbf{P}'(\mathbf{y}^*)\| \leq \alpha$ and $\|[\mathbf{P}'(\mathbf{y}^*)]^n\|^{1/n_0} \leq \delta < 1$ for some $n_0 \geq 1$ and real numbers α and δ .

Then for any $\varepsilon > 0$ with $\delta_{-}^{n_0} + \varepsilon < 1$ there exists a number $\sigma > 0$ such that for all $x \in B(y^*, \sigma)$ and all $n \ge 1$ we have

$$\mathbf{x} \in \mathbf{D}(\mathbf{p}^n)$$
 and $\|\mathbf{p}^n(\mathbf{x}) - \mathbf{y}^*\| \le c\lambda^{\ell} \|\mathbf{x} - \mathbf{y}^*\|$.

Here $c = \max\{\alpha^j + \varepsilon \mid 1 \le j \le n_0\}$, $\lambda = \delta^{n_0} + \varepsilon$ and $\ell = entier(\frac{n-1}{n_0})$.

<u>PROOF</u>. Let $\varepsilon > 0$. Set $\lambda = \delta^{n_0} + \varepsilon$, suppose $\lambda < 1$. Let $j \in \mathbb{N}$ and $j \leq n_0$. The operator P^j is defined in a neighbourhood of y^* , $P^j(y^*) = y^*$ and P^j is continuous in y^* . From the *chainrule* (cf. [KANTOROWITSCH & AKILOW, 1964; section 1.2(XVII)] it follows that P^j is differentiable at y^* and

(2.5.1)
$$[P^{j}]'(y^{*}) = [P'(y^{*})]^{j}.$$

Consequently, a number $\sigma_1 > 0$ exists such that for all $x \in B(y^*, \sigma_1)$ and all $j \le n_0$ we have $x \in D(P^j)$ and

$$\|\mathbf{p}^{\mathsf{j}}(\mathbf{x}) - \mathbf{p}^{\mathsf{j}}(\mathbf{y}^{\mathsf{*}}) - [\mathbf{p}^{\mathsf{j}}] \cdot (\mathbf{y}^{\mathsf{*}}) (\mathbf{x} - \mathbf{y}^{\mathsf{*}}) \| \leq \varepsilon \|\mathbf{x} - \mathbf{y}^{\mathsf{*}}\|.$$

Hence for all $x \in B(y^*, \sigma_1)$ we have

$$\| \mathbf{p}^{j}(\mathbf{x}) - \mathbf{y}^{\star} \| \leq (\alpha^{j} + \varepsilon) \| \mathbf{x} - \mathbf{y}^{\star} \| \qquad (\text{for all } j \leq n_{0})$$

and

$$\|p^{n_{0}}(x) - y^{*}\| \leq \lambda \|x - y^{*}\|.$$

Let $c = \max\{\alpha^j + \varepsilon \mid 1 \le j \le n_0\}$. Set $\sigma = \min\{\sigma_1, \frac{\sigma_1}{c}\}$. Let $n \ge 1$. Then $n = \ell n_0 + k$ where $\ell \ge 0$ and $1 \le k \le n_0$. Then for all $x \in B(y^*, \sigma)$ we have $x \in D(P^n)$ and

$$\|p^{n}(x) - y^{\star}\| = \|[p^{n}]^{\ell}(p^{k}(x)) - y^{\star}\| \le \lambda^{\ell}\|p^{k}(x) - y^{\star}\| \le c\lambda^{\ell}\|x - y^{\star}\|.$$

This proves the lemma. \Box

LEMMA 2.5.3. Let P: D(P) \rightarrow E with D(P) \subset E. Let the assumption (a) of lemma 2.5.2 be fulfilled. Suppose $\| [P'(y^*)]^{n_0} \|^{1/n_0} \ge \delta > 1$ for some $n_0 \ge 1$. Then for all positive numbers ε and σ there exists an $\tilde{x} \in B(y^*, \sigma)$ with $\tilde{x} \neq y^*$ such that $\tilde{x} \in D(P^{n_0})$ and

$$\|\mathbf{p}^{\mathbf{n}_{0}}(\widetilde{\mathbf{x}}) - \mathbf{y}^{\star}\| \geq (\delta^{\mathbf{n}_{0}} - \epsilon) \|\widetilde{\mathbf{x}} - \mathbf{y}^{\star}\|.$$

<u>PROOF</u>. Let ε and σ be positive numbers. From the chainrule it follows that $[p^{n_0}]'(y^*)$ exists. Consequently a number $\tilde{\sigma} \in (0,\sigma]$ exists such that for all $x \in B(y^*,\tilde{\sigma})$ we have $x \in D(p^{n_0})$ and

(2.5.2)
$$\|P^{n_{0}}(x) - P^{n_{0}}(y^{*}) - [P^{n_{0}}]'(y^{*})(x-y^{*})\| \leq \frac{\varepsilon}{2}\|x-y^{*}\|.$$

Moreover an element y ϵ E with $\|y\| = 1$ exists such that

(2.5.3)
$$\| [P'(y^*)]^n_y \| \ge (\delta^{n_0} - \frac{\varepsilon}{2}) \| y \|.$$

Let $\widetilde{\mathbf{x}} = \mathbf{y}^* + \frac{\widetilde{\sigma}}{2} \mathbf{y}$. Then (2.5.1), (2.5.2) and (2.5.3) yield $\|\mathbf{p}^{\mathbf{n}_0}(\widetilde{\mathbf{x}}) - \mathbf{y}^*\| \ge (\delta^{\mathbf{n}_0} - \frac{\varepsilon}{2}) \|\widetilde{\mathbf{x}} - \mathbf{y}^*\| - \frac{\varepsilon}{2} \|\widetilde{\mathbf{x}} - \mathbf{y}^*\|.$

This proves the lemma. \Box

We now turn to the proof of Theorem 2.5.1. (i) If $sr(G'(x^*)) < 1$ then from (2.2.5) it follows that an $n_0 \ge 1$ exists such that

$$\| [G'(\mathbf{x}^{\star})]^{n_{0}} \|_{0}^{\frac{1}{n_{0}}} \leq \delta < 1.$$

Let $\varepsilon > 0$ with $\delta^{0} + \varepsilon < 1$. According to Lemma 2.5.2 a positive number σ exists such that for all $x \in B(x^{*}, \sigma)$ and $n \ge 1$ we have $x \in D(G^{n})$ and

$$\|G^{n}(x) - x^{*}\| \leq c\lambda^{\ell} \|x - x^{*}\|.$$

Here $c = \max\{\alpha^j + \varepsilon \mid 1 \le j \le n_0\}, \alpha = \|G'(x^*)\|, \lambda = \delta^{n_0} + \varepsilon$ and $\ell = entier(\frac{n-1}{n_0})$. Hence [M,F] is stably convergent. (ii) Suppose $sr(G'(x^*)) > 1$ and [M,F] is stably convergent. Thus positive

(ii) Suppose sr(G'(x^{*})) > 1 and [M,F] is stably convergent. Thus positive numbers $\tilde{\sigma}$ and θ and an integer N₀ ≥ 1 exist such that for all $x \in B(x^*, \tilde{\sigma})$ and all $n \ge N_0$ we have $x \in D(G^n)$ and

$$\|G^{n}(x) - x^{*}\| \leq \theta \|x - x^{*}\|.$$

Furthermore from (2.2.5) it follows that an $\widetilde{N} \ge N_0$ and a $\delta > 1$ exist such that

$$\| [G'(\mathbf{x}^*)]^n \|^n \ge \delta \qquad (\text{for all } n \ge \widetilde{N}).$$

Hence Lemma 2.5.3 applies where $\sigma \in (0, \tilde{\sigma}]$, $\varepsilon > 0$ and $n_0 \ge \tilde{N}$ are such that $\delta^{n_0} - \varepsilon > \theta$. This yields a contradiction.

(iii) If $\|G'(x^*)\| < 1$, Lemma 2.5.2 applies with $n_0 = 1$ and $\alpha = \delta < 1$. This proves the result.

(iv) Suppose $\|G'(\mathbf{x}^*)\| = \delta > 1$. If [M,F] is regularly convergent then a positive number σ exists such that $B(\mathbf{x}^*,\sigma) \subset D(G)$ and

$$\|G(\mathbf{x}) - \mathbf{x}^*\| \leq \|\mathbf{x} - \mathbf{x}^*\| \qquad (\text{for all } \mathbf{x} \in B(\mathbf{x}^*, \sigma)).$$

However Lemma 2.5.3 applies with $n_0 = 1$ and $\varepsilon > 0$ such that $\delta - \varepsilon > 1$. This yields a contradiction. Hence [M,F] is not regularly convergent.

REMARK 2.5.1. We note that Theorem 2.5.1 remains valid if E is a Banach space.

The sufficiency of the condition $sr(G'(x^*)) < 1$ for [M,F] to be locally convergent where $E = \mathbb{R}$, was proved by Schröder (cf. [SCHRÖDER, 1870]). Ostrowski proved it if $E = \mathbb{R}^n$ (cf. [OSTROWSKI, 1960; Theorem 22.1]). In [KITCHEN, 1966] it is proved if E is an arbitrary Banach space.

We end this section with a simple result with regard to quadratically convergent iterative processes.

<u>THEOREM 2.5.4</u>. Suppose [M,F] is quadratically convergent. Then with G = M(F) it follows that $x^* \in interior(D(G))$, $G'(x^*)$ exists and $G'(x^*) = 0$.

<u>PROOF</u>. If [M,F] is quadratically convergent then positive constants σ and δ exist such that $B(x^*, \sigma) \in D(G)$ and

 $\|\mathbf{G}(\mathbf{x}) - \mathbf{x}^{\star}\| \leq \delta \|\mathbf{x} - \mathbf{x}^{\star}\|^{2} \qquad (\text{for all } \mathbf{x} \in \mathbf{B}(\mathbf{x}^{\star}, \sigma)).$

Hence $G'(x^*)$ exists and $G'(x^*) = 0$.

2.6. THE IMBEDDING AND THE DIFFERENTIAL EQUATION

In this section we present the imbedding and the differential equation which we use to construct iterative methods by means of iterative imbedding (cf. section 1.1).

We first define some classes of operators that we shall need subsequently. Let

 $(2.6.1) \qquad F_1 = \{F \mid F \in F; D(F) \text{ is open}; F \text{ is twice continuously differentiable} \\ \text{ on } D(F); \text{ there exists a constant } \mu \text{ such that } \|F^{"}(x)\| \leq \mu \\ (\text{for all } x \in D(F)); F^{'}(x^{*}) \text{ is invertible} \}$

(see also (2.3.2)). We shall restrict our attention to problems F(x) = 0 where $F \in F_1$. In order to describe the type of imbedding to be used, we define the following two classes of operators. Let

$$\begin{array}{ll} (2.6.2) & \ensuremath{\textit{K}} = \ensuremath{\left\{ \ensuremath{\textit{K}} \mid \ensuremath{\textit{K}} : \ensuremath{\textit{D}}(\ensuremath{\textit{K}}) \ensuremath{\neq} \ensuremath{\textit{K}} : \ensuremath{\textit{D}}(\ensuremath{\textit{K}}) \ensuremath{\neq} \ensuremath{\textit{M}}(\ensuremath{\textit{K}}) \ensuremath{;} \ensuremath{\stackrel{\partial}{\scriptstyle}}_{11} \ensuremath{\textit{K}}(\ensuremath{\textit{Y}},\ensuremath{\textit{x}}) \ensuremath{;} \ensuremath{\stackrel{\partial}{\scriptstyle}}_{11} \ensuremath{\textit{K}}(\ensuremath{\textit{Y}},\ensuremath{\textit{x}}) \ensuremath{;} \ensuremath{\stackrel{\partial}{\scriptstyle}}_{11} \ensuremath{\textit{K}}(\ensuremath{\textit{Y}},\ensuremath{\textit{x}}) \ensuremath{;} \ensuremath{\stackrel{\partial}{\scriptstyle}}_{11} \ensuremath{\textit{K}}(\ensuremath{\textit{Y}},\ensuremath{\textit{x}}) \ensuremath{\; \otimes} \ensuremath{\textit{M}}(\ensuremath{\textit{K}}) \ensuremath{;} \ensuremath{;} \ensuremath{\stackrel{\partial}{\scriptstyle}}_{11} \ensuremath{\textit{K}}(\ensuremath{\textit{Y}},\ensuremath{\textit{x}}) \ensuremath{\; \otimes} \ensuremath{\; \otimes} \ensuremath{}_{11} \ensuremath{\textit{K}}(\ensuremath{\textit{Y}},\ensuremath{\textit{x}}) \ensuremath{\; \otimes} \ensuremath{$$

Let

 $(2.6.3) \qquad A = \{A \mid A: F_1 \to K; \text{ with } K = A(F) \text{ it holds } W(K) = D(F) \text{ (for all } F \in F_1) \}.$

Examples of elements of A. Let $F \in F_1$.

 $\begin{bmatrix} A(F) \end{bmatrix} (y, x) \equiv F(y) - F(x) .$ $\begin{bmatrix} A(F) \end{bmatrix} (y, x) \equiv F'(x) (y-x) .$ $\begin{bmatrix} A(F) \end{bmatrix} (y, x) \equiv y-x .$

We give three lemmata. The first lemma is a general result that will often be used subsequently.

<u>LEMMA 2.6.1</u>. Let P: $D(P) \rightarrow Z_2$ with $D(P) \subset Z_1$ where Z_1 and Z_2 are Banach spaces. Assume that a positive number δ and $V \subset D(P)$ with V open and convex exist such that for all $x \in V$ the derivative P'(x) exists and $||P'(x)|| \leq \delta$. Then

 $\|P(\mathbf{x}) - P(\mathbf{y})\| \le \delta \|\mathbf{x} - \mathbf{y}\| \quad (\text{for all } \mathbf{x}, \mathbf{y} \in \mathbf{V}).$

PROOF. See [KANTOROWITSCH & AKILOW, 1964; section 1.3(XVII)].

LEMMA 2.6.2. Let Q: $D(Q) \rightarrow Z$ with $D(Q) = V \times V$ where V is an open subset of E and Z is a Banach space. Suppose $\partial_1 Q(y,x)$ and $\partial_2 Q(y,x)$ exist (for all $x, y \in V$) and suppose that $\partial_1 Q$ is continuous on D(Q). Then Q is continuous on D(Q).

<u>PROOF</u>. Let $x_0, y_0 \in V$. Let $\varepsilon > 0$. Then a number $\delta > 0$ exists such that $x, y \in V$, $\|\partial_1 Q(y, x) - \partial_1 Q(y_0, x_0)\| \le \varepsilon$ and

$$\| Q(Y_0, \mathbf{x}) - Q(Y_0, \mathbf{x}_0) - \partial_2 Q(Y_0, \mathbf{x}_0) (\mathbf{x} - \mathbf{x}_0) \| \le \varepsilon \| \mathbf{x} - \mathbf{x}_0 \|$$

whenever $x \in B(x_0, \delta)$ and $y \in B(y_0, \delta)$. Let $x \in B(x_0, \delta)$ and $y \in B(y_0, \delta)$. Let

$$P(z) = Q(z, x) - \partial_1 Q(y_0, x_0) (z - y_0) \qquad (z \in D(P))$$

with $D(P) = B(y_0, \delta)$. From Lemma 2.6.1 it follows that

$$\| \varrho(\mathbf{y}, \mathbf{x}) - \varrho(\mathbf{y}_0, \mathbf{x}) - \vartheta_1 \varrho(\mathbf{y}_0, \mathbf{x}_0) (\mathbf{y} - \mathbf{y}_0) \| \leq \varepsilon \| \mathbf{y} - \mathbf{y}_0 \|.$$

Therefore

$$\|\varrho(y,x) - \varrho(y_0,x_0)\| \le \{\|\partial_1 \varrho(y_0,x_0)\| + \varepsilon\}\|y-y_0\| + \{\|\partial_2 \varrho(y_0,x_0)\| + \varepsilon\}\|x-x_0\| + \|\partial_2 \varrho(y_0,x_0)\| + \varepsilon\}\|x-x_0\|$$

This proves the lemma.

The next lemma is a consequence of Lemma 2.6.2.

LEMMA 2.6.3. If $K \in K$ then both the operators $\vartheta_1 K$ and K are continuous on D(K) .

PROOF. From (2.6.2) it follows that Lemma 2.6.2 applies for $Q = \partial_1 K$. Hence $\partial_1 K$ is continuous on D(K). Consequently, Lemma 2.6.2 also applies for Q = K. This completes the proof.

Let F ϵ F and A ϵ A. Set K = A(F). Let $\mathbf{x}_0 \in \mathrm{D}(\mathrm{F})$. Consider the imbedding

H:
$$[0,1] \times D(F) \rightarrow E$$
,

(2.6.4)

$$H(t,x) = (1-t)K(x,x_0) + tF(x)$$
 (t $\in [0,1], x \in D(F)$).

Thus $H(0, x_0) = 0$ and $H(1, x) \equiv F(x)$.

Suppose H(t,x) = 0 has a unique solution x = X(t) for all $t \in [0,1]$. Then

(2.6.5)
$$H(t,X(t)) = 0$$
 (for all $t \in [0,1]$).

We have

(2.6.6)
$$X(0) = x_0$$
 and $X(1) = x^*$.

We also note that for all t \in [0,1] and x \in D(F) the derivatives $\partial_1 H(t,x)$ and $\partial_2 H(t,x)$ exist and are continuous (in t and x) and satisfy

and

$$\partial_1^{H}(t,x) = -K(x,x_0) + F(x)$$

 $\partial_2^{H}(t,x) = (1-t)\partial_1^{K}(x,x_0) + tF'(x).$

The derivatives $\partial_1 H(0,x)$ and $\partial_1 H(1,x)$ should be considered respectively as the right and left partial derivatives of H with respect to t. Let

(2.6.7)
$$D(A,F) = \{(t,y,z) \mid t \in [0,1]; y,z \in D(F); with K = A(F), the operator [(1-t)]_1K(y,z) + tF'(y)] is invertible \}.$$

Suppose $(t, X(t), x_0) \in D(A, F)$ for all $t \in [0, 1]$. Then $\dot{X}(t)$ exists (see Lemma 2.6.3 and [KANTOROWITSCH & AKILOW, 1964; Theorem 3(XVII)]). Differentiation of (2.6.5) with respect to t yields

$$(2.6.8) \qquad \partial_1 H(t,X(t)) + \partial_2 H(t,X(t))\dot{X}(t) = 0 \qquad (\text{for all } t \in [0,1]).$$

Before we give the differential equation from which we shall derive the iterative methods, we introduce another class of functions, with which we can significantly enlarge the number of iterative methods to be

constructed in chapter 4 (see also section 6.3). We set

 $(2.6.9) \qquad S = \{h \mid h: [0,1] \rightarrow \mathbb{R}; h \text{ is continuous on } [0,1)\}.$ Let $g \in S$. It is obvious (see (2.6.5) and (2.6.8)) that X satisfies $(2.6.10) \qquad \partial_1 H(t,X(t)) + \partial_2 H(t,X(t))X(t) + g(t)H(t,X(t)) = 0$

(for all $t \in [0,1]$).

Since by assumption $(t, X(t), x_0) \in D(A, F)$, it follows that

$$\dot{\mathbf{x}}(t) = -[\partial_{2}H(t,\mathbf{x}(t))]^{-1}\{\partial_{1}H(t,\mathbf{x}(t)) + g(t)H(t,\mathbf{x}(t))\}$$
(for all $t \in [0,1]$),

(2.6.11)

$$x(0) = x_0$$
.

The relation (2.6.11) is equivalent to

 $\dot{x}(t) = \Phi(t, X(t), x_0) \qquad (t \in [0, 1]),$ (2.6.12a) $x(0) = x_0$

where

$$\begin{aligned} \varphi \colon D(\Phi) &\to E, \ D(\Phi) = D(A,F) \quad (cf. (2.6.7)), \\ (2.6.12b) \\ \Phi(t,y,z) &= -[(1-t)\partial_1 K(y,z) + tF'(y)]^{-1} \times \\ &\{-K(y,z) + F(y) + g(t)[(1-t)K(y,z) + tF(y)]\} \\ &((t,y,z) \in D(\Phi)). \end{aligned}$$

Conversely, if Y is a solution of

 $\dot{Y}(t) = \Phi(t, Y(t), x_0) \qquad (t \in [0, 1]),$ (2.6.13) $Y(0) = x_0,$

then

$$\frac{d}{dt} H(t, Y(t)) + g(t) H(t, Y(t)) = 0 \quad (for all t \in [0, 1]),$$

$$H(0, Y(0)) = 0.$$

So that (cf. [BERGER, 1977; Theorem 3.1.23])

$$H(t,Y(t)) = 0 \quad (for all t \in [0,1)).$$

Since K ϵ K and F ϵ F₁, and since Y is left-continuous in t = 1, it follows that

$$H(1,Y(1)) = \lim_{t \neq 1} H(t,Y(t)) = 0.$$

We summarize the above results in a theorem.

<u>THEOREM 2.6.4</u>. Assume $F \in F_1$, $A \in A$ and $g \in S$. Set K = A(F). Let $x_0 \in D(F)$. Let H be defined in (2.6.4).

- (i) If H(t,x) = 0 has a unique solution x = X(t) and $[(1-t)\partial_1 K(X(t),x_0) + tF'(X(t))]$ is invertible for all $t \in [0,1]$, then (2.6.12) holds.
- (ii) If X is a solution of the initial value problem (2.6.12a), then (2.6.5) holds.

The initial value problem (2.6.12a) will be used in chapter 4 to construct iterative methods by means of iterative imbedding.

CHAPTER 3

RUNGE-KUTTA METHODS

In this chapter we present the Runge-Kutta methods that will be used for the construction of iterative methods by means of iterative imbedding. We shall deal with two types of Runge-Kutta methods, which will be presented in the subsections 3.2.1 and 3.2.2, respectively.

In section 3.3 we shall indicate that the problem of nonconvergence of Newton's method is closely related to a certain type of unstable behaviour of Euler's method.

3.1. ONE-STEP METHODS

Consider the initial value problem

$$\dot{Y}(t) = f(t, Y(t))$$
 (t $\in [0, 1]$),

(3.1.1)

$$y(0) = y_0,$$

where f: $[0,1] \times V \rightarrow E$ with $V \subset E$ and $y_0 \in V$ are given. We assume that (3.1.1) has a unique solution Y. Most of the computational methods for solving (3.1.1) approximate the true solution Y of (3.1.1) on a discrete point set $\{t_0, t_1, \dots, t_{N+1}\}$ where $0 = t_0 < t_1 < \dots < t_{N+1} = 1$.

Runge-Kutta methods, which we shall use in the construction of iterative methods, are one-step methods. This means that, starting from y_0 and t_0 , approximations y_n of $Y(t_n)$ (n = 1,2,...,N+1) are obtained by

(3.1.2a)
$$y_{i+1} = y_i + h_i \Psi(t_i, y_i; h_i, f),$$

where

(3.1.2b)
$$h_i = t_{i+1} - t_i$$
 $(i = 0, 1, ..., N).$

The function ψ is characteristic for the method. We shall therefore define a Runge-Kutta method in terms of Ψ . By H we shall denote the sequence of stepsizes

$$(3.1.3) \qquad H = \{h_0, h_1, \dots, h_N\}.$$

Together with the one-step method, H determines the numerical integration procedure. Let a sequence of stepsizes $H = \{h_0, h_1, \dots, h_N\}$ be given. Then the numbers t_1, t_2, \dots, t_{N+1} are supposed to satisfy (3.1.2b) where $t_0 = 0$.

In the next section we describe two types of Runge-Kutta methods for solving numerically problem (3.1.1). We shall confine our considerations to so-called explicit Runge-Kutta methods and generalized Runge-Kutta methods (see e.g. [VAN DER HOUWEN, 1977; sections 2.2-3]).

3.2. DESCRIPTION OF THE RUNGE-KUTTA METHODS

3.2.1. Runge-Kutta methods with scalar coefficients

<u>DEFINITION 3.2.1</u>. Let $L = (\lambda_{i,j})$ be a real strictly lower triangular (m+1) × (m+1) matrix (m $\in \mathbb{N}$). Then the general m-stage Runge-Kutta method with scalar coefficients is defined by

$$(3.2.1a) \qquad \Psi = \sum_{\ell=1}^{m} \lambda_{m+1,\ell} k_{\ell}$$

where

$$k_1(t,y;h,f) \equiv f(t,y),$$

(3.2.1b)

$$k_{\ell}(t,y;h,f) \equiv f(t + v_{\ell}h,y + h\sum_{j=1}^{\ell-1} \lambda_{\ell,j}k_j(t,y;h,f)) \quad (\ell = 2,3,\ldots,m)$$

and

(3.2.1c)
$$\nu_{\ell} = \sum_{j=1}^{\ell-1} \lambda_{\ell,j}$$
 $(\ell = 1, 2, ..., m+1).$

The matrix L is called the *generating matrix* of the Runge-Kutta method, which, obviously, completely determines the method.

For the sake of shortness we shall use the phrase "Runge-Kutta method
with scalar coefficients L" to mean "Runge-Kutta method with scalar coefficients with generating matrix L". Moreover, given a Runge-Kutta method with scalar coefficients L = $(\lambda_{i,j})$, we shall always assume that ν_{ℓ} ($\ell = 1, 2, ..., m+1$) satisfies (3.2.1c). In addition, we shall always assume that

$$(3.2.2a)$$
 $v_{m+1} = 1$

and

(3.2.2b)
$$v_{\rho} \in [0,1]$$
 $(\ell = 2,3,\ldots,m)$.

The Runge-Kutta methods with scalar coefficients defined here are of the so-called explicit type. For a more detailed description of Runge-Kutta methods see [VAN DER HOUWEN, 1977], [LAMBERT, 1973] and [STETTER, 1973].

3.2.2. Runge-Kutta methods with operator coefficients

We assume that $\partial_2 f(t,y)$ exists (for all $t \in [0,1]$ and $y \in interior(V)$). Let $J(t,y) \equiv \partial_2 f(t,y)$.

DEFINITION 3.2.2. Let

(3.2.3a)

$$R(z) = (\rho_{i,j}(z)) \qquad (z \in \mathbb{C}).$$

Here R(z) is a strictly lower triangular (m+1) × (m+1)-matrix and ρ is a rational function with real coefficients for which

(3.2.3b)
$$0 \in D_{\mathbb{C}}(\rho_{i,j})$$
 (i = 2,3,...,m+1 and j = 1,2,...,i-1)

Then the general m-stage Runge-Kutta method with operator coefficients is defined by

(3.2.4a)
$$\Psi = \sum_{\ell=1}^{m} \Lambda_{m+1,\ell} k_{\ell}^{k}$$

where

$$k_{1}(t,y;h,f) \equiv f(t,y),$$

(3.2.4b)

$$k_{\ell}(t,y;h,f) \equiv f(t+\nu_{\ell}h,y+h\sum_{j=1}^{\ell-1}\Lambda_{\ell,j}k_{j}(t,y;h,f)) \quad (\ell = 2,3,\ldots,m),$$

$$(3.2.4c) \qquad \Lambda_{i,j} = \rho_{i,j}(hJ(t,y)) \qquad (i = 2,3,...,m+1 \text{ and } j = 1,2,...,i-1)$$

and

(3.2.4d)
$$v_{\ell} = \sum_{j=1}^{\ell-1} \rho_{\ell,j}(0)$$
 $(\ell = 1, 2, ..., m+1).$

The operator R completely determines the method. For the sake of shortness we shall use the phrase "Runge-Kutta method with operator coefficients R" to mean "Runge-Kutta method with operator coefficients defined in (3.2.4) where R is of the type (3.2.3)". Analogous to the case of scalar coefficients, for a given Runge-Kutta method with operator coefficients R, v_{ℓ} ($\ell = 1, 2, \ldots, m+1$) is always supposed to satisfy (3.2.4d). Furthermore, we shall always assume that

$$(3.2.5a)$$
 $v_{m+1} = 1$

and

(3.2.5b)
$$v_{\rho} \in [0,1]$$
 $(\ell = 2,3,...,m)$.

The Runge-Kutta methods with operator coefficients that are defined here, require one evaluation of the operator J per step. A more detailed description of Runge-Kutta methods with operator coefficients, including methods that require several evaluations of the operator J per step, can be found in [VAN DER HOUWEN, 1977; section 2.3].

3.3. NONCONVERGENCE OF NEWTON'S METHOD

3.3.1. Strong stability of discretization methods

Let $F \in F_1$ and $x_0 \in D(F)$ (cf. (2.6.1)). Let $K \in K$ where $K(y,x) \equiv F(y) - F(x)$ and $g \in S$ where g = 0 (cf. (2.6.2), (2.6.9)).

Suppose the initial value problem (2.6.12a) has a unique solution X. Hence

$$\dot{\mathbf{X}}(t) = -\Gamma(\mathbf{X}(t))F(\mathbf{x}_{0})$$
 (t $\in [0,1]$),
 $\mathbf{X}(0) = \mathbf{x}_{0}$.

(3.3.1)

In (3.3.1) $\Gamma(X(t))$ denotes $[F'(X(t))]^{-1}$. It follows that $X(1) = x^*$ (cf. Theorem 2.6.4(ii)).

Suppose we solve (3.3.1) numerically, using a Runge-Kutta method of the type described in section 3.2. Let x_1 denote the approximation to $X(1) = x^*$ thus obtained.

If $x_0 = x^*$ it is easily verified that $x_1 = x^*$. If $x_0 = x^* + \delta$ where $\delta \in E$ with $\delta \neq 0$, then

$$\|\mathbf{x}(1) - \mathbf{x}^{*}\| < \|\mathbf{x}_{0} - \mathbf{x}^{*}\|.$$

We wish x_1 to be closer to x^* than x_0 is, i.e.

$$(3.3.2) \qquad \|\mathbf{x}_{1} - \mathbf{x}^{*}\| < \|\mathbf{x}_{0} - \mathbf{x}^{*}\|.$$

In other words, we want the Runge-Kutta method to damp out the pertubations δ , just like the true solution does. This requirement is closely related to the concept of strong stability of discretization methods for initial value problems (cf. [STETTER, 1973; section 1.5.3]).

Suppose we solve (3.3.1) numerically, using Euler's method with stepsize h = 1. We then obtain an approximation x_1 to $X(1) = x^*$ that satisfies

(3.3.3)
$$x_1 = x_0 - \Gamma(x_0)F(x_0)$$

Hence, in this case, x_1 is the first Newton iterate. Therefore, the iterative process [M,F] where M is Newton's method is of type (1.1.13) where Euler's method with stepsize h = 1 is used.

Consequently, nonconvergence of Newton's method might be conceived as a case in which Euler's method with stepsize h = 1 is not strongly stable with respect to perturbations in the vector x_0 occurring in the initial value problem (3.3.1).

3.3.2. Class of strongly stable Runge-Kutta methods

Consider the initial value problem

(3.3.4a) $\dot{Y}(t) = CY(t)$ $(t \ge 0)$,

$$(3.3.4b)$$
 $Y(0) = Y_0$

where $y_0 \in \mathbb{R}^n$, $C \in L(\mathbb{R}^n)$ and all eigenvalues μ of C satisfy $\operatorname{Re}(\mu) < 0$. This type of differential equation has the property that perturbations that are introduced at t = 0 are damped out as t grows (cf. [STETTER, 1973; Theorem 2.3.4]).

Obviously, problem (3.3.4) is a rather poor model for problems of the type (3.3.1). This last type of problem, however, turns out to be too complicated in the analysis of strong stability of Runge-Kutta methods.

$$(3.3.5) \qquad \mathbf{R}_{\theta}(z) \equiv \begin{pmatrix} 0 & 0 \\ \left[1 - (1 - \theta)z\right]^{-1} & 0 \end{pmatrix}$$

Suppose we solve the initial value problem (3.3.4a) where $Y(0) = Y_0 + \delta$ with $\delta \in \mathbb{R}^n$. It can be shown that if the numerical integration procedure uses R_{θ} with stepsize h > 0, the effect of δ is damped out in the approximations Y_n of $Y(t_n)$ if t_n grows (cf. [LAMBERT, 1973; pp. 240-241], where methods are described that, when applied to (3.3.4), are equivalent to the methods R_{θ} ($\theta \in [0, \frac{1}{2}]$)). On the other hand, if we use Euler's method with stepsize h > 0, this need not be the case (cf. [LAMBERT, 1973; p. 227]).

Such phenomena may also occur in problems of the type (3.3.4) where t ϵ [0,T] with T < ∞ (cf. [STETTER, 1973; section 2.3.7]).

Let \overline{M} denote an iterative method for which the iterative process $[\overline{M},F]$ is of the type (1.1.13) and which is based on the numerical integration of (3.3.1), using R_{θ} with $\theta \in [0, \frac{1}{2}]$. In view of the above considerations one might expect \overline{M} to be more successful than Newton's method in some cases where the latter fails (at least if $E = IR^{n}$).

In chapter 7 we shall therefore apply iterative methods of the type \bar{M} to problems for which Newton's method fails. In that chapter we shall also

consider iterative processes of the type (1.1.13) that are based on the numerical integration of (3.3.1), using the Runge-Kutta methods with scalar coefficients L_1 or L_2 . L_1 and L_2 are defined by

$$(3.3.6) L_1 = \begin{pmatrix} 0 & 0 & 0 \\ \frac{1}{8} & 0 & 0 \\ 0 & 1 & 0 \end{pmatrix} and L_2 = \begin{pmatrix} 0 & 0 & 0 & 0 & 0 \\ \frac{1}{64} & 0 & 0 & 0 & 0 \\ 0 & \frac{1}{20} & 0 & 0 & 0 \\ 0 & 0 & \frac{5}{32} & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 \end{pmatrix}$$

With respect to problem (3.3.4) these methods have a strong stability behaviour that is better than that of Euler's method when all eigenvalues μ of the linear operator C in (3.3.4) satisfy $\mu \in \mathbb{R}$ and $\mu < 0$ (cf. [VAN DER HOUWEN, 1977; pp. 89-90]).

For the sake of comparison, we shall also consider iterative processes of the type (1.1.13) that are based on the numerical integration of (3.3.1), using Euler's method with stepsizes h < 1.

31

CHAPTER 4

CONSTRUCTION OF ITERATIVE METHODS

In this chapter we derive general formulae for iterative methods which are constructed by means of iterative imbedding (cf. p. 4).

4.1. ITERATIVE METHODS BASED ON RUNGE-KUTTA METHODS WITH SCALAR COEFFICIENTS

Let A ϵ A and g ϵ S be given (cf. (2.6.3) and (2.6.9)). Let L = $(\lambda_{i,j})$ be an m-stage Runge-Kutta method with scalar coefficients and let $H = \{h_0, h_1, \dots, h_N\}$ be a sequence of stepsizes.

Let $F \in F_1$ (cf. (2.6.1)) and set K = A(F).

The iterative method to be described here will be constructed by means of iterative imbedding. More specifically, it will be based on the numerical integration of the initial value problem

$$\dot{x}(t) = -[(1-t)\partial_1 K(x(t), x_0) + tF'(x(t))]^{-1} \times$$

 $\{-K(X(t),x_0) + F(X(t)) + g(t)[(1-t)K(X(t),x_0) + tF(X(t))]\}$

 $(t \in [0,1]),$

(4.1.1)

 $x(0) = x_0$

(cf. (2.6.12)). In short,

.

 $\dot{X}(t) = f(t, X(t))$ (t $\in [0, 1]$),

(4.1.2)

 $x(0) = x_0$,

where $f(t,y) \equiv \Phi(t,y,x_0)$ (cf. (2.6.12b)). In (4.1.1) (or, equivalently,

(4.1.2)) $\mathbf{x}_0 \in D(F)$ is given. The numerical integration procedure is based on the Runge-Kutta method L and the sequence of stepsizes \mathcal{H} .

Suppose the initial value problem (4.1.1) has a unique solution X. Then according to Theorem 2.6.4(ii) it follows that $X(1) = x^*$. The Runge-Kutta approximation x_1 of $X(1) = x^*$ is given by

$$x_1 = y_{N+1}$$

where

$$y_0 = x_0$$

and

$$y_{n+1} = z_{m+1}^n$$
 (n = 0,1,...,N).

Herewith we define for n = 0, 1, ..., N the quantities z_{ℓ}^{n} ($\ell = 1, 2, ..., m+1$) by

$$\mathbf{k}_{\ell-1}^{n} = f(\mathbf{t}_{n} + \mathbf{v}_{\ell-1}\mathbf{h}_{n}, \mathbf{z}_{\ell-1}^{n})$$

(if $\ell > 1$) and

$$z_{\ell}^{n} = y_{n} + h_{n} \sum_{j=1}^{\ell-1} \lambda_{\ell,j} k_{j}^{n}.$$

If we repeat this process in the way described on page 4 we arrive at an iterative process [M,F] of type (1.1.13). M is an iterative method for F_1 , depending on A, g, L and H. For the sake of clarity we shall sometimes denote this iterative method by $\mathbb{M}(A,g,L,H)$ in order to emphasize its dependence on A, g, L and H. Hence

$$(4.1.3)$$
 M = IM (A,g,L,H).

We shall now give, for $F \in F_1$, an expression of G = M(F) in terms of K, g, L and H, where K = A(F).

G: $D(G) \rightarrow E$, (4.1.4a)

$$G = \eta_{N+1}$$

where

 $\eta_0: D(\eta_0) \rightarrow E,$

(4.1.4b) $D(\eta_0) = D(F)$,

$$\eta_0(\mathbf{x}) = \mathbf{x} \qquad (\mathbf{x} \in D(\eta_0))$$

.

and

(4.1.4c)
$$\eta_{n+1} = \zeta_{m+1}^n$$
 (n = 0,1,...,N).

Herewith we define for n = 0, 1, ..., N the functions ζ_{ℓ}^{n} ($\ell = 1, 2, ..., m+1$) by

$$\kappa_{\ell-1}^{n}: D(\kappa_{\ell-1}^{n}) \neq E,$$

$$(4.1.4d) \quad D(\kappa_{\ell-1}^{n}) = \{x \mid x \in D(\zeta_{\ell-1}^{n}); (t_{n} + v_{\ell-1}h_{n}, \zeta_{\ell-1}^{n}(x), x) \in D(\Phi)\},$$

$$\kappa_{\ell-1}^{n}(x) = \Phi(t_{n} + v_{\ell-1}h_{n}, \zeta_{\ell-1}^{n}(x), x) \quad (x \in D(\kappa_{\ell-1}^{n}))$$

(if $\ell > 1$) and

$$\begin{aligned} \zeta_{\ell}^{n}: D(\zeta_{\ell}^{n}) \rightarrow E, \\ (4.1.4e) \quad D(\zeta_{\ell}^{n}) &= \{x \mid x \in D(\eta_{n}); x \in D(\kappa_{j}^{n}) \ (j = 1, 2, \dots, \ell-1), \text{ if } \ell > 1\}, \\ \zeta_{\ell}^{n}(x) &= \eta_{n}(x) + h_{n} \sum_{j=1}^{\ell-1} \lambda_{\ell,j} \kappa_{j}^{n}(x) \qquad (x \in D(\zeta^{n})). \end{aligned}$$

4.2. ITERATIVE METHODS BASED ON RUNGE-KUTTA METHODS WITH OPERATOR COEFFICIENTS

Again, iterative methods that will be described in this section are based on the numerical integration of (4.1.1) (or (4.1.2)), where K, g and x_0 are given. However, the numerical integration procedure will now be based on a Runge-Kutta method with operator coefficients.

4.2.1. Preliminaries

When using a Runge-Kutta method with operator coefficients to solve the initial value problem (4.1.2), an expression for $\partial_2 f(t,y)$ is required.

In this subsection we give a lemma which will be used in the next subsection to derive a formula for $\partial_2 f(t,y)$.

We first give a lemma that will often be used subsequently.

LEMMA 4.2.1. Let C, $\Delta \in L(E)$. Suppose C is invertible and

$$\|\Delta\| < \frac{1}{\|[c]^{-1}\|}$$
.

Then $C + \Delta$ is invertible and

$$\|[c+\Delta]^{-1}\| \leq \frac{\|[c]^{-1}\|}{1-\|[c]^{-1}\| \cdot \|\Delta\|} .$$

PROOF. see [KANTOROWITSCH & AKILOW, 1964; Theorem 4(2.V)].

Let P: U \rightarrow E and Q: U \rightarrow L(E) where U is an open subset of E. If the derivative of Q at x \in U exists, we have Q'(x) \in $L^{(2)}(E)$, (see section 2.1).

LEMMA 4.2.2. Assume that Q(x) is invertible for all x ε U. Define

W: U \rightarrow E, W(x) = $[Q(x)]^{-1}P(x)$ (x \in U).

Let $y \, \in \, \texttt{U}.$ Assume that Q'(y) and P'(y) exist. Then W'(y) exists and

$$W'(y) z = \left[Q(y)\right]^{-1} P'(y) z - \left[Q(y)\right]^{-1} Q'(y) z \left[Q(y)\right]^{-1} P(y)$$
(for all $z \in E$).

<u>PROOF.</u> 1. Let I(E) denote the subset of l(E) that consists of all invertible linear operators on E. From Lemma 4.2.1 it follows that this subset is open in L(E). Let

$$\begin{aligned} \mathbf{R}_1 : \ \mathcal{I}(\mathbf{E}) & \rightarrow \mathcal{L}(\mathbf{E}) , \\ \mathbf{R}_1(\mathbf{C}) &= \left[\mathbf{C}\right]^{-1} \qquad \left(\mathbf{C} \in \mathcal{I}(\mathbf{E})\right) . \end{aligned}$$

From [RALL, 1969; p. 96] it follows that for all $C \in I(E)$ the derivative $R_1^{i}(C)$ exists and $[R_1^{i}(C)](\widetilde{C}) = -[C]^{-1}\widetilde{C}[C]^{-1}$ (for all $\widetilde{C} \in L(E)$). Let

$$R_2: U \rightarrow L(E),$$

$$R_2(x) = R_1(Q(x)) \qquad (x \in U).$$

From the chainrule it follows that R_2 is differentiable at y and

(4.2.1)
$$R'_{2}(y) z = R'_{1}(Q(y))Q'(y) z = -[Q(y)]^{-1}Q'(y)z[Q(y)]^{-1}$$
(for all $z \in E$).

2. Obviously $W(x) \equiv R_2(x)P(x)$. Using the relation (4.2.1) it follows that to prove the lemma, it is sufficient to prove that W'(y) exists and

$$W'(y)z = R_2'(y)zP(y) + R_2'(y)P'(y)z$$
 (for all $z \in E$).

Choose σ > 0 such that B(y, σ) \subset U. For h ϵ E with $\|h\|$ < σ let

$$\epsilon_1(h) = R_2(y+h) - R_2(y) - R_2(y)h$$

and

$$\epsilon_2(h) = P(y+h) - P(y) - P'(y)h.$$

Then

$$W(y+h) - W(y) = R_2(y)[P'(y)h + \varepsilon_2(h)]$$

+
$$[R'_{2}(y)h + \varepsilon_{1}(h)][P(y) + P'(y)h + \varepsilon_{2}(h)].$$

Since, for i = 1, 2,

$$\frac{\|\varepsilon_{1}(h)\|}{\|h\|} \to 0 \quad \text{when} \quad h \to 0 \quad (h \neq 0),$$

this completes the proof. \Box

4.2.2. Formula for iterative methods based on Runge-Kutta methods with operator coefficients

Let A ϵ A and g ϵ S. Let R = ($\rho_{i,j}$) be an m-stage Runge-Kutta method with operator coefficients and let # = {h_0,h_1,...,h_N} be a sequence of stepsizes.

Let F ϵ F₁ and set K = A(F). Let Φ be defined by (2.6.12b).

LEMMA 4.2.3. Let $(t,y,x) \in D(\Phi)$. Then $y \in interior(D(\Phi(t,\cdot,x)))$, $x \in interior(D(\Phi(t,y,\cdot)))$ and the partial derivatives $\partial_2 \Phi(t,y,x)$ and $\partial_3 \Phi(t,y,x)$ exist. We have

 $\partial_2 \Phi(t,y,x) =$

4.2.2)
$$-g(t)I - [(1-t)\partial_1K(y,x) + tF'(y)]^{-1}$$

$$\{-\partial_1 K(\mathbf{y}, \mathbf{x}) + \mathbf{F}'(\mathbf{y}) + [(1-t)\partial_{11} K(\mathbf{y}, \mathbf{x}) + t\mathbf{F}''(\mathbf{y})]\Phi(t, \mathbf{y}, \mathbf{x})\}$$

and

(

$$\partial_{3} \Phi(t,y,x) =$$
(4.2.3)
$$[(1-t)\partial_{1}K(y,x) + tF'(y)]^{-1} \{[1-g(t)(1-t)]\partial_{2}K(y,x) - (1-t)\partial_{21}K(y,x)\Phi(t,y,x)\}.$$

<u>PROOF</u>. Since $(t,y,x) \in D(\Phi)$, the operator $[(1-t)\partial_1 K(y,x) + tF'(y)]$ is invertible. Set $\alpha = \|[(1-t)\partial_1 K(y,x) + tF'(y)]^{-1}\|$. We recall that D(F) is an open subset of E. From Lemma 2.6.3 it follows that $\partial_1 K$ is continuous at (y,x). Further F"(y) exists. Consequently, a number $\sigma > 0$ exists such that $B(x,\sigma) \in D(F)$, $B(y,\sigma) \in D(F)$ and

$$\| \left[(1-t) \partial_1 K(\overline{y}, \overline{x}) + tF'(\overline{y}) \right] - \left[(1-t) \partial_1 K(y, x) + tF'(y) \right] \| < \frac{1}{\alpha}$$

(for all $\bar{\mathbf{x}} \in B(\mathbf{x},\sigma)$ and $\bar{\mathbf{y}} \in B(\mathbf{y},\sigma)$).

From Lemma 4.2.1 it follows that for all $\overline{y} \in B(y,\sigma)$ and $\overline{x} \in B(x,\sigma)$ we have $(t,\overline{y},\overline{x}) \in D(\Phi)$ so that $B(y,\sigma) \subseteq D(\Phi(t,\cdot,x))$ and $B(x,\sigma) \subseteq D(\Phi(t,y,\cdot))$.

From [RALL, 1969; Theorem 18.1] it follows that $Az_1z_2 = Az_2z_1$ (for all z_1 and z_2 in E) with $A = [(1-t)\partial_{11}K(y,x) + tF''(y)]$. Consequently, from Lemma 4.2.2 it follows that $\partial_2 \Phi(t,y,x)$ exists and that (4.2.2) holds.

Since $\partial_{12}^{K}(y,x)$ and $\partial_{21}^{K}(y,x)$ exist, and $\partial_{12}^{K}(y,x)z_1z_2 = \partial_{21}^{K}(y,x)z_2z_1$ (for all z_1 and z_2 in E) (cf. [RALL, 1969; p. 116]), from Lemma 4.2.2 it follows that $\partial_3^{\Phi}(t,y,x)$ exists and that (4.2.3) holds.

Let $\mathbf{x}_0 \in D(F)$ be given. Set $f(t,y) \equiv \Phi(t,y,\mathbf{x}_0)$. From Lemma 4.2.3 it follows that whenever $(t,y,\mathbf{x}_0) \in D(\Phi)$, the partial derivative $\partial_2 \Phi(t,y,\mathbf{x}_0)$ exists. Thus $\partial_2 f(t,y)$ exists and $\partial_2 f(t,y) = \partial_2 \Phi(t,y,\mathbf{x}_0)$ (for all $(t,y) \in D(f)$).

Consider the initial value problem (4.1.1) and suppose it has a unique solution X. Then according to Theorem 2.6.4(ii) it follows that $X(1) = x^*$. The Runge-Kutta approximation x_1 of $X(1) = x^*$ is given by

 $x_1 = y_{N+1}$

where

$$y_0 = x_0$$

and

$$y_{n+1} = z_{m+1}^n$$
 (n = 0, 1, ..., N).

Herewith we define for n = 0, 1, ..., N the quantities z_{ℓ}^{n} ($\ell = 1, 2, ..., m+1$) by

$$k_{\ell-1}^{n} = f(t_{n} + v_{\ell-1}h_{n}, z_{\ell-1}^{n})$$

(if l > 1),

$$\mathcal{L}_{\ell,j}^{(n)} = \rho_{\ell,j}(h_n \partial_2 f(t_n, y_n)) \qquad (j = 1, \dots, \ell-1)$$

(if $\ell > 1$) and

$$z_{\ell}^{n} = y_{n} + h_{n} \sum_{j=1}^{\ell-1} \Lambda_{\ell,j}^{(n)} k_{j}^{n}$$

If we repeat this process in the way described on page 4 we arrive at an iterative process [M,F] of type (1.1.13). M is an iterative method for F_1 , depending on A, g, R and H. Analogous to (4.1.3) we shall sometimes denote this iterative method by $\mathbb{M}(A,g,R,H)$. Hence

(4.2.4) M = IM(A,g,R,H).

We recall that for a rational function with real coefficients $\boldsymbol{\rho}$, the set $D_{E}(\rho)$ is defined by (2.2.7). We shall now give, for F \in F_{1} , an expression of G = M(F) in terms of K, g, R and H, where K = A(F).

.

G: D(G)
$$\rightarrow$$
 E,

(4.2.5a)

$$G = \eta_{N+1}$$

where

$$\eta_0: D(\eta_0) → E,$$

(4.2.5b) $D(\eta_0) = D(F),$

 $\eta_0(x) = x \quad (x \in D(\eta_0))$

and

(4.2.5c)
$$\eta_{n+1} = \zeta_{m+1}^n$$
 (n = 0,1,...,N).

Herewith we define for n = 0,1,...,N the functions ζ_{ℓ}^n (ℓ = 1,2,...,m+1) by

$$\kappa_{\ell-1}^{n}: D(\kappa_{\ell-1}^{n}) \rightarrow E,$$

(4.

2.5d)
$$D(\kappa_{\ell-1}^{n}) = \{ \mathbf{x} \mid \mathbf{x} \in D(\zeta_{\ell-1}^{n}); \ (\mathbf{t}_{n} + \nu_{\ell-1}\mathbf{h}_{n}, \zeta_{\ell-1}^{n}(\mathbf{x}), \mathbf{x}) \in D(\Phi) \},$$
$$\kappa_{\ell-1}^{n}(\mathbf{x}) = \Phi(\mathbf{t}_{n} + \nu_{\ell-1}\mathbf{h}_{n}, \zeta_{\ell-1}^{n}(\mathbf{x}), \mathbf{x}) \qquad (\mathbf{x} \in D(\kappa_{\ell-1}^{n}))$$

(if $\ell > 1$) and

(4.2.5e)

$$\begin{aligned} \zeta_{\ell}^{n}: D(\zeta_{\ell}^{n}) \neq E, \\ D(\zeta_{\ell}^{n}) &= \{x \mid x \in D(\eta_{n}); x \in D(\kappa_{j}^{n}) \text{ and} \\ h_{n}\partial_{2}\Phi(t_{n},\eta_{n}(x),x) \in D_{E}(\rho_{\ell},j) (j = 1,2,\ldots,\ell-1), \text{ if } \ell > 1\}, \\ \zeta_{\ell}^{n}(x) &= \eta_{n}(x) + h_{n} \sum_{j=1}^{\ell-1} \Lambda_{\ell,j}^{(n)} \kappa_{j}^{n}(x) \qquad (x \in D(\zeta^{n})), \end{aligned}$$

where, for $\ell > 1$,

(4.2.5f)
$$\Lambda_{\ell,j}^{(n)} = \rho_{\ell,j}(h_n \partial_2 \Phi(t_n, \eta_n(x), x))$$
 (j = 1,2,..., l-1).

CHAPTER 5

LOCAL CONVERGENCE

As already noted in section 2.5, an iterative process will have a positive radius of convergence if and only if it is locally convergent.

Throughout this chapter we denote with A and g given (fixed) elements belonging to A and S respectively (see (2.6.3) and (2.6.9)). Further, F denotes a given (fixed) operator, F ϵ F₁ (see (2.6.1)) and K = A(F).

Let R be a Runge-Kutta method with operator coefficients and let H be a sequence of stepsizes. In this chapter we investigate the conditions which should be imposed on A, g, R and H for the iterative process [M,F] where M = IM(A,g,R,H) (cf. (4.2.4)) to be locally convergent. In particular we shall investigate under what conditions on A, g, R and H the iterative process [M,F] exhibits one of the three types of convergence behaviour that were introduced in the Definitions 2.5.2 - 4. The most obvious conditions are presented in subsection 5.2.2, in which R is supposed to be a Runge-Kutta method with scalar coefficients, and in subsection 5.2.3.

5.1. NECESSARY CONDITIONS AND SUFFICIENT CONDITIONS

Throughout section 5.1 we denote with $R = (\rho_{i,j})$ and $H = \{h_0, h_1, \dots, h_N\}$ respectively a given (fixed) m-stage Runge-Kutta method with operator coefficients and a given (fixed) sequence of stepsizes. In particular, R may be a Runge-Kutta method with scalar coefficients.

Let M = IM(A,g,R,H) (cf. (4.2.4)). In this section we present a theorem that gives necessary conditions and sufficient conditions for the iterative process [M,F] to have a local convergence behaviour of one of the types that were introduced in the Definitions 2.5.2-4.

To that end we introduce the following functions.

For any t \in [0,1], let

44

(5.1.1)
$$D_t = \{z \mid z \in \mathbb{C}; 1 - t + tz \neq 0\}.$$

For i = 1, 2, let

$$(5.1.2a) \qquad u_i: D(u_i) \to \mathbb{C}$$

where

(5.1.2b)
$$D(u_i) = \{(t,z) \mid z \in D_t, t \in [0,1]\}$$

and for all $(t,z) \in D(u_i)$,

(5.1.2c)
$$u_i(t,z) = \begin{cases} [1-t+tz]^{-1}[1-z] - g(t) & (if i = 1), \\ -[1-g(t)(1-t)][1-t+tz]^{-1} & (if i = 2). \end{cases}$$

. Let

$$\gamma: D(\gamma) \rightarrow \mathbb{C},$$
(5.1.3a)

 $\gamma = \tau_{N+1}$

where

$$\begin{split} \tau_0: \ \mathfrak{C} \rightarrow \mathfrak{C}, \\ (5.1.3b) \\ \tau_0(\mathbf{z}) \ \equiv \ 1 \end{split}$$

and

(5.1.3c)
$$\tau_{n+1} = \alpha_{m+1}^{n}$$
 (n = 0,1,...,N).

Herewith we define for n = 0,1,...,N the functions α_{ℓ}^n (ℓ = 1,2,...,m+1) by

$$\pi_{\ell-1}^{n}: D(\pi_{\ell-1}^{n}) \rightarrow \mathbb{C},$$

(5.1.3d)
$$D(\pi_{\ell-1}^{n}) = D(\alpha_{\ell-1}^{n}) \cap D_{t_{n}+\nu_{\ell-1}h_{n}},$$
$$\pi_{\ell-1}^{n}(z) = u_{1}(t_{n}+\nu_{\ell-1}h_{n},z)\alpha_{\ell-1}^{n}(z) + u_{2}(t_{n}+\nu_{\ell-1}h_{n},z) \quad (z \in D(\pi_{\ell-1}^{n}))$$

(if $\ell > 1$) and

$$\alpha_{\ell}^{n}: D(\alpha_{\ell}^{n}) \rightarrow \mathbb{C}$$

(5.1.3e)

$$D(\alpha_{\ell}^{n}) = \{z \mid z \in D(\tau_{n}); z \in D(\pi_{j}^{n}) \text{ and} \\ h_{n}u_{1}(t_{n},z) \in D_{\mathbb{C}}(\rho_{\ell,j}) (j = 1,2,...,\ell-1), \text{ if } \ell > 1\}, \\ \alpha_{\ell}^{n}(z) = \tau_{n}(z) + h_{n}\sum_{j=1}^{\ell-1} \rho_{\ell,j}(h_{n}u_{1}(t_{n},z))\pi_{j}^{n}(z) \qquad (z \in D(\alpha_{\ell}^{n})).$$

Consider the following conditions (we note that sr(T) with T $\in L(E)$ is defined by (2.2.2); the set $D_E(\gamma)$ is defined by (2.2.7)).

CONDITION 0.
$$\partial_1 K(x^*, x^*)$$
 is invertible and $C \in D_E(\gamma)$ where $C = [\partial_1 K(x^*, x^*)]^{-1} F'(x^*)$.

CONDITION 1. Condition 0 holds and $sr(\gamma(C)) < 1$.

CONDITION 2. Condition 0 holds and $sr(\gamma(C)) \leq 1$.

CONDITION 3. Condition 0 holds and $\|\gamma(C)\| < 1$.

CONDITION 4. Condition 0 holds and $\|\gamma(C)\| \leq 1$.

CONDITION 5. Condition 0 holds and $\gamma(C) = 0$.

Then the following theorem holds.

THEOREM 5.1.1. The following propositions (i) - (v) are true.

- (i) Condition 1 is a sufficient condition for the iterative process [M,F] to be stably convergent.
- (ii) Condition 2 is a <u>necessary</u> condition for the iterative process [M,F] to be <u>stably</u> convergent.
- (iii) Condition 3 is a sufficient condition for the iterative process [M,F] to be regularly convergent.
- (iv) Condition 4 is a <u>necessary</u> condition for the iterative process [M,F] to be <u>regularly</u> convergent.
- (v) Condition 5 is a <u>necessary</u> and <u>sufficient</u> condition for the iterative process [M,F] to be <u>quadratically</u> convergent.

The proof of this theorem will be given in section 5.3.

In many cases it might be very complicated to verify whether or not one of the Conditions 1-5 is satisfied. In the next section we shall give,

amongst other things, less complicated stipulations under which one of the Conditions 1-5 is satisfied.

5.2. FURTHER CONDITIONS ON THE ITERATIVE METHODS

This section is divided into three subsections. The first contains two theorems that give sufficient conditions under which the iterative process [M,F] where $M = \mathbb{M}(A,g,R,H)$ is stably convergent and quadratically convergent, respectively. Here R is a given Runge-Kutta method with operator coefficients and H is a given sequence of stepsizes. These conditions are simpler than those given in Theorem 5.1.1. The last two subsections give still simpler conditions under which [M,F] exhibits one of the types of convergence behaviour that were considered in section 5.1.

5.2.1. Simpler sufficient conditions

for $n = 0, 1, ..., N_{i}$

Throughout subsection 5.2.1 we denote with $R = (\rho_{i,j})$ and $H = \{h_0, h_1, \ldots, h_N\}$ respectively a given (fixed) m-stage Runge-Kutta method with operator coefficients and a given (fixed) sequence of stepsizes. Let M = IM(A,g, R, H). Consider the following condition:

(5.2.1)

$$-h_n g(t_n) \in D_{\mathbb{C}}(\rho_{\ell,j})$$
 and $\rho_{\ell,j}(-h_n g(t_n)) = \rho_{\ell,j}(0)$
 $(\ell = 2, 3, \dots, m+1; j = 1, 2, \dots, \ell-1)$

The following two theorems hold.

<u>THEOREM 5.2.1</u>. Assume that (5.2.1) holds and that $sp([F'(x^*]^{-1}\partial_1 K(x^*,x^*)) = \{1\}$. Then the iterative process [M,F] is stably convergent.

<u>THEOREM 5.2.2</u>. Assume that (5.2.1) holds and that $\partial_1 K(x^*, x^*) = F'(x^*)$. Then the iterative process [M,F] is quadratically convergent.

We shall prove these theorems in section 5.3.

<u>REMARK 5.2.1</u>. If E = IR then, obviously, the assumptions of Theorem 5.2.1 and Theorem 5.2.2 are equivalent.

Suppose E \neq IR, then this need not be the case, as the following example shows.

Choose $u_1, u_2 \in E$ such that $(u_1, u_2) = 0$ and $||u_1|| = ||u_2|| = 1$. Define $T \in L(E)$ by

$$\mathbf{T}\mathbf{x} = (\mathbf{x}, \mathbf{u}_1) \mathbf{u}_2 \qquad (\mathbf{x} \in \mathbf{E}).$$

Thus, $T^2 = 0$ so that $sp(T) = \{0\}$. Suppose that, for the operator $A \in A$,

$$[A(\widetilde{F})](y,x) \equiv \widetilde{F}(y) - \widetilde{F}(x) + T(y-x) \qquad (\widetilde{F} \in F_1).$$

Suppose that $F(x) \equiv x$. Hence $F'(x^*) = I$ and $\partial_1 K(x^*, x^*) = I + T$. Consequently, $\partial_1 K(x^*, x^*) \neq F'(x^*)$. On the other hand, from Theorem 2.2.2 we have $sp([F'(x^*)]^{-1}\partial_1 K(x^*, x^*)) = sp(I+T) = \{1\}$.

<u>REMARK 5.2.2</u>. $\partial_1 K(x^*, x^*) = F'(x^*)$ if, for example

$$[A(F)](y,x) \equiv F(y) - F(x),$$
 or
 $[A(F)](y,x) \equiv F'(x)(y-x).$

The imbedding (1.1.5) with K = A(F) where

$$[A(F)](y,x) \equiv y - x$$

is sometimes used in the discrete imbedding method (see e.g. [MENZEL & SCHWETLICK, 1976]). Since in general $\partial_1 K(x^*, x^*) \neq F'(x^*)$, this choice of A does not seem to be suitable for iterative imbedding (see also the Theorems 5.2.3-8).

<u>REMARK 5.2.3</u>. The condition (5.2.1) holds if R = L where L is a Runge-Kutta method with scalar coefficients. This case will be considered in the next subsection.

If R is a Runge-Kutta method with operator coefficients, the condition (5.2.1) holds if $g(t_n) = 0$ (n = 0,1,...,N). This case will be considered in subsection 5.2.3.

5.2.2. Iterative methods based on Runge-Kutta methods with scalar coefficients

We recall that throughout this subsection A, g and F are given (fixed) elements belonging to A, S and F_1 respectively.

The iterative methods to be considered in this subsection are of the type described in section 4.1. We shall give three theorems that give necessary and sufficient conditions on A in order that for any Runge-Kutta method with scalar coefficients L and any sequence of stepsizes H, the iterative process [M,F] where M = IM(A,g,L,H) may have one of the three types of convergence behaviour with which Theorem 5.1.1 was concerned.

The proofs of these theorems will be given in section 5.3.

THEOREM 5.2.3. The following propositions (i) and (ii) are equivalent.

- (i) For any Runge-Kutta method with scalar coefficients L and any sequence of stepsizes H the iterative process [M,F] where M = IM (A,g,L,H) is stably convergent.
- (ii) $sp([F'(x^*)]^{-1}\partial_1K(x^*,x^*)) = \{1\}.$

This theorem is a consequence of Theorem 5.2.1 and Theorem 5.1.1(ii). The next two theorems are consequent on the last three propositions of Theorem 5.1.1 and of Theorem 5.2.2.

THEOREM 5.2.4. The following propositions (iii) and (iv) are equivalent.

- (iii) For any Runge-Kutta method with scalar coefficients L and any sequence of stepsizes H the iterative process [M,F] where $M = \mathbb{M}(A,g,L,H)$ is <u>regularly</u> convergent.
- (iv) $\partial_1 K(x^*, x^*) = F'(x^*)$.

THEOREM 5.2.5. The following propositions (v) and (vi) are equivalent.

- (v) For any Runge-Kutta method with scalar coefficients L and any sequence of stepsizes H the iterative process [M,F] where M = M(A,g,L,H) is <u>quadratically</u> convergent.
- (vi) $\partial_1 K(x^*, x^*) = F'(x^*)$.

An immediate consequence of these theorems is, of course, that the propositions (iii) and (v) are equivalent.

5.2.3. Iterative methods based on Runge-Kutta methods with operator coefficients

We recall that throughout this subsection A, g and F are given (fixed) elements belonging to A, S and F_1 respectively.

In this subsection we consider iterative methods of the type described in section 4.2. Analogous to the previous subsection we shall give three theorems that give necessary and sufficient conditions on A and g in order that, for any Runge-Kutta method with operator coefficients R and any sequence of stepsizes H, the iterative process [M,F] where $M = \mathbb{M}(A,g,R,H)$ may have one of the three types of convergence behaviour with which Theorem 5.1.1 was concerned.

The proofs of these theorems will be given in the next section.

THEOREM 5.2.6. The following propositions (i) and (ii) are equivalent.

- (i) For any Runge-Kutta method with operator coefficients R and any sequence of stepsizes H the iterative process [M,F] where M = IM(A,g,R,H) is stably convergent.
- (ii) $\operatorname{sp}([F'(x^*)]^{-1}\partial_1 K(x^*,x^*)) = \{1\}$ and g(t) = 0 (for all $t \in [0,1)$).

Theorem 5.2.6 is similar to Theorem 5.2.3. As may be expected we can also obtain results similar to Theorem 5.2.4 and Theorem 5.2.5.

THEOREM 5.2.7. The following propositions (iii) and (iv) are equivalent.

- (iii) For any Runge-Kutta method with operator coefficients R and any sequence of stepsizes H the iterative process [M,F] where M = IM(A,g,R,H) is <u>regularly</u> convergent.
- (iv) $\partial_1 K(x^*, x^*) = F'(x^*)$ and g(t) = 0 (for all $t \in [0, 1)$).

THEOREM 5.2.8. The following propositions (v) and (vi) are equivalent.

- (v) For any Runge-Kutta method with operator coefficients R and any sequence of stepsizes H the iterative process [M,F] where M = IM(A,g,R,H) is <u>quadratically</u> convergent.
- (vi) $\partial_1 K(x^*, x^*) = F'(x^*)$ and g(t) = 0 (for all $t \in [0, 1)$).

An immediate consequence of these theorems is that the propositions (iii) and (v) are equivalent.

5.3. PROOF OF THE THEOREMS

As may be expected, the proofs of the theorems listed above exhibit similarities. Therefore in the next subsection we have collected some lemmata that will be useful for proving the theorems of both section 5.1 and section 5.2. Lemma 5.3.8 (statement (5.3.13d)), Lemma 5.3.10 (statement (5.3.14d)) and the Lemmata 5.3.11 and 5.3.12 will play an important role in the proofs of the theorems.

5.3.1. Preliminary lemmata

Throughout subsection 5.3.1 we denote with $R = (\rho_{i,j})$ and $H = \{h_0, h_1, \ldots, h_N\}$ respectively a given (fixed) m-stage Runge-Kutta method with operator coefficients and a given (fixed) sequence of stepsizes.

Furthermore, G = M(F) where M = IM (A,g,R,H). Since F ϵ F_1 and K = A(F) where A ϵ A, there exist positive constants σ^* , β^* , μ , μ_1 , μ_2 and μ_3 such that B(\mathbf{x}^*, σ^*) \subset D(F), and such that

- (5.3.1a) $\|[\mathbf{F}'(\mathbf{x}^*)]^{-1}\| \leq \beta^*$,
- (5.3.1b) $||F''(x)|| \le \mu$,

$$(5.3.1c) \qquad \|\partial_{11}K(\mathbf{y},\mathbf{x})\| \leq \mu_1, \ \|\partial_{12}K(\mathbf{y},\mathbf{x})\| \leq \mu_2 \text{ and } \|\partial_{21}K(\mathbf{y},\mathbf{x})\| \leq \mu_3,$$

(5.3.1d) K(x,x) = 0

for all x,y $\in B(x^*,\sigma^*)$ (see (2.6.1), (2.6.2) and (2.6.3)). As a consequence of Lemma 2.6.1 and (5.3.1) we have for all x,y,z $\in B(x^*,\sigma^*)$

$$(5.3.2a) \quad \|F'(x) - F'(y)\| \le \mu \|x - y\|,$$

(5.3.2b)
$$\|\partial_{1}K(y,x) - \partial_{1}K(z,x)\| \leq \mu_{1}\|y-z\|$$

(5.3.2c)
$$\|\partial_1 K(y,x) - \partial_1 K(y,z)\| \le \mu_0 \|x-z\|$$
,

(5.3.2d)
$$\|\partial_{2}K(y,x) - \partial_{2}K(z,x)\| \leq \mu_{2}\|y-z\|$$
.

We note that

$$\partial_{1} K(y, x) - \partial_{1} K(x^{*}, x^{*})$$

= $\partial_{1} K(y, x) - \partial_{1} K(y, x^{*}) + \partial_{1} K(y, x^{*}) - \partial_{1} K(x^{*}, x^{*}).$

Hence, for all x,y $\in B(x^*,\sigma^*)$ we have

$$(5.3.3a) \qquad \|\partial_1 K(y,x) - \partial_1 K(x^*,x^*)\| \le \mu_1 \|y - x^*\| + \mu_2 \|x - x^*\|$$

and

$$(5.3.3b) \quad \|[(1-t)\partial_1 K(y,x) + tF'(y)] - [(1-t)\partial_1 K(x^*,x^*) + tF'(x^*)]\|$$

$$\leq (1-t)\{\mu_1 \|y - x^*\| + \mu_2 \|x - x^*\|\} + t\mu \|y - x^*\| \quad (\text{for all } t \in [0,1]).$$

The following lemma is a general result that will often be used subsequently.

LEMMA 5.3.1. Let P: D(P) \rightarrow E with D(P) \subset E. Let x, y \in D(P). Suppose V = $\{z \mid z = x+t(y-x) \text{ with } t \in [0,1]\} \subset interior(D(P)). Let P'(z) exist (for all vert and vert all vert and vert and vert all vert and vert and$ $z \in V$) and suppose a constant $\delta > 0$ exists such that $||P'(z)-P'(x)|| \le \delta ||z-x||$ (for all $z \in V$). Then

$$\| \mathbf{P}(\mathbf{y}) - \mathbf{P}(\mathbf{x}) - \mathbf{P}'(\mathbf{x}) (\mathbf{y}-\mathbf{x}) \| \le \frac{\delta}{2} \| \mathbf{x}-\mathbf{y} \|^{2}$$

PROOF. see [SPIJKER, 1972; Lemma 1].

We return to operator K.

LEMMA 5.3.2.

٠

$$\partial_2 K(\mathbf{x},\mathbf{x}) = -\partial_1 K(\mathbf{x},\mathbf{x})$$
 (for all $\mathbf{x} \in B(\mathbf{x}^*,\sigma^*)$).

<u>PROOF</u>. Let $x, y \in B(x^*, \sigma^*)$. Using (5.3.1d), (5.3.2b,c) and Lemma 5.3.1,

$$\begin{split} \|_{K(\mathbf{x},\mathbf{y})} &- \kappa(\mathbf{x},\mathbf{x}) + \partial_{1}\kappa(\mathbf{x},\mathbf{x})(\mathbf{y}-\mathbf{x}) \| \\ &\leq \|_{K(\mathbf{y},\mathbf{y})} - \kappa(\mathbf{x},\mathbf{y}) - \partial_{1}\kappa(\mathbf{x},\mathbf{y})(\mathbf{y}-\mathbf{x}) \| + \|\partial_{1}\kappa(\mathbf{x},\mathbf{y}) - \partial_{1}\kappa(\mathbf{x},\mathbf{x}) \| \|\mathbf{x}-\mathbf{y} \| \\ &\leq \frac{\mu_{1}}{2} \|_{\mathbf{x}-\mathbf{y}} \|^{2} + \mu_{2} \|_{\mathbf{x}-\mathbf{y}} \|^{2}. \end{split}$$

....

This proves the lemma. \Box

Let $x, y \in B(x^*, \sigma^*)$. Then

$$\partial_{2}^{K}(y,x) - \partial_{2}^{K}(x^{*},x^{*}) = \partial_{2}^{K}(y,x) - \partial_{2}^{K}(x^{*},x) + \partial_{2}^{K}(x^{*},x)$$
$$-\partial_{2}^{K}(x,x) + \partial_{2}^{K}(x,x) - \partial_{2}^{K}(x^{*},x^{*})$$

Therefore, as a consequence of (5.3.2d), Lemma 5.3.2 and (5.3.3a) we have

$$(5.3.4) \qquad \| \partial_2 K(y,x) - \partial_2 K(x^*,x^*) \| \le \mu_3 \| y - x^* \| + (\mu_1 + \mu_2 + \mu_3) \| x - x^* \|$$

$$(\text{for all } x, y \in B(x^*,\sigma^*)).$$

We recall that the operator Φ is defined in (2.6.12b). Each iterative method described in chapter 4 is based on the numerical integration of the initial value problem (2.6.12a). Therefore, Φ will play an important role in the proofs of the theorems. The next five lemmata are concerned with Φ .

LEMMA 5.3.3. Let t \in [0,1] and suppose that (t,x^{*},x^{*}) \in D(Φ). Assume that $\partial_1 K(x^*,x^*)$ is invertible and set

$$C = [\partial_{1}K(x^{*},x^{*})]^{-1}F'(x^{*})$$

Then the operator [(1-t)I+tC] is invertible. Further

$$\mathbf{x}^{\star} \in \operatorname{interior}(\mathbb{D}(\Phi(\mathsf{t},\cdot,\mathbf{x}^{\star})) \cap \mathbb{D}(\Phi(\mathsf{t},\mathbf{x}^{\star},\cdot)))$$

and the derivatives $\partial_i \Phi(t, x^*, x^*)$ (i = 2,3) exist and satisfy

$$\partial_2 \Phi(t, x^*, x^*) = -g(t)I + [(1-t)I + tc]^{-1}[I-c],$$

 $\partial_3 \Phi(t, x^*, x^*) = -\{1 - g(t)(1-t)\}[(1-t)I + tc]^{-1}.$

PROOF. The result follows from Lemma 4.2.3 and Lemma 5.3.2.

LEMMA 5.3.4. Let t ϵ [0,1]. If $(t, x^*, x^*) \epsilon D(\Phi)$ then positive constants β and σ exist such that $(t, y, x) \epsilon D(\Phi)$ and

(5.3.5)
$$\|[(1-t)\partial_1 K(y,x) + tF'(y)]^{-1}\| \leq \beta$$

whenever $x, y \in B(x^*, \sigma)$.

<u>PROOF</u>. Set $\beta = 2 \| [(1-t)\partial_1 K(x^*,x^*) + tF'(x^*)]^{-1} \|$. Set $\hat{\mu} = \max\{\mu_1,\mu_2,\mu\}$ and $\sigma = \min\{\sigma^*, \frac{1}{2\beta\hat{\mu}}\}$. From Lemma 4.2.1 and (5.3.3b) the result follows.

LEMMA 5.3.5. Let $t \in [0,1]$. If $(t,x^*,x^*) \in D(\Phi)$ then positive constants c_1 , c_2 and σ exist such that $(t,y,x) \in D(\Phi)$ and

(5.3.6)
$$\| \Phi(t,y,x) \| \le c_1 \| x - x^* \| + c_2 \| y - x^* \|$$

whenever $x, y \in B(x^*, \sigma)$.

<u>PROOF</u>. Set $\theta_1 = \|F'(x^*)\|$ and $\delta_1 = \theta_1 + \mu\sigma^*$. From (5.3.2a) it follows that $\|F'(z)\| \le \delta_1$ (for all $z \in B(x^*, \sigma^*)$). From Lemma 2.6.1 it now follows that

(5.3.7)
$$\|F(y)\| \leq \delta_1 \|y - x^*\|$$
 (for all $y \in B(x^*, \sigma^*)$).

Set $\theta_2 = \|\partial_1 K(\mathbf{x}^*, \mathbf{x}^*)\|$ and $\delta_2 = \theta_2 + (\mu_1 + \mu_2 + \mu_3)\sigma^*$. From (5.3.3a) it follows that

$$\|\partial_1 K(z,x)\| \leq \theta_2 + (\mu_1 + \mu_2)\sigma^* \qquad (\text{for all } z,x \in B(x^*,\sigma^*)).$$

From (5.3.4) and Lemma 5.3.2 it follows that

$$\|\partial_{2} \kappa(\mathbf{x}^{\star}, \mathbf{z})\| \leq \theta_{1} + (\mu_{1} + \mu_{2} + \mu_{3})\sigma^{\star} \quad (\text{for all } \mathbf{z} \in B(\mathbf{x}^{\star}, \sigma^{\star})).$$

Let $x, y \in B(x^*, \sigma^*)$. Then

$$K(y,x) = K(y,x) - K(x^{*},x) + K(x^{*},x) - K(x^{*},x^{*}).$$

Using Lemma 2.6.1 it follows that

(5.3.8)
$$\|K(y,x)\| \leq \delta_2 \{\|y-x^*\| + \|x-x^*\|\}$$
 (for all $x, y \in B(x^{\circ}, \sigma^{\circ})$).

From Lemma 5.3.4 it follows that numbers $\beta > 0$ and $\sigma \in (0, \sigma^*]$ exist such that for all x,y $\epsilon B(x^*, \sigma)$ we have $(t, y, x) \epsilon D(\Phi)$ and (5.3.5). Together with (5.3.7) and (5.3.8) this yields the result with $c_1 = \beta [1 + |g(t)| (1-t)] \delta_2$

and $c_2 = c_1 + \beta [1 + t | g(t) |] \delta_1$.

LEMMA 5.3.6. Let t ϵ [0,1]. Suppose $(t, x^*, x^*) \in D(\Phi)$. Then positive constants $c_{1,i}, c_{2,i}$ (i = 2,3) and σ exist such that $y \in interior(D(\Phi(t, \cdot, x)))$, $x \in interior(D(\Phi(t, y, \cdot)))$ and the derivatives $\partial_{i} \Phi(t, y, x)$ exist and satisfy

$$(5.3.9) \qquad \|\partial_{i} \phi(t,y,x) - \partial_{i} \phi(t,x^{*},x^{*})\| \leq c_{1,i} \|x-x^{*}\| + c_{2,i} \|y-x^{*}\| \quad (i = 2,3)$$

whenever $x, y \in B(x^*, \sigma)$.

<u>PROOF</u>. 1. From Lemma 5.3.4 and Lemma 5.3.5 it follows that positive constants c_1, c_2, β and a number $\sigma \in (0, \sigma^*]$ exist such that for all $x, y \in B(x^*, \sigma)$ we have $(t, y, x) \in D(\Phi)$ and (5.3.5), (5.3.6). From Lemma 4.2.3 it follows that $x \in interior(D(\Phi(t, y, \cdot))), y \in interior(D(\Phi(t, \cdot, x)))$ and the derivatives $\partial_i \Phi(t, y, x)$ (i = 2,3) exist whenever $x, y \in B(x^*, \sigma)$.

Let $\mathbf{x}, \mathbf{y} \in \mathbf{B}(\mathbf{x}^*, \sigma)$, set $\mathbf{T} = [(1-t)\partial_1 \mathbf{K}(\mathbf{y}, \mathbf{x}) + t\mathbf{F}'(\mathbf{y})]$ and $\mathbf{T}^* = [(1-t)\partial_1 \mathbf{K}(\mathbf{x}^*, \mathbf{x}^*) + t\mathbf{F}'(\mathbf{x}^*)]$.

2. From Lemma 4.2.3 it follows that

$$\partial_{2} \Phi(t,y,x) - \partial_{2} \Phi(t,x^{*},x^{*})$$

= -[T]⁻¹{- $\partial_{1} K(y,x) + F'(y) + [(1-t)\partial_{11} K(y,x) + tF''(y)]\Phi(t,y,x)}+ [T^{*}]^{-1}{-\partial_{1} K(x^{*},x^{*}) + F'(x^{*})}.$

Note that

$$[T^*]^{-1} - [T]^{-1} = [T]^{-1} \{T - T^*\} [T^*]^{-1}.$$

Using this relation we obtain

$$\begin{split} \| \partial_{2} \Phi(t, y, x) &- \partial_{2} \Phi(t, x^{*}, x^{*}) \| \\ &\leq \| [T]^{-1} \| \{ \| \partial_{1} K(y, x) - \partial_{1} K(x^{*}, x^{*}) - F'(y) + F'(x^{*}) \| \\ &+ [(1-t) \| \partial_{11} K(y, x) \| + t \| F''(y) \|] \| \Phi(t, y, x) \| \} \\ &+ \| [T]^{-1} \| \| [T^{*}]^{-1} \| \| T - T^{*} \| \{ \| \partial_{1} K(x^{*}, x^{*}) \| + \| F'(x^{*}) \| \}. \end{split}$$

Using (5.3.5), (5.3.3), (5.3.2a), (5.3.1b,c) and (5.3.6) the result follows for i = 2.

3. From Lemma 4.2.3 it follows that

$$\partial_{3} \Phi(t, y, x) - \partial_{3} \Phi(t, x^{*}, x^{*})$$

$$= (1 - g(t)(1-t)) \{ [T]^{-1} \partial_{2} K(y, x) - [T^{*}]^{-1} \partial_{2} K(x^{*}, x^{*}) \}$$

$$- (1-t) [T]^{-1} \partial_{21} K(y, x) \Phi(t, y, x) .$$

Hence

$$\begin{split} \| \partial_{3} \phi(t,y,x) &- \partial_{3} \phi(t,x^{*},x^{*}) \| \\ &\leq |1 - g(t) (1-t) | \{ \| [T]^{-1} \| \| \partial_{2} K(y,x) - \partial_{2} K(x^{*},x^{*}) \| + \\ &+ \| [T]^{-1} \| \| [T^{*}]^{-1} \| \| T - T^{*} \| \| \partial_{2} K(x^{*},x^{*}) \| \} \\ &+ (1-t) \| [T]^{-1} \| \| \partial_{21} K(y,x) \| \| \phi(t,y,x) \| . \end{split}$$

Using (5.3.5), (5.3.4), (5.3.3b), (5.3.1c) and (5.3.6) the result follows for i = 3. This completes the proof.

LEMMA 5.3.7. Let
$$t \in [0,1]$$
 and suppose $(t,x^*,x^*) \in D(\Phi)$. Let
f: D(f) $\rightarrow E$,
f(z) = $\Phi(t,y,x)$ (z = (y,x), z $\in D(f)$)

with

$$D(f) = \{(y,x) \mid x,y \in E; (t,y,x) \in D(\Phi)\} \subset E \times E.$$

Then a constant $\sigma > 0$ exists such that

$$B(\mathbf{x}^{*},\sigma) \times B(\mathbf{x}^{*},\sigma) \subset D(f), \{t\} \times B(\mathbf{x}^{*},\sigma) \times B(\mathbf{x}^{*},\sigma) \subset D(\Phi),$$

$$f'((\mathbf{y},\mathbf{x})), \partial_{2}\Phi(t,\mathbf{y},\mathbf{x}) \quad and \quad \partial_{3}\Phi(t,\mathbf{y},\mathbf{x}) \quad exist$$

and

(5.3.10)
$$f'((y,x))(h_1,h_2) = \partial_2 \Phi(t,y,x)h_1 + \partial_3 \Phi(t,y,x)h_2 \quad ((h_1,h_2) \in E \times E)$$

whenever $x, y \in B(x^*, \sigma)$.

<u>PROOF</u>. Notice that the operators K, $\partial_1 K$ and $\partial_{11} K$ are continuous on D(F) × D(F) (cf. Lemma 2.6.3 and (2.6.2,3)). Further the operators F, F' and F" are continuous on D(F) (cf. (2.6.1)). From Lemma 5.3.6 it follows that a constant $\sigma > 0$ exists such that $t \times B(x^*, \sigma) \times B(x^*, \sigma) \subset D(\Phi)$ and the derivatives $\partial_1 \Phi(t, y, x)$ (i = 2,3) exist whenever $x, y \in B(x^*, \sigma)$. Consequently $B(x^*, \sigma) \times$ $B(x^*, \sigma) \subset D(f)$. It is easily verified that the mapping $(y, x) \rightarrow \partial_2 \Phi(t, y, x)$ and the operator f are continuous on $B(x^*, \sigma) \times B(x^*, \sigma)$. The result now follows from [BROWN & PAGE, 1970; pp. 284-285 and Theorem 7.4.2].

In the next lemmata Condition 0 will play an important role. We first introduce the function G^* .

Suppose that, for $n = 0, 1, \ldots, N$,

$$\mathbf{x}^{\star} \in \text{interior}(D(\Phi(t_{n}, \cdot, \mathbf{x}^{\star}))), \text{ the derivative } \partial_{2} \Phi(t_{n}, \mathbf{x}^{\star}, \mathbf{x}^{\star})$$

(5.3.11) exists and

$$h_{n}\partial_{2}\Phi(t_{n}, \mathbf{x}^{*}, \mathbf{x}^{*}) \in D_{E}(\rho_{\ell,j}) \qquad (\ell = 2,$$

$$(\ell = 2, 3, \dots, m+1; j = 1, 2, \dots, \ell-1)$$

(cf. (2.2.7)). Let

$$(5.3.12a)$$
 $G^* = G$

where G is defined in (4.2.5a-d) in which the functions ζ_{ℓ}^n are defined by

 $\boldsymbol{\zeta}_{\boldsymbol{\ell}}^{n} \colon \, \mathsf{D}(\boldsymbol{\zeta}_{\boldsymbol{\ell}}^{n}) \, \rightarrow \, \mathsf{E}_{\boldsymbol{\ell}}$

(5.2.12b)
$$D(\zeta_{\ell}^{n}) = \{x \mid x \in D(n_{n}); x \in D(\kappa_{j}^{n}) (j = 1, 2, ..., \ell-1), \text{ if } \ell > 1\},\$$

 $\zeta_{\ell}^{n}(x) = \eta_{n}(x) + h_{n} \sum_{j=1}^{\ell-1} {n \choose \ell, j} \kappa_{j}^{n}(x) \qquad (x \in D(\zeta_{\ell}^{n}))$

 $(n = 0, 1, ..., N; \ \ell = 1, 2, ..., m+1)$. In (5.3.12b)

(5.3.12c)
$$\bigwedge_{\ell,j}^{(n)} = \rho_{\ell,j} (\bigwedge_{n \partial_2} \Phi(t_n, x^*, x^*))$$
 (n = 0,1,...,N; ℓ = 2,3,...,m+1;
 j = 1,2,..., ℓ -1).

For $n = 0, 1, \ldots, N$ we set

(5.3.12d)
$$\eta_n^* = \eta_n$$
,
(5.3.12e) $\zeta_{\ell}^n = \zeta_{\ell}^n$ and, if $\ell > 1$, $\zeta_{\ell-1}^* = \zeta_{\ell-1}^n$ ($\ell = 1, 2, ..., m+1$).

REMARK 5.3.1. Obviously, if R = L, where L is an m-stage Runge-Kutta method with scalar coefficients, then $G^* = G$ where G = M(F) with $M = \mathbb{I}M(A,g,L,H)$.

The operator G^{\star} will be a useful tool in investigating the local convergence behaviour of [M,F] where M = IM(A,g,R,H). To that end we also need an expression of $[G^*]'(x^*)$.

Let the functions γ , τ_n (n = 0,1,...,N+1), α_{ℓ}^n and, for $\ell > 1$, $\pi_{\ell-1}^n$ (n = 0,1,...,N; ℓ = 1,2,...,m+1) be defined in (5.1.3).

LEMMA 5.3.8. Let Condition 0 be satisfied. Then (5.3.11) holds. Let G be defined in (5.3.12). Then for $n = 0, 1, \dots, N$ the propositions (5.3.13a), (5.3.13b) and (5.3.13c) are true.

Positive constants
$$\sigma_n$$
 and δ_n exist such that $B(x^{,}\sigma_n) \subset D(\eta_n^{,})$,
 $C \in D_E(\tau_n)$,
(5.3.13a)
the derivative $\eta_n^{*'}(x)$ exists and $\|\eta_n^{*'}(x) - \tau_n(C)\| \leq \delta_n \|x - x^{*}\|$
whenever $x \in B(x^{*}, \sigma_n)$.

For $l = 1, 2, \dots, m+1$ positive constants $\sigma_{n,l}$, $\delta_{2,n,l}$ and if l > 1, $\delta_{1,n,l-1}$ exist such that

$$B(x^{*},\sigma_{n,\ell}) \subset D(\kappa_{\ell-1}^{*^{n}}), C \in D_{E}(\pi_{\ell-1}^{n}), the derivative [\kappa_{\ell-1}^{*^{n}}]'(x)$$

(5.3.13b)

$$\| [\kappa_{\ell-1}^{*}]'(\mathbf{x}) - \pi_{\ell-1}^{n}(\mathbf{C}) \| \leq \delta_{1,n,\ell-1} \| \mathbf{x} - \mathbf{x}^{*} \|$$

(if *l* > 1) and

$$B(x^*,\sigma_{n,\ell}) \subset D(\zeta_{\ell}^{*^n}), C \in D_E(\alpha_{\ell}^n), the derivative [\zeta_{\ell}^{*^n}]'(x) exists,$$

(5.3.13c)

$$[\zeta_{\ell}^{\star^{n}}]'(\mathbf{x}) - \alpha_{\ell}^{n}(\mathbf{C}) \| \leq \delta_{2,n,\ell} \|\mathbf{x} - \mathbf{x}^{\star}\|$$

whenever $x \in B(x^*, \sigma_{n, \ell})$. In particular, since $G^* = \eta_{N+1}^* = \zeta_{m+1}^*$ and $\gamma = \tau_{N+1} = \alpha_{m+1}^N$,

 $B(x^{*},\sigma_{N,m+1}) \subset D(G^{*}), \text{ the derivative } [G^{*}]'(x^{*}) \text{ exists and}$ (5.3.13d) $\|[G^{*}]'(x) - G^{*}]\|_{L^{2}(G^{*})} = \|[G^{*}]|^{2} \|G^{*}\|_{L^{2}(G^{*})} = \|[G^{*}\|_{L^{2}(G^{*})} = \|[G^{*}\|_{L^$

$$\| [G^{\circ}]'(\mathbf{x}) - \gamma(C) \| \leq \delta_{2,N,m+1} \| \mathbf{x} - \mathbf{x}^{\circ} \|$$

whenever $x \in B(x^*, \sigma_{N, m+1})$.

<u>PROOF</u>. Since Condition 0 is satisfied, (5.3.11) holds (cf. Lemma 5.3.3, (5.1.2) and (5.1.3e)).

1. Let n = 0. Since $\eta_0^*(x) = x$ (for all $x \in D(F)$) and $\tau_0(z) \equiv 1$, the proposition (5.3.13a) is true for n = 0.

2. Suppose (5.3.13a) is true for some $n = n_0$ with $0 \le n_0 \le N$. 2.1. Since $\zeta_1^{*n_0} = n_{n_0}^*$ and $\alpha_1^{n_0} = \tau_{n_0}$, the propositions (5.3.13b) and (5.3.13c) are true for $n = n_0$ and $\ell = 1$.

2.2. Suppose (5.3.13b) and (5.3.13c) are true for $n = n_0$ and all $\ell \leq \ell_0 - 1$ where $2 \leq \ell_0 \leq m+1$. Set $t_{n_0}, \ell_0 - 1 = t_{n_0} + \nu \ell_0 - 1h_{n_0}$. Then positive constants $\overline{\sigma}_1, \ \delta_{n_0}, \ \delta_{1,n_0}, \ell_{-1}$ ($\ell = 2, 3, \dots, \ell_0 - 1$) and $\delta_{2,n_0}, \ell_0 - 1$ exist such that for $n = n_0$ we have (5.3.13a) is true, (5.3.13b) is true (for $\ell = 2, 3, \dots, \ell_0 - 1$) and (5.3.13c) is true (for $\ell = \ell_0 - 1$), if $\sigma_{n_0} = \overline{\sigma}_1$ and $\sigma_{n_0}, \ell = \overline{\sigma}_1$ ($\ell = 1, 2, \dots, \ell_0 - 1$).

 $\begin{array}{l} (\ell = 1, 2, \ldots, \ell_0 - 1) \, . \\ & \text{Since } [\zeta_{\ell_0 - 1}^{\star n_0}]^{*}(\mathbf{x}^{\star}) \text{ exists and } \zeta_{\ell_0 - 1}^{\star n_0}(\mathbf{x}^{\star}) = \mathbf{x}^{\star}, \text{ a constant } \theta > 0 \text{ and a } \\ & \text{constant } \overline{\sigma}_2 \in (0, \overline{\sigma}_1] \text{ exist such that } B(\mathbf{x}^{\star}, \overline{\sigma}_2) \subset D(\zeta_{\ell_0 - 1}^{\star n_0}) \text{ and} \end{array}$

$$\|\zeta_{\ell_0^{-1}}^{*^n_0}(\mathbf{x}) - \mathbf{x}^*\| \le \theta \|\mathbf{x} - \mathbf{x}^*\| \quad (\text{for all } \mathbf{x} \in B(\mathbf{x}^*, \overline{\sigma}_2)).$$

From Lemma 5.3.6 and Lemma 5.3.7 it follows that positive constants $c_{1,i}$, $c_{2,i}$ (i = 2,3) and a constant $\overline{\sigma}_3 > 0$ exist such that (5.3.9) and (5.3.10) hold for t = t with $\sigma = \overline{\sigma}_3$. Set $\overline{\sigma}_4 = \min\{\overline{\sigma}_2, \overline{\sigma}_3/(1+\theta)\}$. Then $B(x^*, \overline{\sigma}_4) \subset D(\kappa_{\ell_0}^{*n_0})$. Let $x \in B(x^*, \overline{\sigma}_4)$. As a consequence of the chainrule

and (5.3.10) it follows that $[\kappa_{\mathcal{L}_0-1}^{\star^n_0}]'(x)$ exists and

$$\begin{bmatrix} \kappa_{\ell_0-1}^{*n_0} \end{bmatrix}^{\prime} (x) = \partial_2 \Phi(t_{n_0}, \ell_0-1, \zeta_{\ell_0-1}^{*n_0} (x), x) \begin{bmatrix} \zeta_{\ell_0-1}^{*n_0} \end{bmatrix}^{\prime} (x)$$

+ $\partial_3 \Phi(t_{n_0}, \ell_0-1, \zeta_{\ell_0-1}^{*n_0} (x), x) .$

Hence, (cf. Lemma 5.3.3),

$$\begin{bmatrix} \kappa_{\ell_0-1}^{*^{n_0}} \end{bmatrix}^{*} (x^{*}) = u_1 (t_{n_0}, \ell_0^{-1}, C) \alpha_{\ell_0-1}^{*^{n_0}} (C) + u_2 (t_{n_0}, \ell_0^{-1}, C)$$
$$= \pi_{\ell_0-1}^{n_0} (C) .$$

Further

$$\begin{split} &\|[\kappa_{\ell_{0}-1}^{*^{n_{0}}}]^{*}(\mathbf{x}) - [\kappa_{\ell_{0}-1}^{*^{n_{0}}}]^{*}(\mathbf{x}^{*})\| \\ &\leq \|\vartheta_{2}^{\Phi}(\mathsf{t}_{\mathsf{n}_{0}},\ell_{0}^{-1},\varsigma_{\ell_{0}-1}^{*^{n_{0}}}(\mathbf{x}),\mathbf{x}) - \vartheta_{2}^{\Phi}(\mathsf{t}_{\mathsf{n}_{0}},\ell_{0}^{-1},\mathbf{x}^{*},\mathbf{x}^{*})\|\|[\varsigma_{\ell_{0}-1}^{*^{n_{0}}}]^{*}(\mathbf{x})\| \\ &+ \|\vartheta_{2}^{\Phi}(\mathsf{t}_{\mathsf{n}_{0}},\ell_{0}^{-1},\mathbf{x}^{*},\mathbf{x}^{*})\|\|[\varsigma_{\ell_{0}-1}^{*^{n_{0}}}]^{*}(\mathbf{x}) - [\varsigma_{\ell_{0}-1}^{*^{n_{0}}}]^{*}(\mathbf{x}^{*})\| \\ &+ \|\vartheta_{3}^{\Phi}(\mathsf{t}_{\mathsf{n}_{0}},\ell_{0}^{-1},\varsigma_{\ell_{0}-1}^{*^{n_{0}}}(\mathbf{x}),\mathbf{x}) - \vartheta_{3}^{\Phi}(\mathsf{t}_{\mathsf{n}_{0}},\ell_{0}^{-1},\mathbf{x}^{*},\mathbf{x}^{*})\| \\ &\leq (c_{1,2}^{*} + c_{2,2}^{\theta})\|\mathbf{x}-\mathbf{x}^{*}\|\{\|\alpha_{\ell_{0}-1}^{n_{0}}(\mathbf{c})\| + \delta_{2,\mathbf{n}_{0}},\ell_{0}^{-1}\overline{\alpha}_{4}^{3}\} \\ &+ \|\vartheta_{2}^{\Phi}(\mathsf{t}_{\mathsf{n}_{0}},\ell_{0}^{-1},\mathbf{x}^{*},\mathbf{x}^{*})\|\delta_{2,\mathbf{n}_{0}},\ell_{0}^{-1}\|\mathbf{x}-\mathbf{x}^{*}\| + (c_{1,3}^{*}+c_{2,3}^{\theta})\|\mathbf{x}-\mathbf{x}^{*}\|. \end{split}$$

Set

١

$$\delta_{1,n_{0},\ell_{0}-1} = (c_{1,2} + c_{2,2}^{\theta}) \{ \| \alpha_{\ell_{0}-1}^{n_{0}}(C) \| + \delta_{2,n_{0},\ell_{0}-1}^{\overline{\sigma}} \}^{+} \\ + \| \partial_{2} \Phi(t_{n_{0},\ell_{0}-1}, x^{*}, x^{*}) \| \delta_{2,n_{0},\ell_{0}-1}^{-1} + c_{1,3}^{+} + c_{2,3}^{\theta} .$$

Then (5.3.13b) holds for $n = n_0$ and $\ell = \ell_0$ when $\sigma_{n_0,\ell_0} = \overline{\sigma}_4$. Consequently $B(\mathbf{x}^*, \overline{\sigma}_4) \subset D(\zeta_{\ell_0}^{*n_0})$ and for any $\mathbf{x} \in B(\mathbf{x}^*, \overline{\sigma}_4)$ the derivative $[\zeta_{\ell_0}^{*n_0}]^*(\mathbf{x})$ exists and

$$[\zeta_{\ell_0}^{*^{n_0}}]'(\mathbf{x}) = \eta_{n_0}^{*'}(\mathbf{x}) + h_n \sum_{j=1}^{\ell_0^{-1}} \Lambda_{\ell_0,j}^{*^{(n_0)}} [\kappa_j^{*^{n_0}}]'(\mathbf{x}).$$

Hence,

$$\begin{bmatrix} \zeta_{\ell_0}^{*^{n_0}} \end{bmatrix}^{\prime} (\mathbf{x}^{*}) = \tau_{n_0}^{(C)} + h_{n_0}^{\sum_{j=1}^{n_0} \rho_{\ell_0,j}(h_{n_0}^{u_1}(t_{n_0,c}^{(C)})) \pi_j^{n_0}(C) \\ = \alpha_{\ell_0}^{n_0}^{(C)} .$$

Further

Hence (5.3.13c) is true for $n = n_0$ and $\ell = \ell_0$. Therefore (5.3.13b), (5.3.13c) are true for $n = n_0$ and all $\ell \le m+1$ 2.3. Suppose $n_0 < N$. Since $\eta_{n_0+1}^{\star} = \zeta_{m+1}^{\star n_0}$ and $\tau_{n_0+1} = \alpha_{m+1}^{n_0}$, the proposition (5.3.13a) is true for $n = n_0+1$. This proves the lemma.

Lemma 5.3.10 describes the similarity between G and G^* near x^* . For the proof of Lemma 5.3.10 we need the following lemma.

LEMMA 5.3.9. Let ρ be a rational function with real coefficients. If $C_0 \in D_E(\rho)$, then positive numbers ε and δ exist such that all $C \in L(E)$ with $\|C - C_0\| < \varepsilon$ belong to $D_E(\rho)$ and satisfy the inequality

$$\|\rho(C) - \rho(C_0)\| \le \delta \|C - C_0\|.$$

<u>PROOF</u>. Let $p(z) \equiv [q(z)]^{-1}p(z)$ whereby p and q are polynomials with real coefficients. Then $\overline{\epsilon}, \delta_1, \delta_2 > 0$ exist such that all $C \in L(E)$ with $\|C - C_0\| < \overline{\epsilon}$ satisfy

$$\|q(C) - q(C_0)\| \le \delta_1 \|C - C_0\|$$

and

$$\| \mathbf{p}(\mathbf{C}) - \mathbf{p}(\mathbf{C}_{0}) \| \leq \delta_{2} \| \mathbf{C} - \mathbf{C}_{0} \|.$$

Further $[c_1]^{-1} - [c_2]^{-1} = [c_2]^{-1}[c_2 - c_1][c_1]^{-1}$ (for all $c_1, c_2 \in L(E)$ which are invertible). Together with Lemma 4.2.1 this yields the result.

LEMMA 5.3.10. Let Condition 0 be satisfied. Then (5.3.11) holds. Let G^* be defined in (5.3.12). Then for $n = 0, 1, \ldots, N$ the propositions (5.3.14a), (5.3.14b) and (5.3.14c) are true.

Positive constants σ_n and δ_n exist such that

$$B(\mathbf{x}^{\star},\sigma_{n}) \subset D(\eta_{n}) \cap D(\eta_{n}^{\star}),$$

(5.3.14a)

(5.3

$$\|\eta_{n}(\mathbf{x}) - \eta_{n}^{*}(\mathbf{x})\| \leq \delta_{n} \|\mathbf{x} - \mathbf{x}^{*}\|^{2} \text{ whenever } \mathbf{x} \in B(\mathbf{x}^{*}, \sigma_{n}).$$

For l = 1, 2, ..., m+1 positive constants $\sigma_{n,l}$, $\delta_{2,n,l}$ and, if l > 1, $\delta_{1,n,l-1}$ exist such that

.14b)
$$B(\mathbf{x}^{\star}, \sigma_{n, \ell}) \subset D(\kappa_{\ell-1}^{n}) \cap D(\kappa_{\ell-1}^{\star^{n}}),$$
$$\|\kappa_{\ell-1}^{n}(\mathbf{x}) - \kappa_{\ell-1}^{\star^{n}}(\mathbf{x})\| \leq \delta_{1, n, \ell-1} \|\mathbf{x} - \mathbf{x}^{\star}\|^{2}$$

(if $\ell > 1$) and

$$(5.3.14c) \qquad B(\mathbf{x}^{\star}, \sigma_{n,\ell}) \subset D(\boldsymbol{\zeta}_{\ell}^{n}) \cap D(\boldsymbol{\zeta}_{\ell}^{\star^{n}}), \\ \|\boldsymbol{\zeta}_{\ell}^{n}(\mathbf{x}) - \boldsymbol{\zeta}_{\ell}^{\star^{n}}(\mathbf{x})\| \leq \delta_{2,n,\ell} \|\mathbf{x} - \mathbf{x}^{\star}\|^{2}$$

whenever $x \in B(x^*, \sigma_{n, \ell})$. In particular, since $G = \zeta_{m+1}^N$ and $G^* = \zeta_{m+1}^*$,

$$B(x^*, \sigma_{N,m+1}) \subset D(G) \cap D(G^*)$$
 and

(5.3.14d)

$$\|G(\mathbf{x}) - G^{*}(\mathbf{x})\| \leq \delta_{2,N,m+1} \|\mathbf{x} - \mathbf{x}^{*}\|^{2}$$

whenever $x \in B(x^*, \sigma_{N, m+1})$.

PROOF. Since Condition 0 is satisfied, (5.3.11) holds. 1. Let n = 0. Since $D(F) = D(n_0) = D(n_0^*)$ and $n_0 = n_0^*$, the proposition (5.3.14a) is true for n = 0. 2. Suppose (5.3.14a) is true for some $n = n_0$ with $0 \le n_0 \le N$. 2.1. Since $\zeta_1^{n_0} = n_{n_0}$ and $\zeta_1^{*n_0} = n_{n_0}^*$, the propositions (5.3.14b), (5.3.14c) are true for $n = n_0$ and $\ell = 1$. 2.2. Suppose (5.3.14b), (5.3.14c) are true for $n = n_0$ and all $\ell \le \ell_0 - 1$ where $2 \leq \ell_0 \leq m+1$. Set $t_{n_0,\ell_0-1} = t_{n_0} + \nu_{\ell_0-1}h_{n_0}$. Then positive constants $\overline{\sigma}_1$, δ_{n_0} , δ_{2,n_0,ℓ_0-1} and $\delta_{1,n_0,\ell-1}$ ($\ell = 2,3,\ldots,\ell_0-1$) exist such that for $n = n_0$ we have (5.3.14a) is true, (5.3.14b) is true (for $\ell = 2,3,\ldots,\ell_0-1$) and (5.3.14c) is true (for $\ell = \ell_0-1$), if $\sigma_{n_0} = \overline{\sigma}_1$ and $\sigma_{n_0,\ell} = \overline{\sigma}_1$ ($\ell = 1,2,\ldots,\ell_0-1$).

From Lemma 5.3.8 it follows that constants $\overline{\sigma}_2 \in (0, \overline{\sigma}_1]$ and $\overline{\delta}_1 > 0$ exist such that $B(\mathbf{x}^*, \overline{\sigma}_2) \subset D(\eta_{n_0}^*) \cap D(\zeta_{\ell_0-1}^{*^{n_0}})$,

(5.3.15a)
$$\|\eta_{n_0}^*(\mathbf{x}) - \mathbf{x}^*\| \le \overline{\delta}_1 \|\mathbf{x} - \mathbf{x}^*\|$$

and

(5.3.15b)
$$\|\zeta_{\ell_0^{-1}}^{\star^{10}}(\mathbf{x}) - \mathbf{x}^{\star}\| \leq \overline{\delta}_1 \|\mathbf{x} - \mathbf{x}^{\star}\|$$
 (for all $\mathbf{x} \in B(\mathbf{x}^{\star}, \overline{\sigma}_2)$).

Moreover, for $\ell = 2, 3, \dots, \ell_0$, $B(x^*, \overline{\sigma}_2) \subset D(\kappa_{\ell-1}^{*^0})$ and

(5.3.15c)
$$\|\kappa_{\ell-1}^{\star^{*0}}(\mathbf{x})\| \leq \overline{\delta}_1 \|\mathbf{x} - \mathbf{x}^{\star}\|$$
 (for all $\mathbf{x} \in B(\mathbf{x}^{\star}, \overline{\sigma}_2)$).

From (5.3.14c) (n = n_0 , $\ell = \ell_0 - 1$) and (5.3.15b) it follows that

(5.3.15d)
$$B(\mathbf{x}^*, \overline{\sigma}_2) \subset D(\zeta_{\ell_0-1}^n)$$
 and $\|\zeta_{\ell_0-1}^n(\mathbf{x}) - \mathbf{x}^*\| \leq \overline{\delta}_2 \|\mathbf{x} - \mathbf{x}^*\|$
(for all $\mathbf{x} \in B(\mathbf{x}^*, \overline{\sigma}_2)$)

where $\overline{\delta}_2 = \overline{\delta}_1 + \delta_{2,n_0,\ell_0-1}\overline{\sigma}_2$. Since $(t,x^*,x^*) \in D(\Phi)$ (for all $t \in \{t_{n_0}, t_{n_0,\ell_0-1}\}$), because of Lemma 5.3.6 and the relations (5.3.15d), (5.3.15c) $(\ell = \ell_0)$, there exist positive constants c_1, c_2 and a constant $\overline{\sigma}_3 \in (0,\overline{\sigma}_2]$ such that $B(x^*,\overline{\sigma}_3) \subset D(\kappa_{\ell_0-1}^{n_0}) \cap D(\kappa_{\ell_0-1}^{*n_0})$ and

$$(5.3.16) \qquad \|\partial_2 \Phi(t,y,x) - \partial_2 \Phi(t,x^*,x^*)\| \le c_1 \|x-x^*\| + c_2 \|y-x^*\|$$

(for all
$$t \in \{t_n, t_n, \ell_0, \ell_0^{-1}\}$$
 and all $x, y \in B(x^*, \overline{\sigma}_3)$).

Set $\overline{\sigma}_4 = [1 + \overline{\delta}_2]^{-1} \overline{\sigma}_3$. Then, from Lemma 2.6.1 it follows that for all $x \in B(x^*, \overline{\sigma}_4)$
$$\begin{aligned} &\|\kappa_{\ell_{0}-1}^{n_{0}}(\mathbf{x}) - \kappa_{\ell_{0}-1}^{*^{n_{0}}}(\mathbf{x})\| \\ &= \|\Phi(t_{n_{0}}, \ell_{0}-1, \zeta_{\ell_{0}-1}^{n_{0}}(\mathbf{x}), \mathbf{x}) - \Phi(t_{n_{0}}, \ell_{0}-1, \zeta_{\ell_{0}-1}^{*^{n_{0}}}(\mathbf{x}), \mathbf{x})\| \\ &\leq \{\|\vartheta_{2}\Phi(t_{n_{0}}, \ell_{0}-1, \mathbf{x}^{*}, \mathbf{x}^{*})\| + c_{1} + c_{2}\overline{\delta}_{2}\overline{\sigma}_{4}\}\delta_{2, n_{0}}, \ell_{0}-1^{\|\mathbf{x}-\mathbf{x}^{*}\|^{2}} \end{aligned}$$

Therefore, a constant δ_{1,n_0,ℓ_0-1} exists such that (5.3.14b) is true for $n = n_0$ and $\ell = \ell_0$ with $\sigma_{n_0,\ell_0} \in (0, \overline{\sigma}_4]$.

Observe that

$$h_{n_0} \partial_2 \Phi(t_{n_0}, x^*, x^*) \in D_E(\rho_{\ell_0}, j)$$
 (j = 1, 2, ..., ℓ_0^{-1}).

Hence, from Lemma 5.3.9 and relation (5.3.16) it follows that constants $\bar{\sigma}_5 \in (0, \bar{\sigma}_4]$ and $\bar{\delta}_3 > 0$ exist such that for all x,y $\in B(x^*, \bar{\sigma}_5)$

$$h_{n_0} \partial_2 \Phi(t_{n_0}, y, x) \in D_E(\rho_{\ell_0}, j)$$

and

$$\| \rho_{\ell_{0}, j}(h_{n_{0}}^{2} 2^{\Phi(t_{n_{0}}, y, x)}) - \rho_{\ell_{0}, j}(h_{n_{0}}^{2} 2^{\Phi(t_{n_{0}}, x^{*}, x^{*})}) \|$$

$$\leq \overline{\delta}_{3}\{c_{1}\|x - x^{*}\| + c_{2}\|y - x^{*}\|\} \qquad (j = 1, 2, \dots, \ell_{0}^{-1}).$$

Let $\overline{\delta}_4 = \overline{\delta}_1 + \delta_{n_0} \overline{\sigma}_5$. Then $B(\mathbf{x}^*, \overline{\sigma}_5) \subset D(\overline{n}_{n_0})$ and $\|\mathbf{n}_{n_0}(\mathbf{x}) - \mathbf{x}^*\| \le \overline{\delta}_4 \|\mathbf{x} - \mathbf{x}^*\|$ (for all $\mathbf{x} \in B(\mathbf{x}^*, \overline{\sigma}_5)$). Set $\overline{\sigma}_6 = [1 + \overline{\delta}_4]^{-1} \overline{\sigma}_5$. Then $B(\mathbf{x}^*, \overline{\sigma}_6) \subset D(\zeta_{\ell_0}^{n_0}) \cap D(\zeta_{\ell_0}^{*n_0})$ and for all $\mathbf{x} \in B(\mathbf{x}^*, \overline{\sigma}_6)$

$$\begin{split} \| \zeta_{\ell_{0}}^{n_{0}}(\mathbf{x}) - \zeta_{\ell_{0}}^{\star^{n_{0}}}(\mathbf{x}) \| \\ &= \| \eta_{n_{0}}(\mathbf{x}) - \eta_{n_{0}}^{\star}(\mathbf{x}) + h_{n_{0}} \sum_{j=1}^{n} \{ \Lambda_{\ell_{0}, j}^{(n_{0})} \kappa_{j}^{n_{0}}(\mathbf{x}) - \Lambda_{\ell_{0}, j}^{\star^{(n_{0})}} \kappa_{j}^{\star^{n_{0}}}(\mathbf{x}) \} \| \\ &\leq \delta_{n_{0}} \| \mathbf{x} - \mathbf{x}^{\star} \|^{2} + h_{n_{0}} \sum_{j=1}^{n} \{ \| \Lambda_{\ell_{0}, j}^{(n_{0})} \| \| \kappa_{j}^{n_{0}}(\mathbf{x}) - \kappa_{j}^{\star^{n_{0}}}(\mathbf{x}) \| + \\ &+ \| \Lambda_{\ell_{0}, j}^{(n_{0})} - \Lambda_{\ell_{0}, j}^{\star^{(n_{0})}} \| \| \kappa_{j}^{\star^{n_{0}}}(\mathbf{x}) \| \} \leq \end{split}$$

$$\leq \{\delta_{n_{0}} + h_{n_{0}} \sum_{j=1}^{\lambda_{0}-1} \{[\|\Lambda_{\ell_{0},j}^{*}\| + \overline{\delta}_{3}(c_{1} + c_{2}\overline{\delta}_{4})\overline{\sigma}_{6}]\delta_{1,n_{0},j} \\ + \overline{\delta}_{3}(c_{1} + c_{2}\overline{\delta}_{4})\overline{\delta}_{1}\} \|\mathbf{x}-\mathbf{x}^{*}\|^{2}.$$

Hence (5.3.14b), (5.3.14c) hold for $n = n_0$ and all $\ell \le m+1$. 2.3. If $n_0 \le N$ then, since $\eta_{n_0+1} = \zeta_{m+1}^{n_0}$ and $\eta_{n_0+1}^* = \zeta_{m+1}^{*n_0}$, (5.3.14a) holds for $n = n_0 + 1$ as well. This proves the lemma.

From Lemma 5.3.10 it follows that $x^* \in interior(D(G))$ whenever Condition 0 is satisfied. The next lemma shows that the reverse is also true.

LEMMA 5.3.11. Condition 0 is satisfied if and only if $x^* \in interior(D(G))$.

<u>PROOF</u>. 1. Suppose $\mathbf{x}^* \in \text{interior}(D(G))$. Obviously, $(t, \mathbf{x}^*, \mathbf{x}^*) \in D(\Phi)$ for all $t = t_n + v_{\ell-1}h_n$ (n = 0,1,...,N; $\ell = 2,3,\ldots,m+1$). In particular $\partial_1 K(\mathbf{x}^*, \mathbf{x}^*)$ is invertible. Using Lemma 5.3.3, it is easily verified that Condition 0 is satisfied.

2. If Condition 0 is satisfied, from Lemma 5.3.10 it follows that $x^* \in interior(D(G))$.

The following important lemma is a consequence of Lemma 5.3.8 and Lemma 5.3.10.

LEMMA 5.3.12. If Condition 0 is satisfied, then $x^* \in interior(D(G))$, the derivative $G'(x^*)$ exists and $G'(x^*) = \gamma(C)$.

<u>PROOF</u>. From Lemma 5.3.8 it follows that $x^* \in interior(D(G^*))$, the derivative $[G^*]'(x^*)$ exists and $[G^*]'(x^*) = \gamma(C)$. From Lemma 5.3.10 it follows that $x^* \in interior(D(G))$. From (5.3.14d) it follows that $G'(x^*)$ exists and $G'(x^*) = [G^*]'(x^*)$.

The next three lemmata will be useful in the proofs of the Theorems 5.2.1 and 5.2.2.

 $(5.3.17a) \quad 1 \in D_{\mathbb{C}}(\pi_{\ell-1}^{n}), \quad \pi_{\ell-1}^{n}(1) = -1$

(if $\ell > 1$) and

(5.3.17b)
$$1 \in D_{\mathcal{C}}(\alpha_{\ell}^{n}), \quad \alpha_{\ell}^{n}(1) = 1 - t_{n} - \nu_{\ell}h_{n}$$

(n = 0,1,...,N; ℓ = 1,2,...,m+1). In particular we have

(5.3.17c)
$$1 \in D_{\sigma}(\gamma)$$
 and $\gamma(1) = 0$.

 $\begin{array}{l} \underline{PROOF}. \ \text{Notice that } u_1(t,1) \equiv -g(t) \ \text{and } u_2(t,1) \equiv -(1-g(t)(1-t)). \\ 1. \ \alpha_1^0 = \tau_0 \ \text{and } \tau_0(z) \equiv 1 \ \text{hold. Hence } (5.3.17a), \ (5.3.17b) \ \text{are true for} \\ n = 0 \ \text{and } \ell = 1. \\ 2. \ \text{Suppose } (5.3.17a), \ (5.3.17b) \ \text{hold for } n = n_0 - 1 \ \text{with } 0 < n_0 \leq N \ \text{and} \\ \ell = 1, 2, \ldots, m+1. \\ 2.1. \ \text{Since } \alpha_1^{n_0} = \tau_{n_0} = \alpha_{m+1}^{n_0-1}, \ \text{we have } 1 \in D_{\mathbb{C}}(\alpha_1^0) \ \text{and } \alpha_1^0(1) = 1 - t_{n_0-1} - v_{m+1}h_{n_0-1} = 1 - t_{n_0}. \ \text{Thus } (5.3.17a), \ (5.3.17b) \ \text{hold for } n = n_0 \ \text{and } \ell = 1. \\ 2.2. \ \text{Suppose } (5.3.17a), \ (5.3.17b) \ \text{are true for } n = n_0 \ \text{and } all \ \ell \leq \ell_0 - 1 \\ \text{where } 2 \leq \ell_0 \leq m+1. \ \text{Set } t_{n_0}, \ell_{0-1} = t_{n_0} + v_{\ell_0-1}h_{n_0}. \ \text{Then } 1 \in D_{\mathbb{C}}(\pi_{\ell_0-1}) \ \text{and} \ \mu_0 + 1 \\ \end{array}$

$$\pi_{\ell_0-1}^{n_0}(1) = -g(t_{n_0,\ell_0-1})(1 - t_{n_0,\ell_0-1}) - \{1 - g(t_{n_0,\ell_0-1})(1 - t_{n_0,\ell_0-1})\} = -1.$$

Further

$$1 \in D_{\mathbb{C}}(\alpha \ell_0)$$

and

This proves the lemma.

The next two lemmata are direct consequences of Lemma 5.3.13.

LEMMA 5.3.14. Let the assumptions of Lemma 5.3.13 hold. Suppose $sp(\tilde{L}F'(x^*)]^{-1} \partial_1 K(x^*,x^*)) = \{1\}$. Then Condition 0 is satisfied and $sp(\gamma(C)) = \{0\}$.

<u>PROOF</u>. Set $T = [F'(x^*)]^{-1} \partial_1 K(x^*, x^*)$. According to Theorem 2.2.2 the operator T is invertible. It is easily verified that $\partial_1 K(x^*, x^*)$ is invertible and $[\partial_1 K(x^*, x^*)]^{-1} = [T]^{-1} [F'(x^*)]^{-1}$. Set $C = [\partial_1 K(x^*, x^*)]^{-1} F'(x^*)$. Since $C = [T]^{-1}$, from Theorem 2.2.2 it follows that $sp(C) = \{1\}$. Consequently, from Lemma 5.3.13 and Theorem 2.2.2 it follows that the statement of the lemma is true.

LEMMA 5.3.15. Let the assumptions of Lemma 5.3.13 hold. Suppose $\partial_1 K(\mathbf{x}^*, \mathbf{x}^*) = \mathbf{F}'(\mathbf{x}^*)$. Then Condition 0 is satisfied and $\gamma(C) = 0$.

<u>**PROOF.</u>** According to Lemma 5.3.14 Condition 0 is satisfied. In this case C = I. It is easily verified that $\gamma(I) = 0$.</u>

5.3.2 Proof of Theorem 5.1.1

a. The propositions (i), (ii), (iii) and (iv) are immediate consequences of Lemma 5.3.11, Lemma 5.3.12 and Theorem 2.5.1.

b. Suppose that Condition 5 is satisfied. Due to Lemma 5.3.8 and Lemma 5.3.10 positive constants σ , δ_1 and δ_2 exist such that $B(x^*, \sigma) \subset D(G) \cap D(G^*)$ and

 $\|[G^{*}]'(x)\| \leq \delta_{1} \|x - x^{*}\|,$ $\|G(x) - G^{*}(x)\| \leq \delta_{2} \|x - x^{*}\|^{2}$

whenever $x \in B(x^*, \sigma)$. Since $G^*(x^*) = x^*$, from Lemma 5.3.1 it follows that

$$\|\mathbf{G}^{\star}(\mathbf{x}) - \mathbf{x}^{\star}\| \leq \frac{\delta_{1}}{2} \|\mathbf{x} - \mathbf{x}^{\star}\|^{2} \qquad (\text{for all } \mathbf{x} \in \mathbf{B}(\mathbf{x}^{\star}, \sigma)).$$

Hence

$$\|\mathbf{G}(\mathbf{x}) - \mathbf{x}^{\star}\| \leq (\delta_2 + \frac{\delta_1}{2}) \|\mathbf{x} - \mathbf{x}^{\star}\|^2 \qquad (\text{for all } \mathbf{x} \in \mathbf{B}(\mathbf{x}^{\star}, \sigma)).$$

Thus [M,F] is quadratically convergent.

c. Suppose [M,F] is quadratically convergent. From Lemma 5.3.11 it follows that Condition 0 is satisfied. From Lemma 5.3.12 and Theorem 2.5.4 it follows that Condition 5 is satisfied. This completes the proof.

5.3.3. Proof of the Theorems 5.2.1 and 5.2.2

Suppose that (5.2.1) holds.

1. <u>PROOF OF THEOREM 5.2.1</u>. Suppose $sp([F'(x^*)]^{-1}\partial_1 K(x^*,x^*)) = \{1\}$. Due to Lemma 5.3.14 Condition 1 holds. Theorem 5.1.1(i) yields the result.

2. <u>PROOF OF THEOREM 5.2.2</u>. Suppose $F'(x^*) = \partial_1 K(x^*, x^*)$. Due to Lemma 5.3.15 Condition 5 holds. Theorem 5.1.1(v) yields the result.

5.3.4. Proof of the Theorems 5.2.3-5

1. <u>PROOF OF THEOREM 5.2.3</u>. a. We show that (i) implies (ii). To that end we need the following two lemmata. For $\theta \in (0,1]$ and $\delta \in \mathbb{R}$, define

(5.3.18)
$$\hat{\mathbf{L}}(\theta, \delta) = \begin{pmatrix} 0 & 0 & 0 \\ \theta & 0 & 0 \\ \delta & 1-\delta & 0 \end{pmatrix}$$

LEMMA 5.3.16. Let $\theta \in (0,1]$ and $\delta \in \mathbb{R}$. Let $L = \hat{L}(\theta,\delta)$ and $H = \{1\}$. Set G = M(F) where $M = \mathbb{IM}(A,g,L,H)$. Let the rational function γ be defined in (5.1.3) where R = L. Suppose $x^* \in interior(D(G))$. Then Condition 0 holds, $\partial_1 K(x^*,x^*)$ is invertible and $[(1-\theta)I + \theta C]$ is invertible where

(5.3.19)
$$C = [\partial_1 K(x^*, x^*)]^{-1} F'(x^*).$$

Further C \in D_E(\gamma), the derivative G'(x*) exists, G'(x*) = γ (C) and γ (C) = C1 + δ C2 where

(5.3.20)
$$C_1 = I + \{[(1-\theta)I + \theta C]^{-1}[I-C] - g(\theta)I\}[I-\theta C] - \{1 - g(\theta)(1-\theta)\}[(1-\theta)I + \theta C]^{-1}$$

and

(5.3.21) $C_2 = \theta \{2 + \theta g(\theta)\} [(1-\theta)I + \theta C]^{-1} C[I-C].$

<u>PROOF</u>. According to Lemma 5.3.11 Condition 0 is satisfied (where R = L). Hence $\partial_1 K(x^*, x^*)$ is invertible, $[(1-\theta)I + \theta C]$ is invertible and $C \in D_E(\gamma)$. From Lemma 5.3.12 it follows that G'(x^*) exists and G'(x^*) = $\gamma(C)$. A little calculation shows that

(5.3.22)
$$\gamma(C) = I + \delta\{-C\} + (1-\delta)\{\{[(1-\theta)I + \theta C]^{-1}[I-C] - g(\theta)I\}[I-\theta C] - [1-g(\theta)(1-\theta)][(1-\theta)I + \theta C]^{-1}\}.$$

Resolving relation (5.3.22) one obtains

$$\gamma(C) = C_1 + \delta C_2.$$

This proves the lemma. \Box

LEMMA 5.3.17. Let $\theta \in (0,1]$ and let C_1 and C_2 be defined in (5.3.20) and (5.3.21) respectively. Then for any $\delta \in \mathbb{R}$

$$\operatorname{sr}(C_1 + \delta C_2) \ge |\delta| \operatorname{sr}(C_2) - \operatorname{sr}(C_1).$$

<u>PROOF</u>. Let $\delta \in \mathbb{R}$. It is easily verified that C_1 and C_2 commute. Hence $C_1 + \delta C_2$ and $-C_1$ commute. Therefore (cf. [RUDIN, 1973; Theorem 11.23])

$$sr(C_1 + \delta C_2 - C_1) \leq sr(C_1 + \delta C_2) + sr(-C_1).$$

Consequently,

$$\operatorname{sr}(\operatorname{C}_{1} + \delta\operatorname{C}_{2}) \geq \operatorname{sr}(\delta\operatorname{C}_{2}) - \operatorname{sr}(-\operatorname{C}_{1}) = |\delta|\operatorname{sr}(\operatorname{C}_{2}) - \operatorname{sr}(\operatorname{C}_{1}). \qquad \Box$$

Suppose (i) holds and $sp([F'(x^*)]^{-1}\partial_1 K(x^*,x^*)) \neq \{1\}$. There follows that $(0,x^*,x^*) \in D(\Phi)$ so that $\partial_1 K(x^*,x^*)$ is invertible. Choose $\theta \in (0,1]$ such that $2 + \theta g(\theta) \neq 0$. (This is possible since $g \in S$, cf. (2.6.9).) Let $\delta \in \mathbb{R}$. Since (i) holds, Lemma 5.3.16 applies. Let C_1 and C_2 be defined in (5.3.20) and (5.3.21), respectively. From Theorem 2.2.2 it follows that $sp(C_2) \neq \{0\}$. Choose $\delta_1 \in \mathbb{R}$ such that $|\delta_1|sr(C_2) - sr(C_1) > 1$. Let $L = \widehat{L}(\theta, \delta_1)$ and $H = \{1\}$. Let the rational function γ be defined in (5.1.3) where R = L. From Lemma 5.3.16 and Lemma 5.3.17 it follows that

$$\operatorname{sr}(\gamma(C)) \geq |\delta_1| \operatorname{sr}(C_2) - \operatorname{sr}(C_1) > 1.$$

Using Theorem 5.1.1(ii) this yields a contradiction. b. From Theorem 5.2.1 it follows that (ii) implies (i). \Box

2. <u>PROOF OF THE THEOREMS 5.2.4, 5.2.5</u>. We shall show that (vi) implies (v), and that (iii) implies (iv). This is of course sufficient to prove both Theorem 5.2.4 and Theorem 5.2.5.

a. From Theorem 5.2.2 it follows that (vi) implies (v).

b. We show that (iii) implies (iv). Assume (iii) holds and suppose $\partial_1 K(\mathbf{x}^*, \mathbf{x}^*) \neq F'(\mathbf{x}^*)$. Choose $\theta \in (0,1]$ such that $2 + \theta g(\theta) \neq 0$. Let $\delta \in \mathbb{R}$. Lemma 5.3.16 applies. Let C, C_1 and C_2 be defined in (5.3.19), (5.3.20) and (5.3.21), respectively. Since $I - C \neq 0$ and C is invertible, we have $\|C_2\| \neq 0$. Choose $\delta_1 \in \mathbb{R}$ such that $\|\delta_1\| \|C_2 - C_1\| > 1$. Let $L = \hat{L}(\theta, \delta_1)$ (cf. (5.3.18)) and $H = \{1\}$. Let the rational function γ be defined in (5.1.3) where R = L. From Lemma 5.3.16 it follows that $C \in D_E(\gamma)$ and

$$\|\gamma(\mathbf{C})\| = \|\delta_{1}\mathbf{C}_{2} + \mathbf{C}_{1}\| \ge \|\delta_{1}\|\|\mathbf{C}_{2}\| - \|\mathbf{C}_{1}\| > 1.$$

Using Theorem 5.1.1(iv) this yields a contradiction. \Box

5.3.5. Proof of the Theorems 5.2.6-8

1. <u>PROOF OF THEOREM 5.2.6</u>. a. We show that (i) *implies* (ii). We shall use the following lemma.

<u>LEMMA 5.3.18</u>. Suppose $sp([F'(x^*)]^{-1}\partial_1 K(x^*,x^*)) = \{1\}$ and $g(\theta) \neq 0$ for some $\theta \in [0,1)$. Then a Runge-Kutta method with operator coefficients R and a sequence of stepsizes H exist such that $x^* \notin D(G)$ where G = M(F) and $M = \mathbb{M}(A,g,R,H)$.

<u>PROOF</u>. Since $sp([F'(x^*)]^{-1}\partial_1 K(x^*,x^*)) = \{1\}$, the operator $\partial_1 K(x^*,x^*)$ is invertible. Set $C = [\partial_1 K(x^*,x^*)]^{-1}F'(x^*)$. Since $sp(C) = \{1\}$ it follows that the operator $[(1-\theta)I + \theta C]$ is invertible. From Lemma 5.3.3 it follows that $\partial_2 \Phi(\theta, x^*, x^*)$ exists and

$$\partial_2 \Phi(\theta, \mathbf{x}^*, \mathbf{x}^*) = -g(\theta) + [(1-\theta)\mathbf{I} + \theta \mathbf{C}]^{-1}[\mathbf{I} - \mathbf{C}].$$

From Theorem 2.2.2 it then follows that

$$\operatorname{sp}((1-\theta)\partial_{0}\Phi(\theta,\mathbf{x}^{*},\mathbf{x}^{*})) = \{-(1-\theta)g(\theta)\} \neq \{0\}.$$

Let R = $(\rho_{i,j})$ be a one-stage Runge-Kutta method with operator coefficient where $\rho_{2,1}(z) \equiv \left[1 + \frac{1}{g(\theta)(1-\theta)} z\right]^{-1}$. Let $H = \{h_0, \dots, h_N\}$, where $H = \{1\}$ if $\theta = 0$ and $H = \{\theta, (1-\theta)\}$ if $\theta \neq 0$. Let G = M(F) where M = IM (A,g,R,H). Then $h_N \partial_2 \Phi(t_N, x^*, x^*) \notin D_E(\rho_{2,1})$. Hence $x^* \notin D(G)$ (cf. (4.2.5e)).

Since any Runge-Kutta method with scalar coefficients is a Runge-Kutta method with operator coefficients, it follows from Theorem 5.2.3 and Lemma 5.3.18 that (i) implies (ii).

b. From Theorem 5.2.1 it follows that (ii) implies (i).

2. <u>PROOF OF THE THEOREMS 5.2.7, 5.2.8</u>. We shall show that (vi) implies (v), and that (iii) implies (iv).

a. From Theorem 5.2.2 it follows that (vi) implies (v).

b. Suppose (iii) holds. From Theorem 5.2.4 it follows that $\partial_1 K(x^*, x^*) = F'(x^*)$. From Theorem 5.2.6 it follows that g(t) = 0 (for all $t \in [0,1)$). This completes the proof.

CHAPTER 6

RADIUS OF CONVERGENCE

In this chapter we determine the radii of convergence of some of the iterative methods described in section 4.1.

In section 6.1 we introduce $F < \sigma, \beta, \gamma >$, the subclass of F_1 (see (2.6.1)) to which the radii of convergence to be considered in Part I of this chapter will be related. In section 6.2 we determine the radius of convergence of Newton's method. In section 6.3 a class of iterative methods, which are denoted by $M_{\overline{\omega}}$, is introduced. These have greater radii of convergence than Newton's method. Any method $M_{\overline{\omega}}$, which is a kind of damped Newton method, is of a type considered in subsection 5.2.2. We also give a result which shows that certain damped Newton methods have greater radii of convergence than Newton's method.

Part II of this chapter is concerned with $F < \sigma, \alpha >$, the subclass of F_1 that is introduced in section 6.4. In section 6.5 we determine the radius of convergence of Newton's method with respect to $F < \sigma, \alpha >$. As in Part I of this chapter, we investigate in section 6.6 the convergence behaviour of the iterative methods \underline{M}_{ω} (introduced in section 6.3) with respect to $F < \sigma, \alpha >$. In this case, we are able to give an explicit expression for the radii of convergence.

We finally note that we base the determination of the radius of convergence of an iterative method M with respect to a subclass F_0 of F_1 on the following principle. We first give a lower bound, say \hat{r} , of $r(M;F_0)$. Then we consider the case E = IR and construct an $f \in F_0$ for which $r(M,f) = \hat{r}$. When E is an arbitrary Hilbert space we "extend" f to an operator F on E such that $F \in F_0$ and $r(M,F) = \hat{r}$. The extension of f to F is described in subsection 6.2.3. PART I

6.1. THE CLASS $F < \sigma, \beta, \gamma >$

In the first part of this chapter we shall be concerned with the following subclass of F_1 (see 2.6.1)).

For given $\sigma \in (0,\infty]$ and $\beta,\gamma > 0$

$$(6.1.1) \qquad F < \sigma, \beta, \gamma > = \{F \mid F \in F_1; B(x^*, \sigma) \subset D(F); \|[F'(x^*)]^{-1}\| \le \beta; \}$$

 $\|\mathbf{F}^{"}(\mathbf{x})\| \leq \gamma \quad (\text{for all } \mathbf{x} \in \mathbf{B}(\mathbf{x}^{*}, \sigma)) \}.$

We notice that for any $F \in F_1$, numbers $\sigma \in (0,\infty]$ and $\beta,\gamma > 0$ exist such that $F \in F < \sigma, \beta, \gamma >$. Hence

(6.1.2)
$$F_{1} = \bigcup_{\substack{\sigma \in (0, \infty] \\ \beta, \gamma > 0}} F_{<\sigma, \beta, \gamma > .}$$

Let $\beta > 0$ and let $F \in L(E)$ with $F = \frac{1}{\beta} I$ (I is the identity). Obviously, $F \in F < \sigma, \beta, \gamma >$ (for all $\sigma \in (0, \infty], \gamma > 0$). Consequently, for each $\sigma \in (0, \infty]$ and all $\beta, \gamma > 0$, the set $F < \sigma, \beta, \gamma >$ is not empty.

6.2. The radius of convergence of Newton's method with respect to $F < \sigma, \beta, \gamma >$

Let $\sigma \in (0,\infty]$ and $\beta,\gamma > 0$. Let M be Newton's method, which means that for F $\in F_1$, the function G = M(F) is defined by

G: D(G) \rightarrow E, (6.2.1) D(G) = {x | x \in D(F); F'(x) is invertible}, G(x) = x - \Gamma(x)F(x) (x \in D(G)).

We recall that $\Gamma(x)$ denotes $[F'(x)]^{-1}$ (for $x \in D(G)$). THEOREM 6.2.1. Let M be the Newton's method, then

(6.2.2) $r(M;F<\sigma,\beta,\gamma>) = \min\{\sigma, \frac{2}{3\beta\gamma}\}.$

We shall prove Theorem 6.2.1 in the next subsections.

6.2.1 Proof of Theorem 6.2.1

In order to prove Theorem 6.2.1 we need the following two lemmata.

LEMMA 6.2.2. If $F \in F < \sigma, \beta, \gamma > and x \in B(x^*, \sigma) \cap B(x^*, \frac{1}{\beta\gamma})$ then F'(x) is invertible and

$$\| \left[\mathbf{F}'(\mathbf{x}) \right]^{-1} \| \leq \frac{\beta}{1 - \beta \gamma \| \mathbf{x} - \mathbf{x}^* \|}$$

<u>PROOF</u>. Since $\mathbf{F} \in F < \sigma, \beta, \gamma >$, it follows that $\| [\mathbf{F}'(\mathbf{x}^*)]^{-1} \| \le \beta$ and $\| \mathbf{F}'(\mathbf{x}) - \mathbf{F}'(\mathbf{x}^*) \| \le \gamma \| \mathbf{x} - \mathbf{x}^* \|$ (for all $\mathbf{x} \in \mathbf{B}(\mathbf{x}^*, \sigma) \cap \mathbf{B}(\mathbf{x}^*, \frac{1}{\beta \gamma})$), (cf. Lemma 2.6.1). Therefore, Lemma 4.2.1 applies, thus proving this lemma. \Box

LEMMA 6.2.3. For F \in F< σ , β , γ > let G be defined by (6.2.1). Then

$$B(x^{*},\sigma) \cap B(x^{*},\frac{1}{\beta\gamma}) \subset D(G)$$

and

$$\|\mathbf{G}(\mathbf{x}) - \mathbf{x}^{\star}\| \leq \frac{\beta \gamma \|\mathbf{x} - \mathbf{x}^{\star}\|^{2}}{2(1 - \beta \gamma \|\mathbf{x} - \mathbf{x}^{\star}\|)}$$

whenever $\mathbf{x} \in B(\mathbf{x}^*, \sigma) \cap B(\mathbf{x}^*, \frac{1}{\beta\gamma})$.

<u>PROOF</u>. The first part of the conclusion is a consequence of Lemma 6.2.2. Let $\mathbf{x} \in B(\mathbf{x}^*, \sigma) \cap B(\mathbf{x}^*, \frac{1}{\beta\gamma})$. From Lemma 5.3.1 it follows that $0 = F(\mathbf{x}^*) = F(\mathbf{x}) + F'(\mathbf{x})(\mathbf{x}^*-\mathbf{x}) + r(\mathbf{x})$, where $\|\mathbf{r}(\mathbf{x})\| \leq \frac{\gamma}{2} \|\mathbf{x}-\mathbf{x}^*\|^2$. Thus

$$\| \mathbf{x} - \Gamma(\mathbf{x}) F(\mathbf{x}) - \mathbf{x}^* \| = \| \Gamma(\mathbf{x}) r(\mathbf{x}) \|.$$

Using Lemma 6.2.2, we obtain

$$\|\Gamma(\mathbf{x})\mathbf{r}(\mathbf{x})\| \leq \frac{\beta\gamma \|\mathbf{x}-\mathbf{x}^{*}\|^{2}}{2(1-\beta\gamma \|\mathbf{x}-\mathbf{x}^{*}\|)} \cdot$$

This completes the proof. \Box

We are now able to prove the following lemma.

LEMMA 6.2.4. Let M be the Newton's method, then

$$r(M; F < \sigma, \beta, \gamma >) \ge \min\{\sigma, \frac{2}{3\beta\gamma}\}.$$

<u>PROOF</u>. Let $F \in F < \sigma, \beta, \gamma >$, and G = M(F). Thus G is defined by (6.2.1). For $\varepsilon \in (0, \frac{2}{3})$, set $\alpha(\varepsilon) = [\frac{2}{3} + 2\varepsilon]^{-1}[\frac{2}{3} - \varepsilon]$. Note that $0 < \alpha(\varepsilon) < 1$. From Lemma 6.2.3 it follows that for any $x \in B(x^*, \sigma) \cap \overline{B}(x^*, (\frac{2}{3} - \varepsilon)\frac{1}{\beta\gamma})$ we have $x \in D(G)$ and

(6.2.3)
$$\|G(x) - x^*\| \le \frac{\beta \gamma \|x - x^*\|^2}{2(1 - \beta \gamma \|x - x^*\|)} \le \frac{\frac{2}{3} - \varepsilon}{2(1 - (\frac{2}{3} - \varepsilon))} \|x - x^*\| = \alpha(\varepsilon) \|x - x^*\|.$$

If $x_0 \in B(x^*,\sigma) \cap B(x^*,\frac{2}{3\beta\gamma})$ then an $\varepsilon \in (0,\frac{2}{3})$ exists such that $x_0 \in B(x^*,\sigma) \cap \overline{B}(x^*,(\frac{2}{3}-\varepsilon)\frac{1}{\beta\gamma})$. The relation (6.2.3) shows that $x_0 \in D(M,F)$ (cf. (2.3.3)) and that the sequence $\{x_k\}$ generated by x_0 and [M,F] satisfies

$$\|\mathbf{x}_{\mathbf{k}} - \mathbf{x}^{\star}\| \leq \left[\alpha(\varepsilon)\right]^{\mathbf{k}} \|\mathbf{x}_{0} - \mathbf{x}^{\star}\| \neq 0 \qquad (\mathbf{k} \neq \infty) \,.$$

Hence $\mathbf{x}_0 \in S(\mathbf{M}, \mathbf{F})$ (cf. (2.3.5)), so that $\mathbf{r}(\mathbf{M}, \mathbf{F}) \ge \min\{\sigma, \frac{2}{3\beta\gamma}\}$. Since \mathbf{F} was an arbitrary element of $F < \sigma, \beta, \gamma >$ this proves the lemma.

Results similar to Lemmata 6.2.3 and 6.2.4 can also be found in [RHEINBOLDT, 1975].

<u>REMARK 6.2.1</u>. Notice that, if $F \in F < \sigma, \beta, \gamma >$, then at the same time, $F_{\sigma} \in F < \sigma, \beta, \gamma >$ where F_{σ} is the restriction of F to $B(x^*, \sigma)$, $(D(F_{\sigma}) = B(x^*, \sigma))$. Obviously $r(M, F_{\sigma}) \le \sigma$, so that $r(M; F < \sigma, \beta, \gamma >) \le \sigma$.

<u>REMARK 6.2.2</u>. Suppose E = IR and let $\sigma > \frac{2}{3\beta\gamma}$. In view of Lemma 6.2.4 and Remark 6.2.1 the proof of Theorem 6.2.1 is completed if we can show that an f ϵ F< σ , β , γ > and an x₀ ϵ D(f) with $||x_0 - x^*|| = \frac{2}{3\beta\gamma}$ exist such that x₀ \notin S(M,f). Consider

$$f_0: \mathbb{R} \to \mathbb{R}$$

$$f_{0}(\xi) = \begin{cases} \frac{1}{2\beta^{2}\gamma} & (\text{if } \xi \geq \frac{1}{\beta\gamma}), \\ \frac{1}{\beta} \xi - \frac{\gamma}{2} \xi^{2} & (\text{if } \xi \in [0, \frac{1}{\beta\gamma})), \\ \frac{1}{\beta} \xi + \frac{\gamma}{2} \xi^{2} & (\text{if } \xi \in (-\frac{1}{\beta\gamma}, 0)), \\ -\frac{1}{2\beta^{2}\gamma} & (\text{if } \xi \leq -\frac{1}{\beta\gamma}). \end{cases}$$

It is easily verified that

a) f_0 is continuously differentiable on \mathbb{R} , $f_0^{"}(\xi)$ exists and $|f_0^{"}(\xi)| \leq \gamma$ (for all $\xi \in \mathbb{R}$ with $\xi \notin \{0, -\frac{1}{\beta\gamma}, \frac{1}{\beta\gamma}\}$); b) the equation $f_0(\xi) = 0$ has a unique solution at $\xi^* = 0$ and $|[f_0^{"}(\xi^*)]^{-1}| = \beta$; c) with $\xi_0 = \frac{2}{3\beta\gamma}$ we have $f_0(\xi_0)$

$$\xi_0 - \frac{f_0(\xi_0)}{f_0'(\xi_0)} = -\xi_0$$
 and $-\xi_0 - \frac{f_0(\xi_0)}{f_0'(\xi_0)} = \xi_0$.

Consequently, $|\xi_0 - \xi^*| = \frac{2}{3\beta\gamma}$ and $\xi_0 \notin S(M, f_0)$. However, since $f_0 \notin F < \sigma, \beta, \gamma > (f_0 \text{ is not twice-differentiable at } \xi = 0, \xi = \frac{1}{\beta\gamma}$ and $\xi = -\frac{1}{\beta\gamma}$, this example does not complete the proof of Theorem 6.2.1. The proof can be completed by using the next lemma. It states that an f with the desired properties exists, which is not only twice but infinitely differentiable on D(f).

<u>LEMMA 6.2.5</u>. If $\sigma > \frac{2}{3\beta\gamma}$ then for any $\epsilon > 0$ there exists an $F \in F < \sigma, \beta, \gamma >$ which is infinitely differentiable on D(F), for which

$$r(M,F) < \frac{2}{3\beta\gamma} + \varepsilon$$

For the case E = IR we prove this lemma, with the function f_0 of Remark 6.2.2 in mind, in the next subsection. If $E \neq IR$ then Lemma 6.2.5 is proved in the subsection 6.2.3 (cf. p. 86).

In view of Lemma 6.2.4 and Remark 6.2.1 the proof of Theorem 6.2.1 is easily completed by application of Lemma 6.2.5.

6.2.2. Proof of Lemma 6.2.5 where E = IR

In this section we shall prove Lemma 6.2.5 if $E = \mathbb{R}$ with innerproduct $(\xi_1, \xi_2) = \xi_1 \xi_2$ (for all $\xi_1, \xi_2 \in \mathbb{R}$).

PART A

We start with a lemma which will be used subsequently.

LEMMA 6.2.6. Let $\phi: D(\phi) \rightarrow \mathbb{R}$ with $D(\phi) \subset \mathbb{R}$. Assume that $D(\phi) \supset (-\varepsilon, \varepsilon)$ for some $\varepsilon > 0$. Suppose that ϕ has a fixed point $\xi = 0$. Assume that ϕ is continuous on $D(\phi)$ and that $|\phi(\xi)| \leq |\xi|$ (for all $\xi \in \mathbb{R}$ with $0 \leq |\xi| < \varepsilon$). If some $\lambda > 0$ satisfies $[-\lambda, \lambda] \subset D(\phi)$, $|\phi(\lambda)| \geq \lambda$ and $|\phi(-\lambda)| \geq \lambda$, then a number $\xi_0 \in [-\lambda, \lambda]$ with $\xi_0 \neq 0$ exists for which $\phi(\xi_0) \in D(\phi)$ and $\phi[\phi(\xi_0)] = \xi_0$.

PROOF. If $\phi(\lambda) \ge \lambda$ or $\phi(-\lambda) \le -\lambda$ then $\phi(\xi_0) = \xi_0$ (for some $\xi_0 \in [-\lambda, \lambda]$ with $\xi_0 \ne 0$), so that the statement is true. Suppose $\phi(\lambda) \le -\lambda$ and $\phi(-\lambda) \ge \lambda$. Then a number $\delta_1 \in [-\lambda, 0)$ exists such that $\phi(\delta_1) = \lambda$ and a number $\delta_2 \in (0, \lambda]$ exists for which $\phi(\delta_2) = -\lambda$. We assume that $|\phi(\xi)| \le \lambda$ (for all $\xi \in [\delta_1, \delta_2]$). This is no restriction. Hence $\phi(\xi) \in D(\phi)$ (for all $\xi \in [\delta_1, \delta_2]$). Since $\phi[\phi(\delta_2)] = \phi(-\lambda) \ge \lambda$ and $|\phi[\phi(\xi)]| \le |\xi|$ (for all $\xi \in (0, \varepsilon)$), a number $\xi_0 \in (0, \delta_2]$ exists such that $\phi[\phi(\xi_0)] = \xi_0$. This proves the lemma.

PART B

We note that in Remark 6.2.2 we were not able to complete the proof of Theorem 6.2.1. But with the function f_0 in mind, we can construct an F that satisfies the proposition of Lemma 6.2.5.

We introduce some function classes. Let $\tau \in (0,\infty]$. We define

(6.2.4a) $C^{\infty}(-\tau,\tau) = \{f \mid f: (-\tau,\tau) \rightarrow \mathbb{R}; f \text{ is infinitely differentiable}$ on $(-\tau,\tau)\}.$

If $\tau = \infty$, we set

(6.2.4b) $C^{\infty}(IR) = C^{\infty}(-\tau, \tau)$.

We give two well-known results (see also [COURANT, 1961; p. 172]).

LEMMA 6.2.7. There exists a function $\psi \in C^{\infty}(\mathbb{R})$ such that

$$ψ(ξ) = 0$$
 (if ξ ≤ 0),
 $ψ(ξ) > 0$ (if ξ > 0).

PROOF. Let ψ : $\mathbb{R} \rightarrow \mathbb{R}$,

(6.2.5)
$$\psi(\xi) = \begin{cases} 0 & (\text{if } \xi \le 0), \\ e^{-1/\xi} & (\text{if } \xi > 0). \end{cases}$$

It is easily verified that the statement of the lemma holds. \Box

<u>LEMMA 6.2.8</u>. Let the numbers γ_1 , γ_2 , δ_1 and δ_2 be given. Suppose $\delta_1 < \delta_2$. Set $\overline{d} = (\gamma_1, \gamma_2, \delta_1, \delta_2)$. Then a monotone function $h_{\overline{d}} \in C^{\infty}(\mathbb{R})$ exists such that

(6.2.6)
$$h_{\overline{a}}(\xi) = \begin{cases} \gamma_1 & (\text{if } \xi \leq \delta_1), \\ \gamma_2 & (\text{if } \xi \geq \delta_2), \end{cases}$$

<u>PROOF</u>. We assume that $\gamma_1 \neq \gamma_2$ ($\gamma_1 = \gamma_2$ is a trivial case). Let $\phi(\xi) \equiv \psi(\delta_2 - \xi)\psi(\xi - \delta_1)$ where ψ is defined in (6.2.5). Set

 $h_{\overline{d}}: \mathbb{R} \to \mathbb{R},$

$$h_{\overline{a}}(\xi) = \gamma_1 + \left[\int_{-\infty}^{\xi} \phi(\tau) d\tau\right] \frac{\gamma_2 - \gamma_1}{\left[\int_{\delta_1}^{\xi_2} \phi(\tau) d\tau\right]} \cdot$$

It is easily verified that $h_{\overline{d}}$ has the properties stated.

PART C

Let σ , β and γ denote the constants introduced at the beginning of section 6.2, which also appear in Lemma 6.2.5. Let $\delta \in (0, \frac{1}{9})$ and set

$$\bar{\bar{a}}_{1} = (0, \gamma, -\frac{1}{\beta\gamma}, (-1+\delta)\frac{1}{\beta\gamma}),$$
$$\bar{\bar{a}}_{2} = (\gamma, -\gamma, -\frac{\delta}{\beta\gamma}, \frac{\delta}{\beta\gamma}),$$

77

$$\overline{d}_3 = (-\gamma, 0, (1-\delta)\frac{1}{\beta\gamma}, \frac{1}{\beta\gamma})$$

Let

$$\hat{\mathbf{h}}(\xi) = \begin{cases} \mathbf{h}_{\overline{\mathbf{d}}_{1}}(\xi) & (\text{if } \xi \leq -\frac{1}{2\beta\gamma}), \\ \mathbf{h}_{\overline{\mathbf{d}}_{2}}(\xi) & (\text{if } \xi \in (-\frac{1}{2\beta\gamma}, \frac{1}{2\beta\gamma})), \\ \mathbf{h}_{\overline{\mathbf{d}}_{3}}(\xi) & (\text{if } \xi \geq \frac{1}{2\beta\gamma}). \end{cases}$$

We define the function f by

f:
$$\mathbb{IR} \rightarrow \mathbb{IR}$$
,
(6.2.9)
 $f(\xi) = \frac{1}{\beta} \xi + \int_{0}^{\xi} (\xi - \tau) \hat{h}(\tau) d\tau$ ($\xi \in \mathbb{IR}$).



graph of \hat{h}	graph of f
Fig. 6.2.1	Fig. 6.2.2

The following lemma holds.

LEMMA 6.2.9. The function f defined in (6.2.9) has the following properties:

$$f(0) = 0, f \in C^{\infty}(\mathbf{IR}),$$

$$f'(\xi) > 0 \text{ (for all } \xi \in \mathbf{IR}) \text{ and }$$

$$f \in F < \sigma, \beta, \gamma > .$$

PROOF. Notice that $\hat{h} \in C^{\infty}(\mathbb{R})$. Furthermore, for all $\xi \in \mathbb{R}$ we have $f'(\xi) = \frac{1}{\beta} + \int_{0}^{\xi} \hat{h}(\tau) d\tau$, $f''(\xi) = \hat{h}(\xi)$ and $f^{(k)}(\xi) = \hat{h}^{(k-2)}(\xi)$ (k = 3, 4, ...). Consequently, $f \in C^{\infty}(\mathbb{R})$ and $f'(0) = \frac{1}{\beta}$. Furthermore, for all $\xi \in \mathbb{R}$ we have a) $|f''(\xi)| = |\hat{h}(\xi)| \leq \gamma$;

b) if $\xi > 0$, $f'(\xi) \ge \frac{1}{\beta} + \int_0^{1/\beta\gamma} \hat{h}(\tau) d\tau > \frac{1}{\beta} - \frac{1}{\beta} = 0$; if $\xi < 0$, $f'(\xi) \ge \frac{1}{\beta} + \int_0^{1/\beta\gamma} \hat{h}(\tau) d\tau > \frac{1}{\beta} - \frac{1}{\beta} = 0$.

Since f(0) = 0 it follows that $\xi = 0$ is the unique solution of $f(\xi) = 0$. This proves the lemma.

<u>LEMMA 6.2.10</u>. Let $\delta \in (0, \frac{1}{9})$. Let M be Newton's method and let f be defined in (6.2.9). Set $\phi = M(f)$. Then $D(\phi) = IR$. Further, a number $\xi_0 \in [-(\frac{2}{3}+2\delta)\frac{1}{\beta\gamma}, (\frac{2}{3}+2\delta)\frac{1}{\beta\gamma}]$ exists such that $\phi[\phi(\xi_0)] = \xi_0$ and $\xi_0 \notin S(M,f)$.

<u>PROOF.</u> From Lemma 6.2.9 and (6.2.1) it follows that $D(\phi) = IR$. Set $\lambda = (\frac{2}{3} + 2\delta)\frac{1}{\beta\gamma}$. (Thus $0 < \lambda < (\frac{2}{3} + \frac{2}{9})\frac{1}{\beta\gamma} < (1-\delta)\frac{1}{\beta\gamma}$.) Then $f(\lambda) > \frac{1}{\beta} \lambda - \frac{\gamma}{2} \lambda^2$ and $f'(\lambda) < \frac{1}{\beta} - \gamma(\lambda - \frac{\delta}{\beta\gamma})$. Hence

$$\begin{split} \lambda &- \frac{f(\lambda)}{f'(\lambda)} < \lambda - \frac{\frac{1}{\beta} \lambda - \frac{\gamma}{2} \lambda^2}{\frac{1}{\beta} - \gamma (\lambda - \frac{\delta}{\beta\gamma})} \\ &= \frac{(\frac{\beta\gamma}{2} \lambda - \delta)}{1 - \beta\gamma\lambda + \delta} (-\lambda) = \frac{(\frac{1}{3} + \delta - \delta)}{\frac{1}{3} - \delta} (-\lambda) < -\lambda. \end{split}$$

Analogously we can show that

$$-\lambda - \frac{f(-\lambda)}{f'(-\lambda)} > \lambda.$$

From Lemma 6.2.3 and Lemma 6.2.6 it follows that a number $\xi_0 \in [-\lambda, \lambda]$ exists such that $\phi[\phi(\xi_0)] = \xi_0$ and $\xi_0 \notin S(M, f)$. This proves the lemma.

By virtue of this last lemma, for all $\varepsilon > 0$ we are able to construct an F that satisfies the statement of Lemma 6.2.5 for the case E = IR.

In the next subsection we shall prove Lemma 6.2.5 where E is an arbitrary Hilbert space.

6.2.3. The E-extension of a real function

From [BROWN & PAGE, 1970; Theorem 9.2.13, 9.2.16] it follows that a subset B of E exists such that the following three statements (i), (ii) and (iii) hold.

- All $u, v \in B$ satisfy (u, v) = 0 (if $u \neq v$) and (u, v) = 1 (if u = v). (i)
- (ii) For any $x \in E$, the set $\dot{B}_x = \{u \mid u \in B; (x,u) \neq 0\}$ is countable.
- (iii) Let $x \in E$ and let $\dot{B}_x = \{ \hat{u} \mid u = u_n \text{ with } n \in \mathbb{N}_0 \}$. Here $\mathbb{N}_0 \subset \mathbb{N}$ is a set of consecutive integers with 1 $\in \mathbb{N}_0$. In the case that \mathbb{N}_0 contains only $\ell < \infty$ numbers, put $u_n = 0$ ($n = \ell + 1, \ell + 2, \ell + 3, ...$). Put $\mathcal{B}_{\mathbf{x}} = \{\mathbf{u} \mid \mathbf{u} = \mathbf{u}_{n} \text{ with } \mathbf{n} \in \mathbf{N}\}.$ Then

$$\mathbf{x} = \sum_{n=1}^{\infty} (\mathbf{x}, \mathbf{u}_n) \mathbf{u}_n$$

and

$$\|\mathbf{x}\|^{2} = \sum_{n=1}^{\infty} (\mathbf{x}, \mathbf{u}_{n})^{2}.$$

From now on, $\mathcal B$ denotes a fixed subset of E for which the above mentioned properties hold.

We call a sequence $\{j_n\}$ in \mathbb{N} in which every positive integer appears once and only once, a reordering of IN.

Let $\{u_n\} \subset \mathcal{B} \cup \{0\}$ satisfy $(u_i, u_j) = 0$ $(i, j \in \mathbb{N}, i \neq j)$.

LEMMA 6.2.11. Let the sequence $\{\eta_n\}$ in IR satisfy $\eta_n = 0$ if $u_n = 0$ (n = 1,2,...). Suppose $\sum_{n=1}^{\infty} \eta_n^2 < \infty$. Then the following four statements (i) - (iv) hold.

- (i) The sequence $\{\sum_{n=1}^{N} \eta_n u_n\}$ converges. Put $y = \sum_{n=1}^{\infty} \eta_n u_n$. (ii) Let $\{j_n\}$ be a reordering of IN. Then $\{\sum_{n=1}^{N} \eta_j u_n\}$ converges and
- (iv) Let c > 0. Suppose $\{\alpha_n\}$ is a sequence in \mathbb{R} that satisfies $|\alpha_n| \le c |\eta_n|$ (n = 1,2,...). Then $\{\sum_{n=1}^{N} \alpha_{n}^{n} u_{n}\}$ converges. Put $z = \sum_{n=1}^{\infty} \alpha_{n}^{n} u_{n}$. Then $\|z\| \le c \|y\|$.

<u>PROOF</u>. (i) Let ε > 0. Then an N₁ ϵ IN exists such that

(6.2.10)
$$\sum_{n=N+1}^{N+m} \eta_n^2 < \varepsilon \quad (\text{for all } N, m \in \mathbb{N} \text{ with } N \ge N_1).$$

Consequently,

$$\|\sum_{n=1}^{N+m} \eta_n u_n - \sum_{n=1}^{N} \eta_n u_n \|^2 = \sum_{n=N+1}^{N+m} \eta_n^2 < \epsilon \quad (\text{for all } N, m \in IN \text{ with} N \in IN \text{ or all } N, m \in IN \text{ or all } N \text{ or all } N, m \in IN \text{ or all } N \text{ or all } N, m \in IN \text{ or all } N \text{ or all$$

This proves (i).

(ii) Let $\varepsilon > 0$. Then an $\mathbb{N}_1 \in \mathbb{N}$ exists such that (6.2.10) holds. Choose $\mathbb{N}_2 \ge \mathbb{N}_1$ such that $n \in \{j_1, j_2, \dots, j_{N_2}\}$ ($n = 1, 2, \dots, \mathbb{N}_1$). Then $\|y - \sum_{n=1}^{N} \|y_n\|^2 \le \varepsilon$ (for all $\mathbb{N} \ge \mathbb{N}_2$). Hence (ii) holds. (iii) Suppose $\mathcal{B}_y = \{u \mid u = v_n \text{ with } n \in \mathbb{N}\}$ and $\zeta_n = (y, v_n)$ ($n = 1, 2, \dots$). It is easily verified that a reordering $\{j_n\}$ of \mathbb{N} exists, such that $\zeta_{j_n} = \eta_n$ ($n = 1, 2, \dots$). Consequently,

$$\|\mathbf{y}\|^{2} = \sum_{n=1}^{\infty} \zeta_{n}^{2} = \sum_{n=1}^{\infty} \zeta_{j}^{2} = \sum_{n=1}^{\infty} \eta_{n}^{2}$$

(cf. [RUDIN, 1953; Theorem 3.56]).

(iv) Let N, $m \ge 0$. Then

$$\|\sum_{n=1}^{N+m} \alpha_n u_n - \sum_{n=1}^{N} \alpha_n u_n\|^2 = \sum_{n=N+1}^{N+m} \alpha_n^2 \le c^2 \sum_{n=N+1}^{N+m} |\eta_n|^2.$$

Hence $\{\sum_{n=1}^{N} \alpha_n u_n^n\}$ is a Cauchy sequence and $\|z\| \leq c \|y\|$.

We shall now define the E-extension of a real function. Let $\tau \in (0,\infty].$ Let

(6.2.11a)
$$\phi \in C^{\infty}(-\tau,\tau)$$
.

Suppose

(6.2.11b) $\phi(0) = 0$.

Let $x \in B(0,\tau)$ and $B_x = \{u \mid u = u_n \text{ with } n \in \mathbb{N}\}$. Let the constant c > 0 satisfy

(6.2.12a) $c \ge |\phi'(\xi)|$ (for all $\xi \in \mathbb{R}$ with $|\xi| \le ||x||$).

Set $\xi_n = (x, u_n)$ (n = 1,2,...). Thus $x = \sum_{n=1}^{\infty} \xi_n u_n$ and $|\xi_n| \le ||x||$ (n = 1,2,...). From Lemma 2.6.1 it follows that

(6.2.12b)
$$|\phi(\xi_n)| \le c|\xi_n|$$
 (n = 1,2,...).

Lemma 6.2.11(iv) implies that $\{\sum_{n=1}^{N} \phi(\xi_n)u_n\}$ is a Cauchy sequence. We set

(6.2.13a)
$$F(x) = \sum_{n=1}^{\infty} \phi(\xi_n) u_n$$

for

(6.2.13b)
$$x = \sum_{n=1}^{\infty} \xi_n u_n$$
.

Let $\{\eta_n\}, \{\zeta_n\} \subset \mathbb{R}$. Let $\sum_{n=1}^{\infty} \eta_n^2 < \infty$ and $\sum_{n=1}^{\infty} \zeta_n^2 < \infty$. Suppose $\sum_{n=1}^{\infty} \eta_n v_n = \sum_{n=1}^{\infty} \zeta_n w_n = x$ with $v_n, w_n \subset \mathcal{B} \cup \{0\}$ (n = 1, 2, ...) and $(v_1, v_j) = (w_1, w_j) = 0$ (i,j $\in \mathbb{N}, i \neq j$). Then $F(x) = \sum_{n=1}^{\infty} \phi(\eta_n) v_n = \sum_{n=1}^{\infty} \phi(\zeta_n) w_n$ (cf. Lemma 6.2.11(ii)), For given ϕ satisfying (6.2.11) we call

F: $B(0,\tau) \rightarrow E$,

(6.2.14)

F(x) is defined in (6.2.13) (for all $x \in B(0,\tau)$),

the E-extension of ϕ .

THEOREM 6.2.12. Let ϕ satisfy (6.2.11) and suppose $\phi'(\xi) \neq 0$ (for all $\xi \ \varepsilon \ (-\tau,\tau)) \, .$ Let F be the E-extension of $\varphi .$ Then

where

(i) x = 0 is the unique solution of F(x) = 0.

(ii) F is infinitely differentiable on D(F). Let $k \in \mathbb{I}N$. For $x \in D(F)$ and $h_i \in E$ (j = 1,2,...,k) Let

$$B_{\mathbf{x}} \cup \{ \bigcup_{j=1}^{k} B_{h_{j}} \} = \{ u \mid u = v_{n} \text{ with } n \in \mathbb{IN} \}.$$

For $n = 1, 2, ..., let \xi_n = (x, v_n)$ and $h_{j,n} = (h_j, v_n)$ (j = 1, 2, ..., k). Then

(6.2.15a)
$$F^{(k)}(x)h_1h_2...h_k = \sum_{n=1}^{\infty} \phi^{(k)}(\xi_n)h_1, h_2, n...h_k, n^v n$$

Suppose the number c_k satisfies

(6.2.15b)
$$c_k \ge |\phi^{(k)}(\xi)|$$
 (for all $\xi \in \mathbb{R}$ with $|\xi| \le ||x||$)

Then

(6.2.15c)
$$\|\mathbf{F}^{(k)}(\mathbf{x})\| \leq c_k$$
.

(iii) F'(x) is invertible (for all $x \in D(F)$). Let $x \in D(F)$ and $y \in E$, and let $B_x \cup B_y = \{u \mid u = u_n \text{ with } n \in IN\}$. Let $\xi_n = (x, u_n)$ and $\eta_n = (y, u_n)$ (n = 1,2,...). Then

$$[\mathbf{F}'(\mathbf{x})]^{-1}\mathbf{y} = \sum_{n=1}^{\infty} [\phi'(\xi_n)]^{-1} \eta_n u_n.$$

In particular

(6.2.16)
$$[F'(x)]^{-1}F(x) = \sum_{n=1}^{\infty} [\phi'(\xi_n)]^{-1}\phi(\xi_n)u_n.$$

PROOF. (i) Obviously, x = 0 is a solution of F(x) = 0. Suppose F(x) = 0 has a solution $x = y^*$. Let $B_{y*} = \{u \mid u = v_n \text{ with } n \in IN\}$ and $\eta_n = (y^*, v_n)$ ($n = 1, 2, \ldots$). Then $F(y^*) = \sum_{n=1}^{\infty} \phi(\eta_n) v_n$. Consequent on Lemma 6.2.11(iii), $0 = \|F(y^*)\|^2 = \sum_{n=1}^{\infty} [\phi(\eta_n)]^2$. Hence $\phi(\eta_n) = 0$, so that $\eta_n = 0$ ($n = 1, 2, \ldots$). This implies $y^* = 0$. (ii) Suppose $F^{(k)}(x)$ exists and satisfies (6.3.15a,c) (for all $x \in D(F)$), with $k \ge 0$ (if k = 0 then (6.3.15a) should read $F(x) = \sum_{n=1}^{\infty} \phi(\xi_n) v_n$).

Let $x \in D(F)$. Let the positive numbers δ and ε satisfy $||x|| + \varepsilon \le \delta < \tau$. We notice that a constant c > 0 exists such that

(6.2.17)
$$|\phi^{(k)}(\xi+\eta) - \phi^{(k)}(\xi) - \phi^{(k+1)}(\xi)\eta| \le c\eta^2$$
 (for all $\xi, \eta \in \mathbb{R}$
with $|\xi| + |\eta| \le \delta$).

Let the number c_{k+1} satisfy

(6.2.18)
$$|\phi^{(k+1)}(\xi)| \leq c_{k+1}$$
 (for all $\xi \in \mathbb{R}$ with $|\xi| \leq ||\mathbf{x}||$).

Let $y, h_1, h_2, \ldots, h_k \in E$. Let $\mathcal{B}_x \cup \mathcal{B}_y \cup \{ \bigcup_{i=1}^k \mathcal{B}_{h_i} \} = \{ u \mid u = v_n \text{ with } n \in \mathbb{N} \}$. For $n = 1, 2, \ldots$, let

(6.2.19)
$$\xi_n = (x, v_n), \quad \eta_n = (y, v_n) \text{ and } h_{i,n} = (h_i, v_n) \quad (i = 1, 2, ..., k).$$

Set $\alpha_n = \phi^{(k+1)}(\xi_n)\eta_n h_{1,n}h_{2,n} \dots h_{k,n}$, then $|\alpha_n| \leq c_{k+1} \|h_1\| \|h_2\| \dots \|h_k\| \|\eta_n|$ (n = 1,2,...). From Lemma 6.2.11(iv) it follows that $\{\Sigma_{n=1}^N \alpha_n v_n\}$ converges. Set $z = \sum_{n=1}^{\infty} \alpha_n v_n$. Then

(6.2.20a)
$$z = \sum_{n=1}^{\infty} \phi^{(k+1)}(\xi_n) \eta_n h_{1,n} h_{2,n} \cdots h_{k,n} v_n$$

and

(6.2.20b)
$$\|z\| \le c_{k+1} \|y\| \|h_1\| \dots \|h_k\|$$

Suppose $B_x \cup B_y \cup \{\bigcup_{i=1}^k B_{h_i}\} = \{u \mid u = \widetilde{v}_n \text{ with } n \in \mathbb{N}\}$, and for $n = 1, 2, ..., \widetilde{\xi}_n = (x, \widetilde{v}_n), \widetilde{\eta}_n = (y, \widetilde{v}_n)$ and $\widetilde{h}_{i,n} = (h_i, \widetilde{v}_n)$ (i = 1, 2, ..., k). Then a reordering $\{j_n\}$ of \mathbb{N} exists such that for n = 1, 2, ..., k have $\widetilde{v}_n = v_j$, $\widetilde{\xi}_n = \xi_{j_n}$, $\widetilde{\eta}_n = \eta_{j_n}$ and $\widetilde{h}_{i,n} = h_{i,j_n}$ (i = 1, 2, ..., k). Let $\widetilde{\alpha}_n = \phi^{(k+1)}(\widetilde{\xi}_n)\widetilde{\eta}_n\widetilde{h}_{1,n}\widetilde{h}_{2,n} \cdots$ $\dots \widetilde{h}_{k,n}$ (n = 1, 2, ...). By virtue of Lemma 6.2.11(ii), the sequence $\{\sum_{n=1}^{N} \widetilde{\alpha}_n \widetilde{v}_n\}$ converges and $\sum_{n=1}^{\infty} \widetilde{\alpha}_n \widetilde{v}_n = z$. Let

$$Q \in L^{(k+1)}(E)$$

(6.2.21)

In view of (6.2.20b), $\|Q\| \leq c_{k+1}$. We show that $Q = F^{(k+1)}(x)$. Let $y, h_1, h_2, \dots, h_k \in E$ satisfy (6.2.19). Suppose $\|y\| \leq \varepsilon$ and $\|h_1\| = \|h_2\| = \dots \|h_k\| = 1$. (Notice that $\{\sum_{n=1}^{\infty} \eta_n^2\}^2 \geq \{\sum_{n=1}^{N} \eta_n^2\}^2 \geq \sum_{n=1}^{N} \eta_n^4$ (for all $N \in \mathbb{N}$).) Then (cf. (6.2.17))

$$\| \mathbf{F}^{(\mathbf{k})} (\mathbf{x}+\mathbf{y}) \mathbf{h}_{1} \mathbf{h}_{2} \cdots \mathbf{h}_{\mathbf{k}} - \mathbf{F}^{(\mathbf{k})} (\mathbf{x}) \mathbf{h}_{1} \mathbf{h}_{2} \cdots \mathbf{h}_{\mathbf{k}} - \mathcal{Q} \mathbf{y} \mathbf{h}_{1} \mathbf{h}_{2} \cdots \mathbf{h}_{\mathbf{k}} \|^{2}$$

$$= \sum_{n=1}^{\infty} \left[\phi^{(\mathbf{k})} (\xi_{n} + \eta_{n}) - \phi^{(\mathbf{k})} (\xi_{n}) - \phi^{(\mathbf{k}+1)} (\xi_{n}) \eta_{n} \right]^{2} \mathbf{h}_{1,n}^{2} \cdots \mathbf{h}_{\mathbf{k},n}^{2}$$

$$\le \mathbf{c}^{2} \sum_{n=1}^{\infty} \eta_{n}^{4} \le \mathbf{c}^{2} \| \mathbf{y} \|^{4}.$$

Hence,

$$\|\mathbf{F}^{(k)}(\mathbf{x}+\mathbf{y}) - \mathbf{F}^{(k)}(\mathbf{x}) - \mathbf{Q}\mathbf{y}\| \le c \|\mathbf{y}\|^2 \quad (\text{for all } \mathbf{y} \in \mathbf{E} \text{ with } \|\mathbf{y}\| \le \epsilon).$$

This implies that $Q = F^{(k+1)}(x)$. Therefore (6.2.15) holds for k+1. Consequently, (6.2.15) holds for all $k \in \mathbb{N}$. (iii) Let $x \in D(F)$. Then a constant c exists such that

(6.2.22)
$$|[\phi'(\xi)]^{-1}| \leq c$$
 (for all $\xi \in \mathbb{R}$ with $|\xi| \leq ||\mathbf{x}||$).

For $y \in E$ let $\mathcal{B}_x \cup \mathcal{B}_y = \{u \mid u = v_n \text{ with } n \in \mathbb{N}\}$. Let

(6.2.23)
$$\xi_n = (x, v_n)$$
 and $\eta_n = (y, v_n)$ $(n = 1, 2, ...)$.

According to (6.2.22) and Lemma 6.2.11(iv), the sequence $\{\Sigma_{n=1}^{N} [\phi'(\xi_{n})]^{-1}\eta_{n}v_{n}\}$ converges, and with

(6.2.24a)
$$z = \sum_{n=1}^{\infty} [\phi'(\xi_n)]^{-1} \eta_n v_n$$

we have

```
(6.2.24b) \|z\| \le c\|y\|.
```

Suppose $B_{\mathbf{x}} \cup B_{\mathbf{y}} = \{u \mid u = \widetilde{v}_{n} \text{ with } n \in \mathbb{N}\}$. Let $\widetilde{\xi}_{n} = (\mathbf{x}, \widetilde{v}_{n})$ and $\widetilde{\eta}_{n} = (\mathbf{y}, \widetilde{v}_{n})$ (n = 1,2,...). From Lemma 6.2.11(ii) it follows that $\sum_{n=1}^{\infty} [\phi'(\widetilde{\xi}_{n})]^{-1} \widetilde{\eta}_{n} v_{n} = z$.

Let

C:
$$E \rightarrow E$$
,

(6.2.25)

$$Cy = z$$
 (z is defined in (6.2.24a) with $y \in E$ satisfying (6.2.23)).

Obviously C is linear in y and from (6.2.24b) it follows that C is bounded. Moreover

F'(x)Cy = CF'(x)y = y (for all $y \in E$).

This completes the proof. \Box

85 .

Let the constants σ , β and γ denote the constants appearing in Lemma 6.2.5, which were introduced at the beginning of section 6.2.

PROOF OF LEMMA 6.2.5. Let $\delta \in (0, \frac{1}{9})$. Let f be defined in (6.2.9). F is the E-extension of f. From Lemma 6.2.9 and Theorem 6.2.12 it follows that the equation $F(\mathbf{x}) = 0$ has a unique solution $\mathbf{x} = 0$, that F is infinitely differentiable on E, that F'(0) is invertible and $[F'(0)]^{-1} = \beta I$, and that $\|F''(\mathbf{x})\| \leq \gamma$ (for all $\mathbf{x} \in \mathbf{E}$). Hence $F \in F < \sigma, \beta, \gamma >$. Let $\mathbf{u} \in \mathbf{B}$. Then $\Gamma(\lambda \mathbf{u})F(\lambda \mathbf{u}) = [f'(\lambda)]^{-1}f(\lambda)\mathbf{u}$ (for all $\lambda \in \mathbf{IR}$), (cf. (6.2.16)). Set G = M(F). Let ϕ be defined in Lemma 6.2.10. It follows that $G(\lambda \mathbf{u}) = \phi(\lambda)\mathbf{u}$ (for all $\lambda \in \mathbf{IR}$). According to Lemma 6.2.10 a number $\xi_0 \in [-(\frac{2}{3} + 2\delta)\frac{1}{\beta\gamma}, (\frac{2}{3} + 2\delta)\frac{1}{\beta\gamma}]$ with $\xi_0 \neq 0$ exists for which $\phi[\phi(\xi_0)] = \xi_0$. Consequently, $G(G(\xi_0 \mathbf{u})) = \xi_0 \mathbf{u}$. Hence $\xi_0 \mathbf{u} \notin S(M,F)$. This proves Lemma 6.2.5.

6.3. ITERATIVE METHODS WITH GREATER RADII OF CONVERGENCE

Let $\sigma \in (0,\infty]$ and $\beta,\gamma > 0$. In this section we present a class of iterative methods (for F_1) which all have greater radii of convergence with respect to $F < \sigma, \beta, \gamma >$ than Newton's method when $\sigma > \frac{2}{3\beta\gamma}$.

Let

(6.3.1a) $A \in A$ with $[A(F)](y,x) \equiv F(y) - F(x)$ (for all $F \in F_1$),

let

(6.3.1b)
$$g \in S$$
 with $g(t) = \begin{cases} \frac{1}{1-t} & (\text{if } t \in [0,1)), \\ 1 & (\text{if } t = 1) \end{cases}$

and let

(6.3.1c) L be Euler's method

(cf. (2.6.1), (2.6.3) and (2.6.9)).

In this section we shall be concerned, among other things, with the iterative methods M that satisfy

(6.3.2) M = IM (A,g,L,H)

(cf. (4.1.3)). Here $H = \{h_0, h_1, \dots, h_N\}$ with $N \in \mathbb{N}$.

Let M satisfy (6.3.2) and let $F \in F_1$. We notice that proposition (vi) of Theorem 5.2.5 with K = A(F) is true. Consequently, the iterative process [M,F] is quadratically convergent.

For F ϵ F₁, the function G = M(F) is defined by (cf. (4.1.4))

G: D(G)
$$\rightarrow$$
 E,

(6.3.3a)

$$G = n_{N+1}$$
.

In (6.3.3a) the function $\boldsymbol{\eta}_{N+1}$ is defined as follows:

$$\begin{array}{ll} (6.3.3b) & D(n_0) \rightarrow E, \\ & \eta_0(x) = D(F), \\ & \eta_0(x) = x \quad (\text{for all } x \in D(n_0)) \end{array}$$

and for n = 0,1,...,N the functions $\boldsymbol{\eta}_{n+1}$ are defined by

$$\eta_{n+1}: D(\eta_{n+1}) \rightarrow E,$$

$$(6.3.3c) \quad D(\eta_{n+1}) = \{x \mid x \in D(\eta_n), \eta_n(x) \in D(F) \text{ and } F'(\eta_n(x)) \text{ is invertible}\},\$$

$$\eta_{n+1}(\mathbf{x}) = \eta_n(\mathbf{x}) - \omega_n \Gamma(\eta_n(\mathbf{x})) F(\eta_n(\mathbf{x})) \quad (\text{for all } \mathbf{x} \in D(\eta_{n+1})).$$

In (6.3.3c)

(6.3.4)
$$\omega_n = \frac{h_n}{1-t_n}$$
 (n = 0,1,...,N).

Let

$$(6.3.5) \qquad \Omega = \{ \overline{\omega} \mid \overline{\omega} = (\omega_0, \omega_1, \dots, \omega_N) \text{ with } N \in \mathbb{I}N; \\ \omega_n \in (0, 1) \quad (n = 0, 1, \dots, N-1) \text{ and } \omega_N = 1 \}.$$

DEFINITION 6.3.1. Let $\bar{\omega} \in \Omega$ with $\bar{\omega} = (\omega_0, \omega_1, \dots, \omega_N)$ and N ϵ IN. By M₋ we denote the iterative method for which, for all F ϵ F₁, the function G = M_{$\bar{\omega}$}(F) is defined in (6.3.3).

Let
$$H = \{h_0, h_1, \dots, h_N\}$$
 with N \in IN. Since $0 < h_n < 1-t_n$

 $(n = 0, 1, \dots, N-1)$ and $h_N = 1 - t_N$, it follows that $\mathbb{IM}(A, g, L, H) = M_{\overline{\omega}}$. Here $\overline{\omega} = (\omega_0, \omega_1, \dots, \omega_N)$ and ω_n $(n = 0, 1, \dots, N)$ is defined by (6.3.4).

Conversely, let $\bar{\omega} \in \Omega$ with $\bar{\omega} = (\omega_0, \omega_1, \dots, \omega_N)$. Then the sequence $\mathcal{H} = \{h_0, h_1, \dots, h_N\}$ where h_n (n = 0,1,...,N) satisfies (6.3.4), is uniquely determined. Consequently, $M_{\bar{\omega}} = IM(A,g,L,\mathcal{H})$.

For notational convenience, instead of iterative methods M of the type (6.3.2) we shall consider the iterative methods M_{ω} ($\overline{\omega} \in \Omega$). In view of the above considerations, this is no restriction.

The following theorem holds.

THEOREM 6.3.1. Let $\overline{\omega} \in \Omega$. Then $M_{\overline{\omega}}$ is quadratically convergent. Further

$$\begin{split} \mathbf{r} \left(\mathbf{M}_{\overline{\omega}} ; \ F < \sigma, \beta, \gamma > \right) &= \sigma \qquad (\text{if } \sigma \leq \frac{2}{3\beta\gamma}) \,, \\ \mathbf{r} \left(\mathbf{M}_{\overline{\omega}} ; \ F < \sigma, \beta, \gamma > \right) \, > \, \frac{2}{3\beta\gamma} \qquad (\text{if } \sigma > \frac{2}{3\beta\gamma}) \,. \end{split}$$

We shall prove this theorem in the next subsection.

Let $F \in F_1$ (cf. (2.6.1)). A well-known class of numerical methods for solving F(x) = 0 are the so-called *damped Newton methods*. For given $x_0 \in D(F)$ these methods generate a sequence $\{x_n\}$ for which

(6.3.6)
$$x_{n+1} = x_n - \omega_n \Gamma(x_n) F(x_n)$$
 (n = 0,1,2,...)

(cf. [ORTEGA & RHEINBOLDT, 1970; chapter 14.4]). Here $0 < \omega_n \le 1$ (n = 0,1, 2,...) and $\omega_k < 1$ for at least one k. Obviously, the methods $M_{\overline{\omega}}$ ($\overline{\omega} \in \Omega$) are damped Newton methods with periodic ccefficients ω_n and with $\omega_0 < 1$, $\omega_1 < 1$, ..., $\omega_{N-1} < 1$, $\omega_N = 1$ for some N ≥ 1 .

If $\sigma > \frac{2}{3\beta\gamma}$, one may conjecture that the general damped Newton methods have a greater radius of convergence with respect to $F < \sigma, \beta, \gamma >$ than Newton's method (cf. [ORTEGA & RHEINBOLDT, 1970; Theorem 14.4.4], [DEUFLHARD, 1974]). This means that, for an $F \in F < \sigma, \beta, \gamma >$, these methods are expected to generate sequences $\{x_n\}$ that converge to x^* , even if $\|x_0 - x^*\| \ge \frac{2}{3\beta\gamma}$. Theorem 6.3.1 shows that this holds for the damped Newton methods $M_{\overline{\omega}}$. The next theorem shows that the conjecture is also true for all damped Newton methods with $\inf\{\omega_n \mid n \ge 0\} > 0$.

<u>THEOREM 6.3.2</u>. Let $0 < \omega_n \le 1$ (n = 0,1,2,...) and $\omega_k < 1$ for at least one k. Let $\inf\{\omega_n \mid n \ge 0\} > 0$. Then a constant $\hat{\mu}$ exists such that the following

propositions hold.

(i) $\hat{\mu} = \sigma$ (if $\hat{\mu} \leq \frac{2}{3\beta\gamma}$) and $\hat{\mu} > \frac{2}{3\beta\gamma}$ (if $\sigma > \frac{2}{3\beta\gamma}$).

(ii) Let $F \in F < \sigma, \beta, \gamma >$. Then for any $x_0 \in B(x^*, \hat{\mu})$ a sequence $\{x_n\}$ exists such that for n = 0, 1, 2, ...

 $x_n \in D(F)$, $F'(x_n)$ is invertible,

$$\mathbf{x}_{n+1} = \mathbf{x}_n - \boldsymbol{\omega}_n \Gamma(\mathbf{x}_n) F(\mathbf{x}_n)$$
,

and $\lim_{n\to\infty} x_n = x^*$.

We shall prove this theorem in the next subsection.

6.3.1. Proof of the Theorems 6.3.1 and 6.3.2

In order to prove the Theorems 6.3.1 and 6.3.2 we require the following lemmata.

LEMMA 6.3.3. Let P: D(P) \rightarrow E with D(P) an open subset of E. Let $x_0 \in D(P)$. Suppose that P is twice differentiable on D(P) and that P'(x_0) is invertible. Suppose positive constants α_0 , β_0 , γ_0 and τ exist such that

(i)
$$\|[P'(x_0)]^{-1}\| \le \beta_0, \|[P'(x_0)]^{-1}P(x_0)\| \le \alpha_0,$$

(ii)
$$h < \frac{1}{2}$$
 where $h = \alpha_0 \beta_0 \gamma_0$

(iii)
$$B(x_0,\tau) \subset D(P)$$
 and $\|P''(x)\| \leq \gamma_0$ (for all $x \in B(x_0,\tau)$),

(iv)
$$r_1 < \tau < r_2$$
 with $r_i = [1 + (-1)^i \sqrt{1-2h}]/(\beta\gamma)$ (i = 1,2)

Under these assumptions the equation P(x) = 0 has a solution \tilde{x} in $\bar{B}(x_0, r_1)$ and \tilde{x} is the unique solution of P(x) = 0 in $B(x_0, \tau)$.

PROOF. The conclusion is a direct consequence of the well-known Newton-Kantorovich theorem (cf. [KANTOROWITSCH & AKILOW, 1964; Theorem 6(1.XVIII)], [GRAGG & TAPIA, 1974]).

LEMMA 6.3.4. If u,x,y,z \in E and z = ωx + (1- ω)y with $\omega \in$ IR, then

$$\|\mathbf{z}-\mathbf{u}\|^{2} = \omega \|\mathbf{x}-\mathbf{u}\|^{2} + (1-\omega) \|\mathbf{y}-\mathbf{u}\|^{2} - \omega (1-\omega) \|\mathbf{x}-\mathbf{y}\|^{2}.$$

PROOF. Observe that the following relations (a) and (b) hold.

(a) $\|z-u\|^2 = \|\omega(x-u) + (1-\omega)(y-u)\|^2 = \omega^2 \|x-u\|^2 + (1-\omega)^2 \|y-u\|^2 + 2\omega(1-\omega)(x-u,y-u).$ (b) $\|x-y\|^2 = \|(x-u) - (y-u)\|^2 = \|x-u\|^2 + \|y-u\|^2 - 2(x-u,y-u).$

Therefore,

$$2(x-u,y-u) = ||x-u||^2 + ||y-u||^2 - ||x-y||^2$$
.

Together with relation (a), this proves the lemma. $\hfill \square$

Throughout this subsection

(6.3.7)
$$\hat{\sigma} = \min\{\sigma, \frac{1}{\beta\gamma}\}.$$

Let $\omega \in (0,1]$. Define

$$\delta_{(i)}: [0, \hat{\sigma}) \rightarrow [0, \infty),$$

(6.3.8)

$$\delta_{\omega}(\varepsilon) = (1-\omega) + \omega \left[\frac{\beta \gamma \varepsilon}{2(1-\beta \gamma \varepsilon)} \right]^{2} - \omega (1-\omega) \left[\frac{(2-\beta \gamma \varepsilon)}{2(1+\beta \gamma \varepsilon)} \right]^{2} \quad (\varepsilon \in [0, \hat{\sigma})).$$

The following lemma holds.

<u>LEMMA 6.3.5</u>. δ_{ω} is continuous and strictly isotone on $[0, \hat{\sigma})$, and $\delta_{\omega}(0) = (1-\omega)^2$.

PROOF. The proof of this lemma is straightforward. $\hfill \Box$

As a consequence of this lemma we have

(6.3.9)
$$\delta_{\omega}(\varepsilon) \geq 0$$
 (for all $\varepsilon \in [0, \hat{\sigma})$).

Let the function θ_{ω} be defined by

$$\boldsymbol{\theta}_{\boldsymbol{\omega}} \colon \left[\boldsymbol{0}, \boldsymbol{\hat{\sigma}}\right) \to \left[\boldsymbol{0}, \boldsymbol{\infty}\right),$$

(6.3.10)

$$\theta_{\omega}(\varepsilon) = \sqrt[4]{\delta_{\omega}(\varepsilon)} \qquad (\varepsilon \in [0, \hat{\sigma})).$$

Let

(6.3.11a) $\mu_{\omega} = \sup\{\varepsilon \mid \varepsilon \in [0, \hat{\sigma}); \theta_{\omega}(\varepsilon) < 1\}$

and

(6.3.11b)
$$\pi_{\omega} = \sup\{\varepsilon \mid \varepsilon \in [0,\sigma); \theta_{\omega}(\varepsilon)\varepsilon < \sigma\}.$$

Obviously, $\mu_{\omega} \leq \pi_{\omega} \leq \hat{\sigma}$ (see also Fig. 6.3.1).

LEMMA 6.3.6. Let $\omega \in (0,1]$. Let θ_{ω} be defined by (6.3.10). The following statements (i), (ii) and (iii) hold.

- (i) $\theta_{\omega}(0) = 1 \omega$ and θ_{ω} is continuous and strictly isotone on $[0, \hat{\sigma})$.
- (ii) If $\hat{\sigma} > \frac{2}{3\beta\gamma}$ then $\theta_{\omega}(\frac{2}{3\beta\gamma}) < 1$ (if $\omega \in (0,1)$) and $\theta_{\omega}(\frac{2}{3\beta\gamma}) = 1$ (if $\omega = 1$). If $\hat{\sigma} \leq \frac{2}{3\beta\gamma}$ then $\lim_{\epsilon \uparrow \hat{\sigma}} \theta_{\omega}(\epsilon) < 1$ (if $\omega \in (0,1)$) and $\lim_{\epsilon \uparrow \hat{\sigma}} \theta_{\omega}(\epsilon) \leq 1$ (if $\omega = 1$).

(iii)
$$\mu_{\omega} > \frac{2}{3\beta\gamma}$$
 (if $\hat{\sigma} > \frac{2}{3\beta\gamma}$) and $\mu_{\omega} = \hat{\sigma}$ (if $\hat{\sigma} \le \frac{2}{3\beta\gamma}$).

PROOF. By virtue of Lemma 6.3.5 it follows that $\theta_{\omega}(0) = 1 - \omega$ and that θ_{ω} is continuous and strictly isotone on $[0,\hat{\sigma})$. If $\hat{\sigma} > \frac{2}{3\beta\gamma}$ then it is easily verified that $\theta_{\omega}(\frac{2}{3\beta\gamma}) < 1$ (if $\omega \in (0,1)$) and $\theta_{\omega}(\frac{2}{3\beta\gamma}) = 1$ (if $\omega = 1$). This completes the proof. \Box



Fig. 6.3.1.

$$\hat{\sigma} = \frac{1}{\beta\gamma}; \omega \in (0,1)$$
.

The following important lemma holds.

LEMMA 6.3.7. Let $F \in F < \sigma, \beta, \gamma >$ and $\omega \in (0,1]$. Let θ_{ω} be defined by (6.3.10). Let $y \in B(x^*, \hat{\sigma})$. Then F'(y) is invertible. Set

 $z = y - \omega \Gamma(y) F(y)$.

Then the following estimate holds.

$$||_{z-x}^{*}|| \leq \theta_{\omega}(||_{y-x}^{*}||)||_{y-x}^{*}||.$$

Further, if $\|\mathbf{y} - \mathbf{x}^{\star}\| \leq \rho < \mu_{\omega}$ then

 $\|z - x^*\| \le \|y - x^*\|.$

Here κ = $\theta_{\omega}(\rho)$ and it follows that $\kappa \in [0,1)$.

<u>PROOF</u>. 1. According to Lemma 6.2.2 the derivative F'(y) is invertible. Set $\varepsilon = \|y - x^*\|$ and let

 $\mathbf{v} = \mathbf{y} - \Gamma(\mathbf{y}) \mathbf{F}(\mathbf{y}) \, .$

Then $z = \omega v + (1-\omega)y$. According to Lemma 6.3.4 we have

(6.3.12)
$$\|z-x^*\|^2 = (1-\omega)\|y-x^*\|^2 + \omega\|v-x^*\|^2 - \omega(1-\omega)\|\Gamma(y)F(y)\|^2$$

Using Lemma 6.2.3 we thus obtain

(6.3.13a)
$$\|\mathbf{z}-\mathbf{x}^{*}\|^{2} \leq (1-\omega)\varepsilon^{2} + \omega \left[\frac{\beta\gamma\varepsilon^{2}}{2(1-\beta\gamma\varepsilon)}\right]^{2} - \omega(1-\omega)\Delta^{2}$$

where

$$(6.3.13b) \qquad \Delta = \| \Gamma(\mathbf{y}) F(\mathbf{y}) \|.$$

In order to show that $\|z - x^*\| \le \theta_{\omega}(\epsilon)\epsilon$ it is sufficient to prove that

(6.3.14)
$$\Delta \geq \frac{(2-\beta\gamma\epsilon)\epsilon}{2(1+\beta\gamma\epsilon)}$$
 (cf. (6.3.8), (6.2.10))

2. Suppose

(6.3.15) $0 < 2\Delta\beta\gamma(1+\beta\gamma\epsilon) < \beta\gamma\epsilon(2-\beta\gamma\epsilon)$.

Consider

P: D(F) → E,
(6.3.16)
$$P(x) = F(x) - F(y)$$
 (x ∈ D(F)).

Then P'(x) = F'(x) (for all $x \in D(F)$). Consequently, $P'(x^*)$ is invertible and

$$[P'(x^{*})]^{-1}P(x^{*}) = -\Gamma(x^{*})F(y)$$

= $\Gamma(x^{*})[F'(x^{*}) - F'(y)]\Gamma(y)F(y) - \Gamma(y)F(y).$

Hence, with $\alpha = \Delta(1+\beta\gamma\epsilon)$, we have

(6.3.17)
$$\| [P'(\mathbf{x}^*)]^{-1} P(\mathbf{x}^*) \| \leq \alpha$$
.

Set $h = \alpha\beta\gamma$. According to (6.3.15) we have

$$(6.3.18) \qquad 0 < h < \frac{1}{2} .$$

Since $2h < \beta \gamma \epsilon (2-\beta \gamma \epsilon)$ it follows that

$$1 - \sqrt{1-2h} < 1 - \sqrt{1-2\beta\gamma\varepsilon+(\beta\gamma\varepsilon)^2} = \beta\gamma\varepsilon.$$

Consequently,

(6.3.19)
$$\frac{1-\sqrt{1-2h}}{\beta\gamma} < \varepsilon < \hat{\sigma} < \frac{1+\sqrt{1-2h}}{\beta\gamma}$$
.

From (6.3.18) and (6.3.19) it follows that Lemma 6.3.3 applies (with $\tau = \hat{\sigma}$). Hence P(x) = 0 has a unique solution \tilde{x} in B($x^*, \hat{\sigma}$) and $\tilde{x} \in \bar{B}(x^*, r_1)$. Since P(y) = 0 and y $\in B(x^*, \hat{\sigma})$, it follows that $\tilde{x} = y$, so that $y \in \bar{B}(x^*, r_1)$. Therefore

$$\|\mathbf{x}^* - \mathbf{y}\| \le \mathbf{r}_1 = \frac{1 - \sqrt{1 - 2h}}{\beta \gamma} \le \epsilon$$
 (cf. (6.3.19)).

This yields a contradiction. Consequently (6.3.14) holds. Hence $\|_{Z-x}^{*}\| \leq \theta_{\omega}(\varepsilon)\varepsilon$.

If $\varepsilon \le \rho < \mu_{\omega}$, then $\theta_{\omega}(\varepsilon) \le \theta_{\omega}(\rho) < 1$ (cf. Lemma 6.3.6(i)). This completes the proof.

PROOF OF THEOREM 6.3.2.

<u>PART A.</u> A number $k \ge 0$ exists for which $\omega_k < 1$. For n = 0, 1, ..., k set $\theta_n = \theta_{\omega_n}$ and $\pi_n = \pi_{\omega_n}$ where θ_{ω_n} and π_{ω_n} are defined by (6.3.10) and (6.3.11b) respectively. For n = 0, 1, ..., k let

$$\phi_n: [0, \hat{\sigma}] \rightarrow [0, \hat{\sigma}],$$

(6.3.20)

$$\phi_{n}(\varepsilon) = \begin{cases} \theta_{n}(\varepsilon)\varepsilon & (\text{if } \varepsilon \in [0,\pi_{j})), \\ \hat{\sigma} & (\text{if } \varepsilon \in [\pi_{j},\hat{\sigma}]). \end{cases}$$

Consider the following function.

 ψ : $[0,\hat{\sigma}] \rightarrow [0,\hat{\sigma}]$, (6.3.21a)

ψ

$$(\varepsilon) = \varepsilon_{k+1}$$

Herewith we define the quantities ε_n (n = 0,1,...,k+1) by

(6.3.21b)
$$\varepsilon_0 = \varepsilon$$

and

(6.3.21c)
$$\epsilon_{n+1} = \phi_n(\epsilon_n)$$
 (n = 0,1,...,k).

The following lemma holds.

LEMMA 6.3.8.

- (i) The function ψ defined by (6.3.21) is isotone on $[0,\hat{\sigma}]$ and continuous on $[0,\hat{\sigma})$.
- (ii) A constant $\hat{\mu}$ exists such that $\hat{\mu} = \hat{\sigma}$ (if $\hat{\sigma} \le \frac{2}{3\beta\gamma}$), $\frac{2}{3\beta\gamma} \le \hat{\mu} \le \hat{\sigma}$ (if $\hat{\sigma} > \frac{2}{3\beta\gamma}$) and $\lim_{\epsilon \neq \hat{\mu}} \psi(\epsilon) \le \min\{\hat{\sigma}, \frac{2}{3\beta\gamma}\}$.

(iii) Let $\varepsilon \in [0, \hat{\mu})$. Let ε_n (n = 0, 1, ..., k) be defined by (6.3.21b,c). Then $\varepsilon_n < \pi_n \le \hat{\sigma}$ (n = 0, 1, ..., k).

<u>PROOF</u>. From Lemma 6.3.6(i) it follows that ψ is isotone on $[0, \hat{\sigma}]$ and continuous on $[0, \hat{\sigma}]$. It is easily verified that $\psi(\frac{2}{3\beta\gamma}) < \frac{2}{3\beta\gamma}$ (if $\hat{\sigma} > \frac{2}{3\beta\gamma}$) and $\lim_{\epsilon \neq \hat{\sigma}} \psi(\epsilon) < \hat{\sigma}$ (if $\hat{\sigma} \leq \frac{2}{3\beta\gamma}$), (cf. Lemma 6.3.6(ii)). Consequently, a constant $\hat{\mu}$ exists such that statement (ii) is true.

Let $\varepsilon \in [0, \hat{\mu})$ and let ε_n (n = 0,1,...,k) be defined by (6.3.21b,c). Suppose $\varepsilon_{n_0} \ge \pi_{n_0}$ for some n_0 with $0 \le n_0 \le k$. Then $\phi_{n_0}(\varepsilon_{n_0}) = \hat{\sigma}$ and it is easily verified that $\psi(\varepsilon) = \hat{\sigma}$. This yields a contradiction.

<u>PART B.</u> Let $\hat{\mu}$ satisfy statement (ii) of Lemma 6.3.8. Let F ϵ F $\langle \sigma, \beta, \gamma \rangle$. Let $\mathbf{x}_0 \in B(\mathbf{x}^*, \hat{\mu})$ and set $\epsilon = \|\mathbf{x}_0 - \mathbf{x}^*\|$. Let ϵ_n (n = 0,1,...,k+1) be defined by (6.3.21b,c). It is easily verified that for n = 0,1,...,k $\|\mathbf{x} - \mathbf{x}^*\| \leq \epsilon_n < \pi_n < \hat{\sigma}$, F'(\mathbf{x}_n) is invertible, and, with

$$\mathbf{x}_{n+1} = \mathbf{x}_n - \boldsymbol{\omega}_n \Gamma(\mathbf{x}_n) F(\mathbf{x}_n)$$

we have

$$\|\mathbf{x}_{n+1} - \mathbf{x}^{\star}\| \le \theta_n (\|\mathbf{x}_n - \mathbf{x}^{\star}\|) \|\mathbf{x}_n - \mathbf{x}^{\star}\| \le \theta_n (\varepsilon_n) \varepsilon_n = \phi_n (\varepsilon_n) = \varepsilon_{n+1}$$

(cf. Lemma 6.3.7, Lemma 6.3.6(i) and Lemma 6.3.8(iii)). Hence

$$\|\mathbf{x}_{k+1} - \mathbf{x}^{\star}\| \leq \psi(\|\mathbf{x}_0 - \mathbf{x}^{\star}\|) < \min\{\widehat{\sigma}, \frac{2}{3\beta\gamma}\}.$$

<u>PART C</u>. From part B of the proof it follows that a number $\rho \in (0, \frac{2}{3\beta\gamma})$ with $\rho < \hat{\sigma}$ exists such that $x_{k+1} \in B(x^*, \rho)$. Set

$$\hat{\theta} = \sup\{\theta_{\omega_n}(\rho) \mid n \ge k+1\}.$$

Since $\rho < \mu_{\omega}$ (for all $\omega \in (0,1]$), (cf. Lemma 6.3.6(iii)), it follows that $\theta_{\omega}(\rho) < 1$ (for all $\omega \in (0,1]$). Further, $\theta_{\omega}(\rho)$ is continuous in ω (for all $\omega \in [\lambda,1]$ with $\lambda = \inf\{\omega_n \mid n \ge k+1\}$). Hence $\hat{\theta} < 1$. From Lemma 6.3.7 it follows that for $n = k+1, k+2, \ldots$

$$\|y - \omega_n^{T}(y)F(y) - x^*\| \le \hat{\theta}\|y - x^*\| \quad (\text{for all } y \in B(x^*, \rho)).$$

Consequently, the statement of the theorem holds. \Box

PROOF OF THEOREM 6.3.1. Theorem 6.3.1 is a direct consequence of Theorem 6.3.2 and Remark 6.2.1.

PART II

Let M denote Newton's method and let the method $M_{\overline{\alpha}}$ (with $\overline{\omega} \in \Omega$) be defined by Definition 6.3.1. In part I of this chapter we were able to prove that $r(M_{\overline{\omega}}; F < \sigma, \beta, \gamma >) \ge \dot{r}(M; F < \sigma, \beta, \gamma >)$ (for all $\sigma \in (0, \infty]$ and $\beta, \gamma > 0$). However, we did not manage to determine $r(M_{\overline{\omega}}; F < \sigma, \beta, \gamma >)$. In this second part of chapter 6 we shall present the subclass $F < \sigma, \alpha > (\sigma \in (0, \infty] \text{ and } \alpha \in [-\frac{1}{2}, \frac{1}{2}])$ of F_1 . It will appear that in this case we can determine not only $r(M; F < \sigma, \alpha >)$ but also $r(M_{\overline{\omega}}; F < \sigma, \alpha >)$.

Before we introduce $F<\sigma,\alpha>$ we observe the following. Let F \in F_1 with F \in $L_1(E)$ where

$$(6.4.0.1) \quad L_1(E) = \{F \mid F: E \to E, \text{ and } F(x) \equiv b + L(x) \text{ with } b \in E, L \in L(E) \}.$$

Let M be Newton's method. It is easily verified that, with G = M(F), we have $G(\mathbf{x}) = \mathbf{x}^*$ (for all $\mathbf{x} \in E$). Consequently, for any starting point $\mathbf{x}_0 \in E$, Newton's method requires only one step to solve $F(\mathbf{x}) = 0$. More generally speaking, let $\sigma \in (0, \infty]$, let $F \in F_1$ and suppose that the following holds:

$$B(x^{*},\sigma) \subset D(F), \quad F'(x) \text{ is invertible and}$$

$$(6.4.0.2)$$

$$(x-x^{*},\Gamma(x)F(x)) = \frac{1}{2} \{ \|x-x^{*}\|^{2} + \|\Gamma(x)F(x)\|^{2} \} \quad (\text{for all } x \in B(x^{*},\sigma)).$$

A little calculation shows that (6.4.0.2) implies $\|G(\mathbf{x}) - \mathbf{x}^{\star}\|^2 = 0$ (for all $\mathbf{x} \in B(\mathbf{x}^{\star}, \sigma)$). Hence Newton's method requires only one step to solve $F(\mathbf{x}) = 0$ whenever $\mathbf{x}_0 \in B(\mathbf{x}^{\star}, \sigma)$. The subclass $F < \sigma, \frac{1}{2} > \text{ consists of all functions } F$ for which (6.4.0.2) is true. Further $F < \sigma, \frac{1}{2} > C < F < \sigma, \alpha_2 > C < F < \sigma, \alpha_1 > C < F < \sigma, -\frac{1}{2} > = F_1$ (for all $\alpha_1, \alpha_2 \in [-\frac{1}{2}, \frac{1}{2}]$ with $\alpha_1 < \alpha_2$). See also Fig. 6.4.1 on page 99. Consequently, $\mathbf{r}(\mathbf{M}; F < \sigma, \cdot >)$ is isotone on $[-\frac{1}{2}, \frac{1}{2}]$, $\mathbf{r}(\mathbf{M}; F < \sigma, -\frac{1}{2} >) = 0$ (cf. (6.1.2) and Theorem 6.2.1) and $\mathbf{r}(\mathbf{M}; F < \sigma, \frac{1}{2} >) = \sigma$.

6.4. THE CLASS $F < \sigma, \alpha >$

Before we present the class $F{<}\sigma,\alpha{>}$ we give two lemmata.

LEMMA 6.4.1. Let $F \in F_1$. Then for any $x \in D(F)$ for which F'(x) is invertible we have

(6.4.1)
$$-\frac{1}{2} \{ \|\mathbf{x}-\mathbf{x}^{*}\|^{2} + \|\Gamma(\mathbf{x})F(\mathbf{x})\|^{2} \} \le (\mathbf{x}-\mathbf{x}^{*},\Gamma(\mathbf{x})F(\mathbf{x})) \le \frac{1}{2} \{ \|\mathbf{x}-\mathbf{x}^{*}\|^{2} + \|\Gamma(\mathbf{x})F(\mathbf{x})\|^{2} \}.$$

Further, positive numbers ρ and δ exist such that $B(x^{\star},\rho)\subset D(F)$, and F'(x) is invertible and

(6.4.2)
$$(\mathbf{x}-\mathbf{x}^{*},\Gamma(\mathbf{x})F(\mathbf{x})) \geq [\frac{1}{2} - \delta \|\mathbf{x}-\mathbf{x}^{*}\|^{2}] \{\|\mathbf{x}-\mathbf{x}^{*}\|^{2} + \|\Gamma(\mathbf{x})F(\mathbf{x})\|^{2}\}$$

(for all $\mathbf{x} \in B(\mathbf{x}^{*},\rho)$).

<u>PROOF</u>. 1. Let $x \in D(F)$ and suppose F'(x) is invertible. Then

$$(\mathbf{x} - \mathbf{x}^{*}, \Gamma(\mathbf{x}) F(\mathbf{x})) - \frac{1}{2} \{ \| \mathbf{x} - \mathbf{x}^{*} \| - \| \Gamma(\mathbf{x}) F(\mathbf{x}) \| \}^{2} \le (\mathbf{x} - \mathbf{x}^{*}, \Gamma(\mathbf{x}) F(\mathbf{x}))$$

$$\le (\mathbf{x} - \mathbf{x}^{*}, \Gamma(\mathbf{x}) F(\mathbf{x})) + \frac{1}{2} \{ \| \mathbf{x} - \mathbf{x}^{*} \| - \| \Gamma(\mathbf{x}) F(\mathbf{x}) \| \}^{2}.$$

From Schwarz's inequality (see [KANTOROWITSCH & AKILOW, 1964; section 7.1 (II)]) it follows that

$$-\|_{\mathbf{X}-\mathbf{X}}^{\star}\|\|\Gamma(\mathbf{x})F(\mathbf{x})\| \leq (\mathbf{x}-\mathbf{x}^{\star},\Gamma(\mathbf{x})F(\mathbf{x})) \leq \|_{\mathbf{X}-\mathbf{X}}^{\star}\|\|\Gamma(\mathbf{x})F(\mathbf{x})\|.$$

Hence (6.4.1) holds.

2. Let M denote Newton's method. From Lemma 6.2.3 and (6.1.2) it follows that [M,F] is quadratically convergent. Hence $\rho, \delta_1 > 0$ exist such that $B(x^*, \rho) \subset D(F)$, and F'(x) is invertible and

$$\|\mathbf{x}-\Gamma(\mathbf{x})F(\mathbf{x})-\mathbf{x}^{*}\|^{2} \leq \delta_{1}^{2}\|\mathbf{x}-\mathbf{x}^{*}\|^{4} \qquad (\text{for all } \mathbf{x} \in B(\mathbf{x}^{*},\rho)).$$

Consequently,

$$2(\mathbf{x}-\mathbf{x}^{*},\Gamma(\mathbf{x})F(\mathbf{x})) \geq \|\mathbf{x}-\mathbf{x}^{*}\|^{2} + \|\Gamma(\mathbf{x})F(\mathbf{x})\|^{2} - \delta_{1}^{2}\|\mathbf{x}-\mathbf{x}^{*}\|^{4}$$
(for all $\mathbf{x} \in B(\mathbf{x}^{*},\rho)$).

~

Therefore, with $\delta = \frac{1}{2} \delta_1^2$, (6.4.2) is true.

Let

$$\chi: \ F_1 \times (0,\infty] \times \mathbb{R} \to \mathbb{R},$$

(6.4.3)

$$\chi(\mathbf{F};\sigma,\alpha) = \begin{cases} \inf_{\substack{\mathbf{x}=\mathbf{x}^{*}, \mathbf{f}(\mathbf{x}) \in \mathbf{f}(\mathbf{x}) \\ \mathbf{0} < \|\mathbf{x}=\mathbf{x}^{*}\| < \sigma \| \|\mathbf{x}=\mathbf{x}^{*}\|^{2} + \|\mathbf{f}(\mathbf{x}) \mathbf{F}(\mathbf{x})\|^{2}} - \alpha \\ (\inf_{\substack{\mathbf{x}=\mathbf{x}^{*}, \sigma \\ \mathbf{x}=\mathbf{x}^{*}} - \sigma) \in \mathbf{D}(\mathbf{F}) \text{ and } \mathbf{F}'(\mathbf{x}) \text{ is invertible} \\ \text{for all } \mathbf{x} \in \mathbf{B}(\mathbf{x}^{*},\sigma)), \\ -\frac{1}{2} - \alpha \text{ (otherwise).} \end{cases}$$

The following lemma holds.

LEMMA 6.4.2. For any F \in F $_1$ we have

(6.4.4)
$$-\frac{1}{2} - \alpha \leq \chi(\mathbf{F};\sigma,\alpha) \leq \frac{1}{2} - \alpha$$
 (for all $\sigma \in (0,\infty]$ and $\alpha \in \mathbb{R}$)

and

(6.4.5)
$$\lim_{\sigma \neq 0} \chi(F;\sigma,\alpha) = \frac{1}{2} - \alpha \quad (\text{for all } \alpha \in \mathrm{IR}).$$

PROOF. This lemma is a direct consequence of Lemma 6.4.1.

Let $F \in F_1$ and $\sigma \in (0,\infty]$. From (6.4.4) it follows that

 $(6.4.6) \qquad \chi(F;\sigma,\alpha) \geq 0 \qquad (\text{for all } \alpha \leq -\frac{1}{2}),$

(6.4.7) $\chi(F;\sigma,\alpha) < 0$ (for all $\alpha > \frac{1}{2}$).

For $\sigma \in (0,\infty]$ and $\alpha \in [-\frac{1}{2},\frac{1}{2}]$ we define

$$(6.4.8) \qquad F < \sigma, \alpha > = \{F \mid F \in F_1; \chi(F;\sigma,\alpha) \ge 0\}.$$

When $\sigma \in (0,\infty]$ and $\alpha \in (-\frac{1}{2},\frac{1}{2}]$ then the following relation holds.
$$(6.4.9) \qquad F < \sigma, \alpha > = \{F \mid F \in F_1; B(x^*, \sigma) \subset D(F); F'(x) \text{ is invertible and} \\ (x-x^*, \Gamma(x)F(x)) \ge \alpha \{\|x-x^*\|^2 + \|\Gamma(x)F(x)\|^2\} \text{ (for all } x \in B(x^*, \sigma))\}.$$

Further

(6.4.10)
$$F < \sigma_r - \frac{1}{2} > = F_1$$
 (for all $\sigma \in (0, \infty]$).

Let $\sigma \in (0,\infty]$. We notice that

(6.4.11)
$$F < \sigma, \alpha_1 > \supset F < \sigma, \alpha_2 >$$
 (for all numbers α_1, α_2 with
 $-\frac{1}{2} \le \alpha_1 \le \alpha_2 \le \frac{1}{2}$).

Let L_1 (E) be defined by (6.4.0.1). It is easily verified that

(6.4.12)
$$F_1 \cap L_1(E) \subset F < \sigma, \frac{1}{2}$$
 (for all $\sigma \in (0, \infty]$).

Thus for each $\sigma \in (0,\infty]$ and $\alpha \in [-\frac{1}{2},\frac{1}{2}]$, the set $F < \sigma, \alpha >$ is not empty. Let $\sigma \in (0,\infty)$ and $\alpha \in (-\frac{1}{2},\frac{1}{2}]$. Let $F \in F < \sigma, \alpha >$ and let $F_{\sigma/2}$ denote the restriction of F to $B(x^*,\sigma/2)$. Then $F_{\sigma/2} \in F_1$ but $F_{\sigma/2} \notin F < \sigma, \alpha >$ (cf. (6.4.9)). Consequently

(6.4.13)
$$\emptyset \notin F < \sigma, \alpha > \notin F_1$$
 (for all $\sigma \in (0, \infty]$ and $\alpha \in (-\frac{1}{2}, \frac{1}{2}]$).

Fig. 6.4.1 illustrates the properties of $F<\sigma,\alpha>$ listed above.



REMARK 6.4.1.

1. Let $\sigma \in (0,\infty]$. It can be shown that $F < \sigma, \alpha_1 > \neq F < \sigma, \alpha_2 >$ whenever $\alpha_1, \alpha_2 \in [0, \frac{1}{2}]$ and $\alpha_1 \neq \alpha_2$ (this is a consequence of the Lemmata 6.6.8 and 6.6.11).

2. If E = IR, then it is easily verified that $F < \sigma, \alpha > = F < \sigma, 0 >$ whenever $\alpha \in (-\frac{1}{2}, 0]$ and $\sigma \in (0, \infty]$.

If $E \neq \mathbb{R}$, then this need not be the case. We shall illustrate this for the case $E = \mathbb{R}^2$ with innerproduct $(x,y) \equiv \xi_1 \eta_1 + \xi_2 \eta_2$ (for $x = (\xi_1, \xi_2)$, $y = (\eta_1, \eta_2) \in \mathbb{R}^2$). Let

$$F: \mathbb{R}^2 \to \mathbb{R}^2$$
,

(6.4.14)

$$F(\mathbf{x}) = (\xi_2^4(1+\xi_1^2)^4 + \xi_1(1+\xi_1^2), \xi_2(1+\xi_1^2)) \quad (\text{for all } \mathbf{x} \in \mathbb{R}^2,$$

where $\mathbf{x} = (\xi_1, \xi_2)$.

It is easily verified that F(0) = 0, $F \in F_1$ and that F'(x) is invertible (for all $x \in IR^2$). With $x_0 = (2,1)$ it can be shown that $-\frac{1}{2} < \alpha_0 < 0$, where

$$\boldsymbol{\alpha}_{0} = (\mathbf{x}_{0}, \Gamma(\mathbf{x}_{0}) \mathbf{F}(\mathbf{x}_{0})) / \{ \| \mathbf{x}_{0} \|^{2} + \| \Gamma(\mathbf{x}_{0}) \mathbf{F}(\mathbf{x}_{0}) \|^{2} \}.$$

Consequently, since $\|\mathbf{x}_0\| = \sqrt{5}$, we have $\chi(\mathbf{F};\sqrt{5},\alpha_0) < 0$. From Lemma 6.4.2 it follows that $\lim_{\sigma \neq 0} \chi(\mathbf{F};\sigma,\alpha_0) = \frac{1}{2} - \alpha_0 > 0$. It can be shown that in this case $\chi(\mathbf{F};\sigma,\alpha_0)$ is continuous on $(0,\infty]$. Consequently, a number $\sigma_0 \in (0,\sqrt{5}]$ exists such that $\mathbf{F} \in F < \sigma_0, \alpha_0 >$ and $\mathbf{F} \notin F < \sigma_0, 0 >$. Hence $F < \sigma_0, \alpha_0 > \neq F < \sigma_0, 0 >$.

Even if E is an arbitrary Hilbertspace with $E \neq IR$, using the above example it can be shown that numbers α and σ with $-\frac{1}{2} < \alpha < 0$ and $\sigma > 0$ exist such that $F < \sigma, \alpha > \neq F < \sigma, 0 >$.

Let $\sigma \in (0,\infty]$ and F $\in F_1$. Notice that (cf. (6.4.3) and (6.4.4))

(6.4.15) $\chi(\mathbf{F};\sigma,\cdot)$ is strictly antitone and continuous on $\left[-\frac{1}{2},\frac{1}{2}\right]$, and $\chi(\mathbf{F};\sigma,-\frac{1}{2}) \ge 0$.

We define

$$\begin{aligned} \mathbf{a:} \ \ F_1 \times (0,\infty] \to [-\frac{1}{2},\frac{1}{2}], \\ (6.4.16) \\ \mathbf{a}(\mathbf{F};\sigma) \ = \ \sup\{\alpha \ \big| \ \alpha \ \epsilon \ [-\frac{1}{2},\frac{1}{2}]; \ \chi(\mathbf{F};\sigma,\alpha) \ \ge \ 0\} \quad (\mathbf{F} \ \epsilon \ F_1, \ \sigma \ \epsilon \ (0,\infty]). \end{aligned}$$

Let $F \in F_1$. It is easily verified that (cf. Fig. 6.4.2 and (6.4.5))

(6.4.17) $a(F; \cdot)$ is antitone on $(0, \infty]$ and $\lim_{\sigma \neq 0} a(F; \sigma) = \frac{1}{2}$.

From (6.4.15) and (6.4.16) it follows that

(6.4.18) $F \in F < \sigma, a(F; \sigma) >$ (for all $\sigma \in (0, \infty]$).

Let
$$\alpha \in \left[-\frac{1}{2}, \frac{1}{2}\right)$$
 and F $\in F_1$. Notice that (cf. (6.4.3) and (6.4.5))

(6.4.19) $\chi(\mathbf{F};\cdot,\alpha)$ is antitone on $(0,\infty]$, left-continuous on $(0,\infty)$, and $\lim_{\sigma \neq 0} \chi(\mathbf{F};\sigma,\alpha) = \frac{1}{2} - \alpha > 0.$

We define

(6.4.20)

$$s: F_{1} \times [-\frac{1}{2}, \frac{1}{2}) \rightarrow (0, \infty],$$

$$s(F; \alpha) = \sup\{\sigma \mid \sigma \in (0, \infty]; \chi(F; \sigma, \alpha) \ge 0\} \quad (F \in F_{1}, \alpha \in [-\frac{1}{2}, \frac{1}{2})).$$

Let F ϵ F₁. It is easily verified that (cf. Fig. 6.4.2)

(6.4.21)
$$s(F; \cdot)$$
 is antitone on $[-\frac{1}{2}, \frac{1}{2}]$.

From (6.4.19) and (6.4.20) it follows that

(6.4.22)
$$F \in F < s(F; \alpha), \alpha >$$
 (for all $\alpha \in [-\frac{1}{2}, \frac{1}{2}]$).

6.5. THE RADIUS OF CONVERGENCE OF NEWTON'S METHOD WITH RESPECT TO $F\!\!<\!\!\sigma,\!\alpha\!\!>$

Let $\sigma \in (0,\infty]$ and $\alpha \in \left[-\frac{1}{2},\frac{1}{2}\right]$.

THEOREM 6.5.1. Let M be the Newton's method. Then

$$r(M; F<\sigma, \alpha>) = \begin{cases} 0 & (if \alpha \le \frac{2}{5}), \\ \sigma & (if \alpha > \frac{2}{5}). \end{cases}$$

We shall prove this theorem in the next subsection.

6.5.1. Proof of Theorem 6.5.1

First, we introduce an auxiliary function, which will be used frequently in this chapter. Let

$$\theta: (0, \frac{1}{2}] \times \{1, 2\} \to \mathbb{R},$$

(6.5.1)

$$\theta(\xi,k) = \frac{1 + (-1)^k \sqrt{1 - 4\xi^2}}{2\xi} \qquad ((\xi,k) \in D(\theta)).$$

Direct computation shows that the following relations hold.

(6.5.2)
$$\xi \{1 + \theta(\xi, k)^2\} = \theta(\xi, k)$$
 (k = 1,2)

and

(6,5,3)
$$\theta(\xi,1) = \left[\theta(\xi,2)\right]^{-1}$$
 (for all $\xi \in (0,\frac{1}{2}]$).

The following relations, that are easily verified, will be used subsequently.

(6.5.4)

$$\theta(\cdot, 2) \text{ is strictly antitone on } (0, \frac{1}{2}], \lim_{\xi \neq 0} \theta(\xi, 2) = \infty,$$

 $\theta(\frac{2}{5}, 2) = 2, \ \theta(\frac{1}{2}, 2) = 1.$

(6.5.5)
$$\theta(\cdot,1)$$
 is strictly isotone on $(0,\frac{1}{2}]$, $\theta(\frac{2}{5},1) = \frac{1}{2}$, $\theta(\frac{1}{2},1) = 1$.

The following lemma holds.

LEMMA 6.5.2. Suppose $\alpha \in (0, \frac{1}{2}]$. Let $F \in F < \sigma, \alpha >$. Then $B(\mathbf{x}^*, \sigma) \subset D(F)$, and $F'(\mathbf{x})$ is invertible and

 $(6.5.6) \qquad \theta(\alpha,1) \|_{\mathbf{X}-\mathbf{X}^{\star}} \| \leq \|\Gamma(\mathbf{X})F(\mathbf{X})\| \leq \theta(\alpha,2) \|_{\mathbf{X}-\mathbf{X}^{\star}} \| \quad (\text{for all } \mathbf{X} \in B(\mathbf{X}^{\star},\sigma)).$

<u>PROOF</u>. In view of (6.4.9) we have $B(x^*,\sigma) \subset D(F)$. Let $x \in B(x^*,\sigma)$. Then F'(x) is invertible and (cf. (6.4.9))

$$(x-x^{*},\Gamma(x)F(x)) \geq \alpha \{ \|x-x^{*}\|^{2} + \|\Gamma(x)F(x)\|^{2} \}.$$

Using Schwarz's inequality, we obtain

$$\alpha \| \Gamma(\mathbf{x}) \mathbf{F}(\mathbf{x}) \|^2 - \| \mathbf{x} - \mathbf{x}^* \| \| \Gamma(\mathbf{x}) \mathbf{F}(\mathbf{x}) \| + \alpha \| \mathbf{x} - \mathbf{x}^* \|^2 \leq 0.$$

This implies (6.5.6). \Box

Let

$$\psi: (0, \frac{1}{2}] \to \mathbb{R},$$
(6.5.7)

$$\psi(\xi) = \theta(\xi, 2) - 1 \qquad (\xi \in (0, \frac{1}{2}]).$$

From (6.5.4) it follows that

(6.5.8) ψ is strictly antitone on $(0, \frac{1}{2}]$ and $\psi(\frac{2}{5}) = 1$.

The following lemma holds.

LEMMA 6.5.3. Suppose $\alpha \in (0, \frac{1}{2}]$. Let $F \in F < \sigma, \alpha >$. Set G = M(F) where M is Newton's method. Then $B(x^*, \sigma) \subset D(G)$ and

(6.5.9) $\|\mathbf{G}(\mathbf{x}) - \mathbf{x}^{\star}\| \leq \psi(\alpha) \|\mathbf{x} - \mathbf{x}^{\star}\|$ (for all $\mathbf{x} \in \mathbf{B}(\mathbf{x}^{\star}, \sigma)$).

<u>PROOF</u>. In view of (6.4.9) we have $B(x^*,\sigma) \subset D(G)$. Let $x \in B(x^*,\sigma)$. It follows that (cf. (6.4.9))

$$\|G(x) - x^{*}\|^{2} = \|x - \Gamma(x)F(x) - x^{*}\|^{2}$$

= $\|x - x^{*}\|^{2} + \|\Gamma(x)F(x)\|^{2} - 2(x - x^{*}, \Gamma(x)F(x))$
 $\leq (1 - 2\alpha) \{\|x - x^{*}\|^{2} + \|\Gamma(x)F(x)\|^{2}\}.$

Hence, using Lemma 6.5.2,

(6.5.10) $\|G(\mathbf{x}) - \mathbf{x}^*\|^2 \le (1 - 2\alpha) \{1 + \theta(\alpha, 2)^2\} \|\mathbf{x} - \mathbf{x}^*\|^2.$

Since $[\psi(\alpha)]^2 = 1 + \theta(\alpha, 2)^2 - 2\theta(\alpha, 2)$, from (6.5.2) and (6.5.10) it follows that (6.5.9) is true.

A consequence of Lemma 6.5.3 is the following lemma.

LEMMA 6.5.4. Let M be the Newton's method. Then

$$r(M;F<\sigma,\alpha>) \geq \begin{cases} 0 & (\text{if } \alpha \in [-\frac{1}{2},\frac{2}{5}]), \\ \sigma & (\text{if } \alpha \in (\frac{2}{5},\frac{1}{2}]). \end{cases}$$

<u>PROOF</u>. Suppose $\alpha \in (\frac{2}{5}, \frac{1}{2}]$ and let $F \in F < \sigma, \alpha >$. Notice that ψ is strictly antitone on $(0, \frac{1}{2}]$ and that $\psi(\frac{2}{5}) = 1$ (cf. (6.5.8)). Hence, in view of Lemma 6.5.3, we have $B(\mathbf{x}^*, \sigma) \subset S(M, F)$. This proves the lemma.

<u>REMARK 6.5.1</u>. Suppose E = IR and suppose $\alpha \in (0, \frac{2}{5}]$. In view of Lemma 6.5.4 and Remark 6.2.1 the proof of Theorem 6.5.1 is completed if we can show that for all $\varepsilon > 0$ there exist an f ϵ $F < \sigma, \alpha >$ and an $x_0 \in D(f)$ with $\|x - x^*\| < \varepsilon$, such that $x_0 \notin S(M, f)$. Consider

f:
$$\mathbb{IR} \rightarrow \mathbb{IR}$$
,

$$f(\xi) = \begin{cases} \frac{1}{\xi^{\theta(\alpha,2)}} & (\text{if } \xi > 0), \\ 0 & (\text{if } \xi = 0), \\ -f(-\xi) & (\text{if } \xi < 0). \end{cases}$$

It is easily verified that f is continuous on $\mathbb{R}, \xi = 0$ is the unique solution of $f(\xi) = 0$, and, for all $\xi \neq 0$, f is twice continuously differentiable and f'(ξ) $\neq 0$. Further

$$[f'(\xi)]^{-1}f(\xi) = \theta(\alpha, 2)\xi \quad (\text{for all } \xi \neq 0).$$

Consequently (cf. (6.5.2)),

$$\xi \left[f'(\xi) \right]^{-1} f(\xi) = \xi^2 \theta(\alpha, 2) = \xi^2 \alpha \{ 1 + \theta(\alpha, 2)^2 \}$$

$$= \alpha \left[\xi^2 + \{ \left[f'(\xi) \right]^{-1} f(\xi) \}^2 \right] \quad (\text{for all } \xi \neq 0) \,.$$

If $\xi \neq 0$ then (cf. (6.5.8))

$$\xi - \left[f'(\xi)\right]^{-1}f(\xi) = \left[1 - \theta(\alpha, 2)\right]\xi = -\psi(\alpha)\xi \leq -\xi.$$

It is easily verified that $\xi \notin S(M,F)$.

However, since $f \notin F < \sigma, \alpha > (f \text{ is not differentiable at } \xi = 0)$, with this example we have *not* completed the proof of Theorem 6.5.1. With the next lemma, however, we can complete the proof of Theorem 6.5.1 by showing that an $f \in F < \sigma, \alpha >$ with the desired properties exists, which is not only twice but infinitely differentiable on D(f).

LEMMA 6.5.5. Let M be the Newton's method. Suppose $\alpha \in [-\frac{1}{2}, \frac{2}{5}]$. Then for any $\varepsilon > 0$ there exists an $F \in F < \sigma, \alpha >$ which is infinitely differentiable on D(F) for which r(M,F) $\leq \varepsilon$.

We shall prove this lemma, with the function f of Remark 6.5.1 in mind, in a more general setting in subsection 6.6.2 (cf. p.119).

6.6. The radius of convergence of $M_{\overline{\omega}}$ with respect to $F{<}\sigma, \alpha{>}$

Let Ω be defined in (6.3.5); for $\overline{\omega} \in \Omega$ the iterative method \underline{M}_{ω} is defined by Definition 6.3.1.

In this section we give a theorem concerning the radius of convergence of $M_{\overline{\omega}}$ with respect to $F<\sigma,\alpha>$, where $\sigma \in (0,\infty]$ and $\alpha \in [-\frac{1}{2},\frac{1}{2}]$. To that end let

$$\psi_1: (0,1] \times (0,\frac{1}{2}] \to \mathbb{R},$$

(6.6.1)

Here θ is defined in (6.5.1). Let $\omega \in (0,1]$. In view of (6.5.2-5) the following relations (6.6.2a,b,c) hold.

$$(6.6.2a) \qquad \psi_1(\omega, \cdot) \text{ is continuous and strictly antitone on } (0, \frac{1}{2}], \\ \lim_{\xi \neq 0} \psi_1(\omega, \xi) = \infty, \ \psi_1(\omega, \frac{1}{2}) = 1 - \omega. \\ (6.6.2b) \qquad \text{If } \omega = 1 \text{ then } \psi_1(\omega, \frac{2}{5}) = 1, \text{ if } \omega \in (0, 1) \text{ then } \psi_1(\omega, \xi) < 1$$

(for all
$$\xi \in [\frac{2}{5}, \frac{1}{2}]$$
)

(6.6.2c)
$$\psi_1(\omega,\xi) > \psi_1(2\xi,\xi) = \sqrt{1-4\xi^2} > 0$$
 (for all $\xi \in (0,\frac{1}{2}]$ with $2\xi \neq \omega$).

Consider the functions

$$\phi_1: \ \Omega \times (0, \frac{1}{2}] \rightarrow \mathbb{I}\mathbb{R},$$

(6.6.3)

$$\begin{split} \phi_1(\bar{\omega},\xi) &= \max\{1, \prod_{j=0}^n \psi_1(\omega_j,\xi) \quad (n = 0,1,\ldots,N-1)\} \\ &\quad (\xi \in (0,\frac{1}{2}] \text{ and } \bar{\omega} \in \Omega \text{ with } \bar{\omega} = (\omega_0,\omega_1,\ldots,\omega_N)) \end{split}$$

and

$$\phi_2: \Omega \times (0, \frac{1}{2}] \rightarrow \mathbb{R},$$

(6.6.4)

$$\begin{split} \phi_2(\overline{\omega},\xi) &= \prod_{j=0}^N \psi_1(\omega_j,\xi) \\ &\quad (\xi \in (0,\frac{1}{2}] \text{ and } \overline{\omega} \in \Omega \text{ with } \overline{\omega} = (\omega_0,\omega_1,\ldots,\omega_N)) \,. \end{split}$$

Let $\bar{\omega} \in \Omega$. Note that $\omega_{N} = 1$. In view of (6.6.2a,b) we have

(6.6.5)
$$\phi_1(\bar{\omega}, \cdot)$$
 is continuous and antitone on $(0, \frac{1}{2}], \phi_1(\bar{\omega}, \xi) = 1$
(for all $\xi \in [\hat{\alpha}, \frac{1}{2}]$), where $0 < \hat{\alpha} < \frac{2}{5}$,

and

(6.6.6) $\phi_2(\bar{\omega}, \cdot)$ is continuous and strictly antitone on $(0, \frac{1}{2}]$,

$$\lim_{\xi \neq 0} \phi_2(\bar{\omega},\xi) = \infty \text{ and } \phi_2(\bar{\omega},\frac{2}{5}) < 1.$$

The following theorem holds.

THEOREM 6.6.1. Let $\sigma \in (0,\infty]$, $\alpha \in [-\frac{1}{2},\frac{1}{2}]$ and $\overline{\omega} \in \Omega$. Then the equation $\phi_2(\overline{\omega},\xi) = 1$ has a unique solution $\xi = \alpha_{\overline{\omega}}$ in $(0,\frac{1}{2}]$ and $\alpha_{\overline{\omega}} \in (0,\frac{2}{5})$. We have

$$(6.6.7) \qquad r(M_{\overline{\omega}}; F < \sigma, \alpha >) = \begin{cases} \frac{\sigma}{\phi_1(\overline{\omega}, \alpha)} & (\text{if } \alpha_{\overline{\omega}} < \alpha \leq \frac{1}{2}), \\ \delta & (\text{if } \alpha = \alpha_{\overline{\omega}}), \\ 0 & (\text{if } -\frac{1}{2} \leq \alpha < \alpha_{\overline{\omega}}). \end{cases}$$

Here δ satisfies

$$(6.6.8) \qquad 0 \le \delta \le \frac{\sigma}{\phi_1(\bar{\omega}, \alpha_{\varpi})} \ .$$

The proof of Theorem 6.6.1 will be given in the next two subsections. It depends on the following principle. Let $\sigma \in (0,\infty]$, $\alpha \in (0,\frac{1}{2}]$, $\overline{\omega} \in \Omega$ and $\omega \in (0,1]$. For $F \in F < \sigma, \alpha >$ and $y \in B(x^*, \sigma)$ set

$$z = y - \omega \Gamma(y) F(y) .$$

It can be shown (see Lemma 6.6.5) that

$$\|_{z-x}^{*}\| \leq \psi_{1}(\omega, \alpha) \|_{y-x}^{*}\|.$$

From this relation it can be shown that

$$r(\underline{M}_{\overline{\omega}}; F < \sigma, \alpha >) \geq \begin{cases} \frac{\sigma}{\phi_1(\overline{\omega}, \alpha)} & (\text{if } \phi_2(\overline{\omega}, \alpha) < 1), \\ \\ 0 & (\text{otherwise}). \end{cases}$$

Using specially chosen iterative processes $[M_{\widetilde{\omega}},F]$ it can be shown that (6.6.7) holds.

We give some illustrations.





2. Graph of $r(M_{\overline{\omega}}; F < \sigma, \alpha >); \sigma \in (0, \infty]$ and $\overline{\omega} \in \Omega$ are fixed.



3. Graph of r(M; F<\sigma, \alpha>), where M is Newton's method; $\sigma~ \epsilon~(0,\infty]$ is fixed.



Fig. 6.6.3



4. Graph of $r(M_{\overline{\omega}}; F < \sigma, \alpha >); \alpha \in [-\frac{1}{2}, \frac{1}{2}]$ and $\overline{\omega} \in \Omega$ are fixed $(\alpha \neq \alpha_{\overline{\omega}})$.



Fig. 6.6.4c

We give some corollaries of Theorem 6.6.1. <u>COROLLARY 6.6.2</u>. Let $\sigma \in (0, \infty]$ and $\overline{\omega} \in \Omega$. Let M be Newton's method. Then (6.6.9) $r(M_{\overline{\Omega}}; F < \sigma, \alpha >) \ge r(M; F < \sigma, \alpha >)$ (for all $\alpha \in [-\frac{1}{2}, \frac{1}{2}]$).

More specifically,

(6.6.10) $r(M_{\overline{\omega}}; F < \sigma, \alpha >) = r(M; F < \sigma, \alpha >) = \sigma$ (for all $\alpha \in (\frac{2}{5}, \frac{1}{2}])$ and there exist constants $\alpha_{\overline{\omega}}$ and $\hat{\alpha}_{\overline{\omega}}$ with $0 < \alpha_{\overline{\omega}} \le \hat{\alpha}_{\overline{\omega}} < \frac{2}{5}$ such that (6.6.11a) $0 < r(M_{\overline{\omega}}; F < \sigma, \alpha >) \le \sigma$ and $r(M; F < \sigma, \alpha >) = 0$ (for all $\alpha \in (\alpha_{\overline{\omega}}, \frac{2}{5}]$)

and

(6.6.11b)
$$r(M_{\overline{\alpha}}; F < \sigma, \alpha >) = \sigma$$
 (for all $\alpha \in (\alpha_{\overline{\alpha}}, \frac{2}{5}])$.

<u>PROOF</u>. According to (6.6.5) and (6.6.6) it follows that numbers $\alpha_{\overline{\omega}}, \widetilde{\alpha} \in (0, \frac{2}{5})$ exist such that $\phi_1(\overline{\omega}, \alpha) = 1$ (for all $\alpha \in [\widetilde{\alpha}, \frac{1}{2}]$), $\phi_2(\overline{\omega}, \alpha) < 1$ (for all $\alpha \in [\alpha_{\overline{\omega}}, \frac{1}{2}]$) and $\phi_2(\overline{\omega}, \alpha_{\overline{\omega}}) = 1$. Let $\widehat{\alpha}_{\overline{\omega}} = \max\{\widetilde{\alpha}, \alpha_{\overline{\omega}}\}$. From Theorem 6.6.1 and Theorem 6.5.1 it follows that the statement is true.

Let $\sigma \in (0,\infty]$, $\alpha \in [-\frac{1}{2},\frac{1}{2}]$ and $\overline{\omega} \in \Omega$. Corollary 6.6.2 shows that $M_{\overline{\omega}}$ has a radius of convergence with respect to $F < \sigma, \alpha >$ that is not smaller and, for certain values of α is even greater, than the radius of convergence of Newton's method. See also Remark 6.3.1.

Corollary 6.6.2 can also be interpreted as follows. Let $F \in F_1$. Then $F \in F < s(F; \alpha), \alpha > (\text{for all } \alpha \in [-\frac{1}{2}, \frac{1}{2}))$ (cf. (6.4.20), (6.4.22)). Consequently, $r(M,F) \ge \rho_0 = \sup\{s(F; \alpha) \mid \alpha \in (\frac{2}{5}, \frac{1}{2}]\}$ and $r(M_{\overline{\omega}}, F) \ge \rho_{\overline{\omega}} = \sup\{s(F; \alpha) \mid \alpha \in (\widehat{\alpha}_{\overline{\omega}}, \frac{1}{2}]\}$. In view of (6.4.21) it follows that $\rho_{\overline{\omega}} \ge \rho_0$. Corollary 6.6.2 guarantees that [M,F] generates a sequence $\{x_k\}$ that converges to x^* whenever the starting point $x_0 \in B(x^*, \rho_0)$. Furthermore, $[M_{\overline{\omega}}, F]$ generates a sequence $\{x_k\}$ that converges to x^* whenever $x_0 \in B(x^*, \rho_{\overline{\omega}}) \ge B(x^*, \rho_0)$.

Let $N_0 \in IN$ be given. Let $\xi_{N_0} \in (0, \frac{2}{5})$ satisfy

(6.6.12)
$$(\sqrt{1-4\xi_{N_0}^2})^{N_0}(\theta(\xi_{N_0}, 2)-1) = 1.$$

 $\xi_{N_{
m O}}$ is unique (cf. (6.5.4)). Define

(6.6.13)
$$\overline{\pi}[N_0] = (\pi_0[N_0], \pi_1[N_0], \dots, \pi_{N_0}[N_0])$$
 where
 $\pi_j[N_0] = 2\xi_{N_0}$ $(j = 0, 1, \dots, N_0^{-1})$ and $\pi_{N_0}[N_0] = 1$.

Hence, $\overline{\pi}[N_0] \in \Omega$ and the following result is a consequence of Theorem 6.6.1. <u>COROLLARY 6.6.3</u>. Let $\sigma \in (0,\infty]$. Let $\overline{\omega} \in \Omega$ with $\overline{\omega} = (\omega_0, \omega_1, \dots, \omega_N)$, $N \leq N_0$ and $\overline{\omega} \neq \overline{\pi}[N_0]$. Then

(6.6.14)
$$r(M_{\overline{\pi}[N_0]}; F < \sigma \alpha >) \ge r(M_{\overline{\omega}}; F < \sigma, \alpha >)$$
 (for all $\alpha \in [-\frac{1}{2}, \frac{1}{2}]$).

More specifically,

(6.6.15) $r(M_{\overline{\pi}[N_0]}; F < \sigma, \alpha >) = \sigma$ (for all $\alpha \in (\xi_{N_0}, \frac{1}{2}])$ and there exists a constant $\mu_{\overline{\omega}}$ with $\xi_{N_0} < \mu_{\overline{\omega}} < \frac{1}{2}$ such that (6.6.16) $r(M_{\overline{\omega}}; F < \sigma, \alpha >) = 0$ (for all $\alpha \in [-\frac{1}{2}, \mu_{\overline{\omega}})$).

<u>**PROOF.**</u> From (6.6.2c) it follows that for $j = 0, 1, \dots, N_0^{-1}$ we have

$$\psi_1(\pi_j[N_0],\xi_{N_0}) < \psi_1(\omega,\xi_{N_0}) \quad (\text{for all } \omega \in (0,1] \text{ with } \omega \neq \pi_j[N_0])$$

and

$$\psi_1(\pi_j[N_0],\xi_{N_0}) < 1$$

Consequently,

$$1 = \phi_2(\overline{\pi}[N_0], \xi_{N_0}) < \phi_2(\overline{\omega}, \xi_{N_0})$$

and

$$1 = \phi_1(\overline{\pi}[N_0], \xi_{N_0}) \leq \phi_1(\overline{\omega}, \xi_{N_0}).$$

Hence, from Theorem 6.6.1, (6.6.5) and (6.6.6) it follows that the statements (6.6.14 - 16) are true. \Box

Corollary 6.6.3 may be interpreted as follows. Of all iterative methods $M_{\overline{\omega}}$ for which $\overline{\omega} \in \Omega$ with $\overline{\omega} = (\omega_0, \omega_1, \dots, \omega_N)$ and $N \leq N_0$, the iterative method $M_{\overline{\pi}[N_0]}$ is optimal in the following sense. Let $\sigma \in (0, \infty]$ and $\alpha \in [-\frac{1}{2}, \frac{1}{2}]$. Any iterative method $M_{\overline{\omega}}$ for which $\overline{\omega} = (\omega_0, \omega_1, \dots, \omega_N)$ with $N \leq N_0$ and $\overline{\omega} \neq \overline{\pi}[N_0]$, possesses a radius of convergence with respect to $F < \sigma, \alpha >$ that is never greater and, for certain values of α is even smaller, than the radius of convergence of $M_{\overline{\pi}[N_0]}$.

We conclude this section by presenting another consequence of Theorem 6.6.1.

COROLLARY 6.6.4. For any $\alpha \in (0, \frac{1}{2}]$ an $\overline{\omega} \in \Omega$ exists for which

$$r(M_{\omega}; F < \sigma, \alpha >) = \sigma$$
 (for all $\sigma \in (0, \infty]$).

<u>PROOF</u>. It is easily verified that $\xi_{N_0} \neq 0$ (if $N_0 \neq \infty$), (cf. (6.6.12)). Hence, Corollary 6.6.4 follows from (6.6.15).

6.6.1. Proof of Theorem 6.6.1

Throughout this subsection, σ and α denote the constants that appear in Theorem 6.6.1 and $\overline{\omega}$ the element of Ω . We start with a lemma which was alluded to in the previous section (page 107).

LEMMA 6.6.5. Let $\omega \in (0,1]$. Suppose $\alpha \in (0,\frac{1}{2}]$. Let $F \in F < \sigma, \alpha >$. Then $B(x^*,\sigma) \subset D(F)$. Let $y \in B(x^*,\sigma)$. Then F'(y) is invertible. Set $z = y - \omega \Gamma(y)F(y)$. Then

(6.6.17) $||_{z-x}^{*}|| \leq \psi_{1}(\omega, \alpha) ||_{y-x}^{*}||.$

Here ψ_1 is defined in (6.6.1).

PROOF. Notice that

$$(y-x^{*},\Gamma(y)F(y)) \geq \alpha \{ \|y-x^{*}\|^{2} + \|\Gamma(y)F(y)\|^{2} \}.$$

Consequently,

$$\| z_{-\mathbf{x}}^{*} \|^{2} = \| y_{-\mathbf{x}}^{*} \|^{2} + \omega^{2} \| \Gamma(y) F(y) \|^{2} - 2\omega(y_{-\mathbf{x}}^{*}, \Gamma(y) F(y))$$

$$\leq (1 - 2\omega\alpha) \| y_{-\mathbf{x}}^{*} \|^{2} + (\omega^{2} - 2\omega\alpha) \| \Gamma(y) F(y) \|^{2}.$$

Using Lemma 6.5.2 it follows that

$$\|_{\mathbf{Z}-\mathbf{x}}^{*}\|^{2} \leq \begin{cases} \{1-2\omega\alpha+(\omega^{2}-2\omega\alpha)\left[\theta\left(\alpha,1\right)\right]^{2}\}\|_{\mathbf{Y}-\mathbf{x}}^{*}\|^{2} & (\text{if } \omega \leq 2\alpha), \\ \\ \{1-2\omega\alpha+(\omega^{2}-2\omega\alpha)\left[\theta\left(\alpha,2\right)\right]^{2}\}\|_{\mathbf{Y}-\mathbf{x}}^{*}\|^{2} & (\text{if } \omega > 2\alpha). \end{cases}$$

From (6.5.2) it thus follows that

$$\|\mathbf{z}-\mathbf{x}^{*}\|^{2} \leq \begin{cases} \left\{ \boldsymbol{\omega}^{2} \left[\boldsymbol{\theta}\left(\boldsymbol{\alpha},1\right) \right]^{2} - 2\boldsymbol{\omega}\boldsymbol{\theta}\left(\boldsymbol{\alpha},1\right) + 1 \right\} \|\mathbf{y}-\mathbf{x}^{*}\|^{2} & (\text{if } \boldsymbol{\omega} \leq 2\boldsymbol{\alpha}), \\ \\ \left\{ \boldsymbol{\omega}^{2} \left[\boldsymbol{\theta}\left(\boldsymbol{\alpha},2\right) \right]^{2} - 2\boldsymbol{\omega}\boldsymbol{\theta}\left(\boldsymbol{\alpha},2\right) + 1 \right\} \|\mathbf{y}-\mathbf{x}^{*}\|^{2} & (\text{if } \boldsymbol{\omega} > 2\boldsymbol{\alpha}). \end{cases}$$

Thus (cf. (6.6.1) and (6.6.2c)) relation (6.6.17) is true.

Let ϕ_1 and ϕ_2 be defined in (6.6.3) and (6.6.4) respectively. A consequence of the previous lemma is

LEMMA 6.6.6. Suppose $\alpha \in (0, \frac{1}{2}]$. Let $F \in F < \sigma, \alpha > and set G = M_{\overline{\omega}}(F)$. Then for all $x \in B(x^*, \sigma/\phi_1(\overline{\omega}, \alpha))$ we have $x \in D(G)$ and

(6.6.18)
$$\|G(\mathbf{x}) - \mathbf{x}^*\| \le \phi_2(\bar{\omega}, \alpha) \|\mathbf{x} - \mathbf{x}^*\|$$

<u>PROOF</u>. Let $x \in B(x^*, \sigma/\phi_1(\overline{\omega}, \alpha))$. Using Lemma 6.6.5 we have, with $y_0 = x$,

$$\|\boldsymbol{y}_{n+1} - \boldsymbol{x}^{\star}\| \leq \psi_1(\boldsymbol{\omega}_n, \boldsymbol{\alpha}) \|\boldsymbol{y}_n - \boldsymbol{x}^{\star}\| \leq \prod_{j=0}^n \psi_1(\boldsymbol{\omega}_j, \boldsymbol{\alpha}) \|\boldsymbol{y}_0 - \boldsymbol{x}^{\star}\| < \sigma$$

where

$$y_{n+1} = y_n - \omega_n \Gamma(y_n) F(y_n)$$
 (n = 0,1,...,N-1).

Since $G(x) = y_N - \Gamma(y_N)F(y_N)$ it follows that

$$\|G(\mathbf{x}) - \mathbf{x}^{*}\| \leq \psi_{1}(1, \alpha) \|_{\mathcal{Y}_{N}} - \mathbf{x}^{*}\| \leq \prod_{j=0}^{N} \psi_{1}(\omega_{j}, \alpha) \|_{\mathbf{x} - \mathbf{x}^{*}}\|.$$

This proves the statement.

We are now in a position to give a lower bound on $r(M_{\overline{\omega}};\;F{<}\sigma,\alpha{>})$.

LEMMA 6.6.7.

(6.6.19)
$$r(\underline{M}_{\overline{\omega}}; F < \sigma, \alpha >) \ge \begin{cases} \sigma & \text{(if } \alpha \in (0, \frac{1}{2}] \text{ and } \phi_2(\overline{\omega}, \alpha) < 1), \\ 0 & \text{(otherwise).} \end{cases}$$

PROOF. Relation (6.6.19) is a direct consequence of Lemma 6.6.6.

The next subsection will be devoted to the proof of the following lemma, which may be conceived as a generalization of Lemma 6.5.5.

LEMMA 6.6.8. Suppose $\alpha \in (0, \frac{1}{2}]$. Then the following propositions hold. (i) If $\phi_1(\bar{\omega}, \alpha) > 1$ then, for any $\tau \in \mathbb{R}$ for which $\phi_1(\bar{\omega}, \alpha)\tau > \sigma$, an $F \in \bar{F} < \sigma, \alpha >$ which is infinitely differentiable on D(F) exists for which

(6.6.20)
$$r(M_{-},F) \leq \tau$$

(ii) If $\phi_2(\bar{\omega}, \alpha) > 1$ then, for any $\varepsilon > 0$, an F ϵ F< σ, α > which is infinitely differentiable on D(F) exists for which

(6.6.21)
$$r(M_{\overline{\omega}},F) \leq \varepsilon$$
.

Notice that

1. $r(M_{\overline{\omega}}; F < \sigma, \alpha >) \le \sigma$ (cf. Remark 6.2.1). Hence (6.6.7) follows from Lemma 6.6.7, (6.6.6) and Lemma 6.6.8.

2. $F < \sigma, \alpha_1 > \neg F < \sigma, \alpha_2 >$ (if $-\frac{1}{2} \le \alpha_1 \le \alpha_2 \le \frac{1}{2}$). Hence $r(M_{\overline{\omega}}; F < \sigma, \alpha >)$ is isotone on $\left[-\frac{1}{2},\frac{1}{2}\right]$ so that relation (6.6.8) holds.

Therefore, the statement of Theorem 6.6.1 is true.

6.6.2. Proof of the Lemmata 6.5.5 and 6.6.8

PART A.

We assume in this part of the proof that E = IR.

LEMMA 6.6.9. Let the integer $n \ge 0$ and let $\sigma \in (0, \infty]$. Let $\alpha_i \in (0, \frac{1}{2}]$, $\tau_j \in (0,\sigma)$ and $k_j \in \{1,2\}$ (j = 0,1,...,n). Suppose $\tau_i \neq \tau_j$ if $i \neq j$ (i,j = 0,1,...,n). Then for any $\hat{\alpha}$, with $0 < \hat{\alpha} \le \min\{\alpha_j \mid 0 \le j \le n\}$, an f \in F< \sigma, \alpha > \cap C[∞] (- \sigma, \sigma) (see (6.2.4)) exists such that

(i) $x^* = 0$ is the unique solution of the equation f(x) = 0,

(6.6.22) (ii) $f'(\xi) \neq 0$ (for all $\xi \in (-\sigma, \sigma)$),

(iii)
$$[f'(\xi)]^{-1}f(\xi) = \theta(\alpha_{j},k_{j})\xi$$
 (if $|\xi| = \tau_{j}$ (j = 0,1,...,n), or
 $|\xi| \in [\tau_{j},\sigma)$ and $\tau_{j} \ge \tau_{i}$ (i = 0,1,...,n)).

<u>PROOF</u>. We assume that $0 < \tau_0 < \tau_1 < \ldots < \tau_n$, this is no restriction. Let $\hat{\alpha}$ satisfy $0 < \hat{\alpha} \le \min\{\alpha_j \mid 0 \le j \le n\}$. Set $\tau_{n+1} = \sigma$ and let $\tau_{-1} \in (0, \tau_0)$. Let $\varepsilon > 0$ with $\varepsilon < \frac{1}{3} \min\{\tau_j - \tau_{j-1} \mid 0 \le j \le n+1\}$. Set $c_j = \theta(\alpha_j, k_j)$ $(j = 0, 1, \dots, j)$ n), $c_{-1} = 1$. Let $\overline{d}_j = (c_{j-1}, c_j, \tau_j - 2\varepsilon, \tau_j - \varepsilon)$ $(j = 0, 1, \dots, n)$ and let $\widetilde{h}_j = h_{\overline{d}_j}$ (cf. (6.2.6)) (j = 0, 1, ..., n). Let

$$\hat{h}: [0,\sigma) \rightarrow (0,\infty),$$

(6.6.23)



Define

(6.6.24)

$$f(\xi) = \begin{cases} \exp[\int_{\tau_{-1}}^{\xi} \frac{1}{\hat{h}(\lambda)\lambda} d\lambda] & (\text{if } \xi \in (0,\sigma)), \\ 0 & (\text{if } \xi = 0), \\ -f(-\xi) & (\text{if } \xi \in (-\sigma,0)) \end{cases}$$

1. We show that $f \in C^{\infty}(-\sigma, \sigma)$.

a. f is continuous on $(0,\sigma)$ and

f: $(-\sigma,\sigma) \rightarrow \mathbb{R}$,

(6.6.25)
$$f'(\xi) = \frac{1}{\hat{h}(\xi)\xi} f(\xi) > 0$$
 (for all $\xi \in (0,\sigma)$).

In general, for $k \ge 2$

$$f^{(k)}(\xi) = \frac{f(\xi)}{\left[\hat{h}(\xi)\xi\right]^k} p_k(\xi) \quad (\text{for all } \xi \in (0,\sigma))$$

where p_k is a differentiable function that is composed of the functions $\hat{h},\hat{h}',\ldots,\hat{h}^{(k-1)}$.

b. If $\xi \in (0, \tau_{-1})$, then

$$f(\xi) = \exp\left[\int_{\tau_{-1}}^{\xi} \frac{1}{\lambda} d\lambda\right] = \frac{\xi}{\tau_{-1}}$$

Hence f is continuous on $(-\tau_{-1}, \tau_{-1})$ and $f^{(k)}(\xi)$ exists (for all $k \in \mathbb{N}$ and all $\xi \in (-\tau_{-1}, \tau_{-1})$).

c. For $\xi \in (-\sigma, 0)$ we have $f(\xi) = -\phi(-\xi)$ where ϕ denotes the restriction of f to $(0,\sigma)$. Since ϕ is infinitely differentiable on $(0,\sigma)$, the function f is infinitely differentiable on $(-\sigma, 0)$ with

(6.6.26)
$$f^{(k)}(\xi) = (-1)^{k+1} f^{(k)}(-\xi)$$
 (for all $k \in \mathbb{N}$ and all $\xi \in (-\sigma, 0)$).

Hence $f \in C^{\infty}(-\sigma,\sigma)$.

2. We prove that $f \in F < \sigma, \hat{\alpha} > 0$. Obviously (cf. (6.6.25), (6.6.26)), $f'(\xi) > 0$ (for all $\xi \in (-\sigma, \sigma)$). Hence $\xi = 0$ is the unique solution of $f(\xi) = 0$. Notice that, since $\hat{\alpha} > 0$ and $\hat{\alpha} \le \min\{\alpha_j \mid 0 \le j \le n\}$ it follows that

$$\frac{1-\sqrt{1-4\hat{\alpha}^2}}{2\hat{\alpha}} = \theta(\hat{\alpha},1) \le \min\{\theta(\alpha_j,1) \mid 0 \le j \le n\} \le \hat{h}(\xi)$$
$$\le \max\{\theta(\alpha_j,2) \mid 0 \le j \le n\} \le \theta(\hat{\alpha},2) = \frac{1+\sqrt{1-4\hat{\alpha}^2}}{2\hat{\alpha}} \text{ (for all } \xi \in [0,\sigma)\text{)}$$

(see (6.5.4) and (6.5.5)). Hence

$$\hat{h}(\xi) \geq \hat{\alpha}\{1 + [\hat{h}(\xi)]^2\} \qquad (\text{for all } \xi \in [0,\sigma)).$$

Thus

$$\hat{h}(\xi)\xi^2 \ge \hat{\alpha}\{\xi^2 + [\hat{h}(\xi)\xi]^2\} \qquad (\text{for all } \xi \in [0,\sigma)).$$

Therefore (cf. (6.6.25)) it follows that

(6.6.27)
$$\xi \frac{f(\xi)}{f'(\xi)} \ge \hat{\alpha} \{\xi^2 + \left[\frac{f(\xi)}{f'(\xi)}\right]^2\} \quad (\text{for all } \xi \in [0,\sigma)).$$

For $\xi \in (-\sigma, 0)$ we have (cf. (6.6.26))

(6.6.28)
$$\frac{f(\xi)}{f'(\xi)} = -\frac{f(-\xi)}{f'(-\xi)}$$
.

Hence (6.6.27) holds for $\xi \in (-\sigma, 0)$ as well. Thus $f \in F < \sigma, \hat{\alpha} > .$

3. Relation (iii) follows from (6.6.23), (6.6.25) and (6.6.28).

(6.6.29a)
$$|y_{j+1}| = \tau_{j+1}$$
 (j = 0,1,...,n)

Here

$$(6.6.29b) \quad y_{j+1} = y_j - \omega_j [f'(y_j)]^{-1} f(y_j) \quad (j = 0, 1, ..., n)$$

and

(6.6.29c) $y_0 = \xi$.

<u>PROOF</u>. Let $\xi \in (-\sigma, \sigma)$ satisfy $|\xi| = \tau_0$. Assume ℓ is an integer with $0 \le \ell \le n$. Suppose (6.6.29) holds for all j with $0 \le j \le \ell-1$. Thus $y_\ell \in (-\sigma, \sigma)$, and

$$\mathbf{y}_{\ell+1} = \mathbf{y}_{\ell} - \boldsymbol{\omega}_{\ell} [\mathbf{f}'(\mathbf{y}_{\ell})]^{-1} \mathbf{f}(\mathbf{y}_{\ell}) = (1 - \boldsymbol{\omega}_{\ell} \boldsymbol{\theta}(\boldsymbol{\alpha}_{\ell}, \mathbf{k}_{\ell})) \mathbf{y}_{\ell}.$$

Hence $|y_{\ell+1}| = \tau_{\ell+1}$ (cf. (6.6.1), (6.6.2c)). This proves the lemma.

The following lemma shows that the Lemmata 6.5.5, 6.6.8 hold if E = IR.

LEMMA 6.6.11. Let $\alpha \in (0, \frac{1}{2}]$ and $\overline{\omega} \in \Omega$.

- (i) Suppose $\phi_1(\overline{\omega}, \alpha) > 1$. Let $\sigma \in (0, \infty)$. Then for all $\tau \in \mathbb{R}$ with $\phi_1(\overline{\omega}, \alpha)\tau > \sigma$ and $\epsilon \in F < \sigma, \alpha > \cap \mathbb{C}^{\infty}(-\sigma, \sigma)$ and a number $\xi_0 \in D(f)$ with $|\xi_0| \le \tau$ exist such that f(0) = 0, $f'(\xi) \ne 0$ (for all $\xi \in (-\sigma, \sigma)$) and $\xi_0 \notin D(M_{\overline{\omega}}, f)$.
- (ii) Suppose $\phi_2(\overline{\omega}, \alpha) > 1$. Then for any $\varepsilon > 0$ an $f \in F_{\infty}, \alpha > \cap C^{\infty}(\mathbb{R})$ and a number $\xi_0 \in D(f)$ with $|\xi_0| \le \varepsilon$ exist such that f(0) = 0, $f'(\xi) \ne 0$ (for all $\xi \in \mathbb{R}$), $\phi(\xi_0) \in D(\phi)$, $\phi[\phi(\xi_0)] = \xi_0$ and $\xi_0 \notin S(M_{\overline{\omega}}, f)$. Here $\phi = M_{\overline{\omega}}(f)$.
- (iii) Suppose $\alpha = \frac{2}{5}$. Then for any $\varepsilon > 0$ an $f \in F<\infty, \alpha > \cap C^{\infty}(\mathbb{R})$ and a number $\xi_0 \in D(f)$ with $|\xi_0| = \varepsilon$ exist such that f(0) = 0, $f'(\xi) \neq 0$ (for all $\xi \in \mathbb{R}$), $\phi(\xi_0) \in D(\phi)$, $\phi[\phi(\xi_0)] = \xi_0$ and $\xi_0 \notin S(M, f)$. Here M is Newton's method and $\phi = M(f)$.

 $\begin{array}{l} \underline{PROOF}. \ (i) \ \text{Suppose } \phi_1(\overline{\omega}, \alpha) > 1. \ \text{Consequently } \alpha < \frac{1}{2} \ \text{and an } n < N \ \text{exists such} \\ \text{that } \Pi_{j=0}^n \ \psi_1(\omega_j, \alpha) > 1. \ \text{Let } \tau \ \epsilon \ (0, \sigma) \ \text{such that } \Pi_{j=0}^n \ \psi_1(\omega_j, \alpha) > \frac{\sigma}{\tau}. \ \text{We assume} \\ \text{that } \Pi_{j=0}^i \ \psi_1(\omega_j, \alpha) \le \frac{\sigma}{\tau} \ (\text{for all } i < n). \end{array}$

 $\begin{array}{l} \int_{0}^{j=0} 1 \int_{0}^{1} t_{j} dt_{j} dt_{j}$

(ii) Suppose $\phi_2(\bar{\omega}, \alpha) > 1$. Hence $\alpha < \frac{1}{2}$. Let $\varepsilon > 0$. As in the first part of the proof we can construct numbers α_j (j = 0,1,...,N) and τ_j

 $(j = 0, 1, \dots, N+1), \text{ where } \tau_0 = \varepsilon \text{ and } \tau_{N+1} > \tau_0, \text{ such that } \alpha \leq \min\{\alpha_j \mid 0 \leq j \leq N\}, \\ \text{and } \tau_{j+1} = \psi(\alpha_j, \alpha_j) \tau_j \text{ and } \tau_i \neq \tau_j \text{ for } i < j (j = 0, 1, \dots, N). \text{ From Lemma} \\ \text{6.6.10 it follows that an } f \in F < \infty, \alpha > \cap C^{\infty}(\mathbb{R}) \text{ exists such that, for all} \\ \xi \in \mathbb{R} \text{ with } |\xi| = \varepsilon, \text{ we have } \phi(\xi) = \tau_{N+1} > \tau_0 = |\xi|. \text{ Here } \phi = M_{\overline{\alpha}}(f). \text{ Since} \\ \phi \text{ is continuous on } \mathbb{R} \text{ and } [M_{\overline{\alpha}}, f] \text{ is quadratically convergent (cf. p.87),} \\ \text{from Lemma 6.2.6 it follows that a number } \xi_0 \in [-\varepsilon, \varepsilon] \text{ with } \xi_0 \neq 0 \text{ exists} \\ \text{such that } \phi[\phi(\xi_0)] = \xi_0. \text{ This proves (ii).} \\ (\text{iii) Suppose } \alpha = \frac{2}{5}. \text{ Let } \varepsilon > 0 \text{ and let } \tau_0 = \varepsilon. \text{ From Lemma 6.6.9 and (6.5.4)} \\ \end{cases}$

it follows that an $f \in F_{\infty,\alpha} \cap \mathbb{C}^{\infty}(\mathbb{R})$ exists such that $[f'(\xi)]^{-1}f(\xi) = 2\xi$ (for all $\xi \in \mathbb{R}$ with $|\xi| = \tau_0$). This proves (iii).

PART B.

We are now in a position to prove the Lemmata 6.5.5 and 6.6.8 where E is an arbitrary Hilbert space, as will be clear from the next lemma.

<u>LEMMA 6.6.12</u>. Let $\sigma \in (0,\infty]$ and $\alpha \in (0,\frac{1}{2}]$. Let $f \in C^{\infty}(-\sigma,\sigma)$ and suppose that $f'(\xi) \neq 0$ (for all $\xi \in (-\sigma,\sigma)$) and f(0) = 0. Moreover, let

(6.6.30)
$$\xi \frac{f(\xi)}{f'(\xi)} \ge \alpha \{ \xi^2 + \left[\frac{f(\xi)}{f'(\xi)} \right]^2 \}$$
 (for all $\xi \in (-\sigma, \sigma)$).

Let F be the E-extension of f (cf. (6.2.14)). Then F ϵ F<0, α > and x^{*} = 0.

PROOF. In view of Theorem 6.2.12 and (6.4.9) it is sufficient to prove that

(6.6.31)
$$(\mathbf{x},\Gamma(\mathbf{x})\mathbf{F}(\mathbf{x})) \ge \alpha \{ \|\mathbf{x}\|^2 + \|\Gamma(\mathbf{x})\mathbf{F}(\mathbf{x})\|^2 \}$$
 (for all $\mathbf{x} \in B(0,\sigma)$).

Let $x \in B(0,\sigma)$. Suppose that $\mathcal{B}_{x} = \{u | u = u_{n} \text{ with } n \in \mathbb{N}\}$ and $\xi_{n} = (x,u_{n}) (n = 1,2, \dots)$. Hence $|\xi_{n}| < \sigma$ $(n = 1,2,\dots)$. From Theorem 6.2.12(iii) it follows that

$$(\mathbf{x}, \Gamma(\mathbf{x}) F(\mathbf{x})) = \sum_{n=1}^{\infty} \xi_n \frac{f(\xi_n)}{f'(\xi_n)} .$$

Since

$$\|\mathbf{x}\|^{2} + \|\Gamma(\mathbf{x})F(\mathbf{x})\|^{2} = \sum_{n=1}^{\infty} \{\xi_{n}^{2} + [\frac{f(\xi_{n})}{f'(\xi_{n})}]^{2}\},\$$

relation (6.6.30) shows that (6.6.31) holds.

PROOF OF THE LEMMATA 6.5.5, 6.6.8.

1. Suppose $\phi_1(\overline{\omega}, \alpha) > 1$. Let $\tau \in \mathbb{R}$ satisfy $\phi_1(\overline{\omega}, \alpha)\tau > \sigma$. Let $f \in C^{\infty}(-\sigma, \sigma)$ and ξ_0 satisfy statement (i) of Lemma 6.6.11. Let F be the E-extension of f. Thus $F \in F < \sigma, \alpha >$ and there exists an element u of E with $\|u\| = 1$ such that

$$\Gamma(\lambda u) F(\lambda u) = \frac{f(\lambda)}{f'(\lambda)} u \quad (\text{for all } \lambda \in (-\sigma, \sigma))$$

Set $x_0 = \xi_0 u$. Then $x_0 \notin D(M_{\overline{w}}, F)$. Finally, from Theorem 6.2.12 it follows that F is infinitely differentiable on D(F). This proves statement (i) of Lemma 6.6.8.

2. By a similar argument to the above, (using Lemma 6.6.11(ii)) one can show that statement (ii) of Lemma 6.6.8 is true, and (using Lemma 6.6.11 (iii)) prove Lemma 6.5.5.

·

CHAPTER 7

NUMERICAL EXPERIMENTS

In this chapter we present some numerical results. The test examples will all be finite dimensional problems. Examples of the application of the imbedding method to problems in infinite dimensional (Banach) spaces can be found, amongst others, in the following three references: [WACKER, 1972] (where the discrete imbedding method is used), [BOSARGE, 1971] (where Davidenko's method is used), [KLEINMICHEL, 1968] (where iterative imbedding is used), (cf. section 1.1).

We present some numerical results with iterative methods of the type investigated in Chapters 5 and 6.

All computations have been carried out on a CDC Cyber 73/173-28 computer (accuracy: 48 binary digits in the mantissa).

7.1. ITERATIVE METHODS TO BE TESTED

We have divided the iterative methods to be tested into three classes. Let $A \in A$ (cf. (2.6.3)) where $[A(F)](y,x) \equiv F(y) - F(x)$ (for all $F \in F_1$ (cf. (2.6.1)). Let $F \in F_1$.

<u>CLASS 1</u>. Let $N \ge 0$ and $\mathcal{H} = \{h_0, h_1, \dots, h_N\}$ with $h_i = \frac{1}{N+1}$ (i = 0,1,...,N). Let g = 0. Let R_{θ} (for $\theta \in IR$) be the one-stage Runge-Kutta method with operator coefficient for which

 $\rho_{2,1}(z) \equiv \left[1 - (1 - \theta)z\right]^{-1}$

(see also subsection 3.3.2). We notice that, when solving a linear differential equation with constant coefficients, the method R_{θ} with $\theta = \frac{1}{2}$ is equivalent to the *Trapezoidal rule*, and the method R_{θ} with $\theta = 0$ is equivalent to the *Backward Euler method*. The Trapezoidal rule is more accurate than the Backward Euler method. However the latter method has better stability properties than the former one (cf. [LAMBERT, 1973; pp. 235-236]). If $\theta = 1$ then R_A reduces to Euler's method.

Set $M_{N,\theta} = IM(A,g,R_{\theta},H)$. Then $G = M_{N,\theta}(F)$ satisfies (cf. section 4.2)

G: D(G) \rightarrow E,

(7.1.1a)

 $G(x) = \eta_{N+1}(x)$ (x $\in D(G)$).

In (7.1.1a) $\eta_j(x)$ (j = 0,1,...,N+1) is defined by

 $\eta_0(\mathbf{x}) = \mathbf{x}$ (7.1.1b) and

$$\eta_{j+1}(x) = \eta_{j}(x) - \frac{1}{N+1} \left[I - \frac{(1-\theta)}{N+1} \Gamma(\eta_{j}(x))F''(\eta_{j}(x))\Gamma(\eta_{j}(x))F(x) \right]^{-1} \Gamma(\eta_{j}(x))F(x) \quad (j = 0, 1, ..., N).$$

We notice that for $\theta = \frac{1}{2}$ and N = 0 the method $M_{N,\theta}$ is known in the literature as the method of tangent hyperbolas (cf. [ORTEGA & RHEINBOLDT, 1970; p. 188] for a bibliography on this method). Furthermore for $\theta = 0$ and N = 0 the method $M_{N,\theta}$ has been investigated in [DI LENA & TRIGIANTE, 1976]. In that paper it was supposed that E = IR. In the computations that were performed on some problems in IR, the method exhibited better convergence behaviour than the convergence behaviour of Newton's method, especially when the starting points were not close to the desired solution.

From Theorem 5.2.8 it follows that the iterative processes $[M_{N,\theta},F]$ (N \geq 0, 0 ϵ IR) are all quadratically convergent.

In higher dimensional vector spaces computation of F"(z) requires in general an exorbitant amount of work. Consequently, for $\theta \neq 1$, the iterative methods $M_{N,\theta}$ are rather cumbersome from the computational point of view. We shall therefore modify $M_{N,\theta}$ in such a way that F"($n_j(x)$) need not be computed. Let $y, x \in D(F)$. Suppose F'(y) is invertible. For any $\varepsilon > 0$ there exists a number $\rho > 0$ such that for all τ with $0 < |\tau| < \rho$

$$\frac{1}{\tau} \| \mathbf{F}'(\mathbf{y} + \tau \Gamma(\mathbf{y}) \mathbf{F}(\mathbf{x})) - \mathbf{F}'(\mathbf{y}) - \tau \mathbf{F}''(\mathbf{y}) \Gamma(\mathbf{y}) \mathbf{F}(\mathbf{x}) \| < \varepsilon.$$

When τ is small, the operator $\frac{1}{\tau} \{F'(y + \tau \Gamma(y)F(x)) - F'(y)\}$ is therefore approximately equal to $F''(y)\Gamma(y)F(x)$. Thus we can approximate the iteration function G, defined in (7.1.1) by an iteration function \widetilde{G} ,

$$\widetilde{G}: D(\widetilde{G}) \rightarrow E,$$

(7.1.2a)

 $\widetilde{G}(\mathbf{x}) = \widetilde{\eta}_{\mathbf{N+1}}(\mathbf{x}) \qquad (\mathbf{x} \in D(\widetilde{G})).$

In (7.1.2a) $\widetilde{\eta}_{j}(x)$ (j = 0,1,...,N+1) is defined by

(7.1.2b)
$$\widetilde{\eta}_{0}(\mathbf{x}) = \mathbf{x}$$

$$\widetilde{\eta}_{j+1}(\mathbf{x}) = \widetilde{\eta}_{j}(\mathbf{x}) - \frac{1}{N+1} \left[\mathbf{F}'(\widetilde{\eta}_{j}(\mathbf{x})) - \frac{(1-\theta)}{(N+1)\tau} \left[\mathbf{F}'(\widetilde{\eta}_{j}(\mathbf{x}) + \tau \mathbf{F}(\widetilde{\eta}_{j}(\mathbf{x})) \mathbf{F}(\mathbf{x})) - \mathbf{F}'(\widetilde{\eta}_{j}(\mathbf{x})) \right]^{-1} \mathbf{F}(\mathbf{x})$$

$$(\mathbf{j} = 0, 1, \dots, N).$$

 $D(\widetilde{G})$ is defined in a way similar to D(G) (cf. section 4.2.2). \widetilde{G} can be conceived as an iteration function of an iterative method, which we denote by $\widetilde{M}_{N,\theta}$. Hence $\widetilde{G} = \widetilde{M}_{N,\theta}(F)$.

N, θ N, θ N, θ We notice that for N = 0, iterative methods $\widetilde{M}_{N,\theta}$ have been investigated by several authors (cf. [TRAUB, 1964; p. 164], where it is assumed that E = IR; for the case that E is an arbitrary Banach space, an example is given, for instance, in [KOGAN, 1967]). Just as with the method of tangent hyperbolas, the main purpose of investigating these methods was their local convergence behaviour (*near* x^{*}). Our main interest, however is in what happens when the starting point x₀ is *remote* from x^{*}.

<u>CLASS 2</u>. We shall also consider the optimal methods $M_{\overline{\omega}}$ defined in Definition 6.3.1, where $\overline{\omega} = \overline{\pi}[N]$ (cf. (6.6.13)) with N ϵ IN given. We notice that the iterative processes $[M_{\overline{\pi}[N]}, F]$ (N ϵ IN) are all quadratically convergent (cf. p. 87).

<u>CLASS 3</u>. Let N \geq 0 and let $\mathcal{H} = \{h_0, h_1, \dots, h_N\}$ with $h_i = \frac{1}{N+1}$ (i = 0,1,...,N). Let $g_1, g_2 \in S$ with

$$(7.1.3a)$$
 $g_1 = 0$

and

(7.1.3b)
$$g_2(t) = \begin{cases} \frac{1}{1-t} & (\text{if } t \in [0,1)), \\ 1 & (\text{if } t = 1). \end{cases}$$

We shall also consider the methods $M_{i,j,N}$ where

$$(7.1.4) \qquad M_{i,j,N} = IM(A,g_{i},L_{j},H) \qquad (i = 1,2; j = 1,2,3).$$

Here, L_1 and L_2 are defined in (3.3.6). The Runge-Kutta method L_3 is the so-called *Modified Euler method* (cf. [LAMBERT, 1973; p. 118]) and is defined by

(7.1.5)
$$L_3 = \begin{pmatrix} 0 & 0 & 0 \\ \frac{1}{2} & 0 & 0 \\ 0 & 1 & 0 \end{pmatrix}$$
.

From Theorem 5.2.5 it follows that the iterative processes $[M_{i,j,N},F]$ (i = 1,2; j = 1,2,3) are quadratically convergent.

We shall test iterative methods $\tilde{M}_{N,\theta}$ of Class 1 with $\theta = 0, \frac{1}{2}, 1$. The choice of these methods has been motivated in section 3.3.

Methods of classes 2 and 3 have the following features, which motivated their choice.

- (a) The methods of Class 2, for which we could determine the radius of convergence, are all optimal according to the theory given in Part II of Chapter 6.
- (b) The methods $M_{i,j,N}$ (i,j = 1,2) are based on Runge-Kutta methods that are superior to Euler's method for certain types of differential equations (see section 3.3).
- (c) The methods $M_{i,j,N}$ (i,j = 1,2) are based on *first order* Runge-Kutta methods which require per step at *least two* evaluations of the function f (appearing on the right-hand side of the differential equation). The methods $M_{i,3,N}$ (i = 1,2) are based on a *second order* Runge-Kutta method which requires *two* such function evaluations per step.
- (d) The methods $M_{\overline{\pi}[N]}$ and $M_{2,j,N}$ (j = 1,2,3) are both of the type (4.1.3), with the same A and g. We thus have $M_{\overline{\pi}[N]} = \mathbb{M}(A,g_2,L_0,\widetilde{H})$ and $M_{2,j,N} = \mathbb{M}(A,g_2,L_j,H)$ where A, g_2 , H and L_j (j = 1,2,3) are defined above. L_0 denotes Euler's method and $\widetilde{H} = (\widetilde{h}_0,\widetilde{h}_1,\ldots,\widetilde{h}_N)$ denotes the sequence of stepsizes which corresponds to the optimal $\overline{\omega} = (\widetilde{\omega}_0,\widetilde{\omega}_1,\ldots,\widetilde{\omega}_N) = \overline{\pi}[N]$ by means of (6.3.4).
- (e) With j \in {1,2,3} and N \ge 0 both fixed, the iterative methods M _____i,N

(i = 1, 2) differ only with regard to function g_i (i = 1, 2).

7.2. NUMERICAL RESULTS WITH THE METHODS DESCRIBED IN SECTION 7.1

Iterative methods of the Classes 1, 2 and 3 have been applied to two problems.

PROBLEM 1. This problem arises from a finite-difference approach to the onedimensional two-point boundary value problem

$$\frac{d}{ds} \{s^2 \frac{d}{ds} V(s)\} - s^2 f(V(s)) = 0 \qquad (0 < s < 1),$$

(7.2.1)

$$V'(0) = 0, V(1) = 1$$

where $f(v) \equiv \varepsilon^{-1} \frac{v}{v+\lambda}$ (see [KELLER, 1968; p. 162 et seq.]). On physical grounds the solution of (7.2.1) looked for should be positive and continuous. It can be shown that a unique positive solution exists, which is strictly isotone. In [MURRAY, 1968] it is shown that $V(0) \sim 2[\varepsilon\lambda]^{-\frac{1}{2}} \exp\{-[\varepsilon\lambda]^{-\frac{1}{2}}\}$ (if $\varepsilon\lambda << 1$).

Let $m \in IN$ and let $\varepsilon, \lambda > 0$. Consider the (m+1)-dimensional problem F(x) = 0, where for $x = (\xi_0, \xi_1, \dots, \xi_m)$ and $F(x) = (\phi_0(x), \phi_1(x), \dots, \phi_m(x))$

$$\begin{aligned} \phi_{0}(\mathbf{x}) &= [\mathbf{s}_{\frac{1}{2}}]^{2} (\xi_{0} - \xi_{1}), \\ \phi_{j}(\mathbf{x}) &= -[\mathbf{s}_{j-\frac{1}{2}}]^{2} \xi_{j-1} + ([\mathbf{s}_{j-\frac{1}{2}}]^{2} + [\mathbf{s}_{j+\frac{1}{2}}]^{2}) \xi_{j} - [\mathbf{s}_{j+\frac{1}{2}}]^{2} \xi_{j+1} \\ &+ \Delta^{2} [\mathbf{s}_{j}]^{2} f(\xi_{j}) \qquad (j = 1, 2, \dots, m-1), \\ \phi_{m}(\mathbf{x}) &= -[\mathbf{s}_{m-\frac{1}{2}}]^{2} \xi_{m-1} + ([\mathbf{s}_{m-\frac{1}{2}}]^{2} + [\mathbf{s}_{m+\frac{1}{2}}]^{2}) \xi_{m} - [\mathbf{s}_{m+\frac{1}{2}}]^{2} \\ &+ \Delta^{2} [\mathbf{s}_{m}]^{2} f(\xi_{m}). \end{aligned}$$

Here,
$$\Delta = \frac{1}{m+1}$$
 and $s_i = i \Delta$ $(i = \frac{1}{2}, 1, ..., m + \frac{1}{2})$. Set
(7.2.3) $D(F) = \{x \mid x \in \mathbb{R}^{m+1}; x = (\xi_0, \xi_1, ..., \xi_m) \text{ where } \xi_j > 0$
 $(j = 0, 1, ..., m)\}.$

It can be shown that F'(x) is invertible (for all $x \in D(F)$) and that F(x) = 0

has a solution that is unique in D(F). Consequently, $F \in F_1$. The starting point x_0 was chosen to be $x_0 = (\xi_0^0, \xi_1^0, \dots, \xi_m^0)$ where $\xi_j^0 = (1-\epsilon\lambda)[s_j]^2 + \epsilon\lambda$ (j = 0,1,...,m). The computations were performed for m = 99 and $\lambda = 0.1$. For ε we took $\varepsilon = 0.05$, 0.01 and 0.001.

In all these three cases, Newton's method failed. More specifically, it generated sequences $\{x_k^{}\}$ that converged to a "solution" outside D(F).

Any iterative process was considered to yield a sequence $\{\boldsymbol{x}_k\}$ converging to a solution of the equation F(x) = 0 whenever, for some k with $1 \leq k \leq 20$,

$$(7.2.4) \quad \|\mathbf{x}_{k} - \mathbf{x}_{k-1}\| \le \delta_{1} (1 + \|\mathbf{x}_{k}\|) \quad \text{or} \quad \|\mathbf{F}(\mathbf{x}_{k})\| \le \delta_{2}$$

(see e.g. [BUS, 1975]). For both δ_1 and δ_2 we took 10⁻⁶. The norm used was the Euclidean norm. The number of iteration steps required for the stopping criterion to be satisfied is only given if a process succeeded in generating a sequence $\{x_k\}$ that converged to the desired solution. In all other cases we assume that the iterative process failed and indicate this by FAILURE.

TABLE 7.2.1.

Method Class 1 (cf. (7.1.2); $\tau = 10^{-4}$). Problem 1 ($\lambda = 0.1$, m = 99).

method	N	0	0	0	1	1	1
Problem	θ	1	0.5	0	1	0.5	0
ε = 0.05		FAILURE	3	4	4	2	3
$\varepsilon = 0.01$		FAILURE	FAILURE	5	FAILURE	3	4
$\varepsilon = 0.001$		FAILURE	FAILURE	7	FAILURE	4	5

method	N	3	3	3	7
Problem	θ	1	0.5	0	1
ε = 0.05		3	2	3	3
$\varepsilon = 0.01$		4	2	4	3
$\varepsilon = 0.001$		FAILURE	2	4	FAILURE

TABLE 7.2.2.
Method Class 2 (cf. (6.3.3), (6.6.13)).
Problem 1 ($\lambda = 0.1, m = 99$).

method N	1	3	7
Problem			
$\varepsilon = 0.05$	3	2	1
$\epsilon = 0.01$	FAILURE	FAILURE	FAILURE
ε = 0.001	FAILURE	FAILURE	FAILURE

TABLE 7.2.3.

Method Class 3, M (cf. (7.1.4), (7.1.3a), (3.3.6) and (7.1.5)). Problem 1 (λ = 0.1, m = 99).

			- 1		1 1	1	1	3	3
method	N	0	0	0	1	-			
		1	2	3	1	2	3	1	3
Problem	1	1				_		2	2
0.05		FATLURE	8	4	3	3	2	5	2
E = 0.05		1111201-		_	TINTIDE	FATLURE	4	4	3
$\epsilon = 0.01$		FAILURE	FAILURE	/	FAILORE	TATLOIL			
			TA TT LIDE	FATLIBE	FATLURE	FAILURE	.7	FAILURE	3
$ \epsilon = 0.001$		FAILURE	FALLORE	FALLORE					

TABLE 7.2.4.

Method Class 3, M_{2,j,N} (cf. (7.1.4), (7.1.3b), (3.3.6) and (7.1.5)). Problem 1 (λ = 0.1, m = 99).

•	1		0	0		1	1	3	3
method	N	0	0		1	2	3	1	3
Problem	i	1	2	3	1	~			
	-		7	8	3	3	3	2	2
$\varepsilon = 0.05$		FALLORE	, ,	J			TATIDE	2	3
$\epsilon = 0.01$		FAILURE	FAILURE	FAILURE	FAILURE	FAILURE	FAILORE	2	-
0.01				TATIDE	FATLURE	FAILURE	FAILURE	FAILURE	FAILURE
$\varepsilon = 0.001$		FAILURE	FAILURE	FALLORE	FILLOID				

<u>PROBLEM 2</u>. As a second example we consider the problem F(x) = 0, where for $x = (\xi_1, \xi_2)$ and $F(x) = (\phi_1(x), \phi_2(x))$

(cf. [CARNAHAN, LUTHER & WILKES, 1969; p.319]). This problem has a solution $x^{\star}\approx$ (0.30, 2.84). Set

$$(7.2.6) D(F) = \{y_0 \mid y_0 \in \mathbb{R}^2; \text{ a continuous curve } X \text{ on } [0,1] \text{ with } X(0) = y_0 \text{ and } X(1) = x^* \text{ exists, such that } F'(X(t)) \text{ exists and } \text{ is invertible and } F(X(t)) - (1-t)F(y_0) = 0 \text{ (for all } t \in [0,1])\}.$$

It can be shown that D(F) is open. The equation F(x) = 0 also has two "solu-. tions", y^* and z^* , outside D(F). $y^* = (0.5,\pi)$ and $z^* \approx (-0.26, 0.62)$. The starting point was chosen to be $x_0 = (0.4, 3)$. It appears that $x_0 \in D(F)$ (cf. [BOGGS, 1971]). In this case Newton's method failed in the sense that it generated a sequence $\{x_k\}$ converging to z^* instead of x^* .

 $\frac{\text{TABLE 7.2.5}}{(\text{cf. (7.1.2); } \tau = 10^{-4})}.$ Problem 2.

method	N	0	0	0	1	1	1
	θ	1	0.5	(,	1	0.5	0
		FAILURE	5	7	FAILURE	3	5

method	N	3	3	3	7
	θ	1	0.5	0	1
		4	3	4	3

TABLE 7.2.6.

Method Class 2 (cf. (6.3.3), (6.6.13)). Problem 2.

method	N	1	3	7
		FAILURE	FAILURE	2

				TAB	LE 7.2.7.			
Method	Class	3	(cf.	(7.1.4),	(7.1.3),	(3.3.6)	and	(7.1.5)).
				Pro	oblem 2.			

1 1	N	0	0	0	1	1	1	3	3	
	j	1	2	3	1	2	3	1	3	
method	^M 1,j,N	4 FATLURE	3	3 FAILURE	3 3	3 2	3 FAILURE	. 3 2	3 3	
	^M 2,j,N	PATHOICE	-					1		Т

<u>CONCLUSION</u>. The methods $\widetilde{M}_{N,\theta}$ of Class 1 with $\theta = 0$ and $\theta = \frac{1}{2}$, and the methods $M_{1,3,N}$ of Class 3 appear to be more reliable than the other methods whose test results have been presented above (including Newton's method $\widetilde{M}_{0,1}$, which failed).

The work per step for a method $\widetilde{M}_{N,\theta}$ (with $\theta \neq 1$) and for a method $M_{1,3,N}$ is roughly twice the work for a method $\widetilde{M}_{N,\theta}$ (with $\theta = 1$) and for a method $\widetilde{M}_{N,\theta}$ $M_{\overline{\pi}[N]}$ - with the same $N \ge 0$. Even if we compare methods $\widetilde{M}_{N,\theta}$ (with $\theta = 0, \frac{1}{2}$) and $M_{1,3,N}$ with the methods $\widetilde{M}_{N,\theta}$ (with $\theta = 1$) and $M_{\overline{\pi}[N]}$ that require the same amount of work per iteration step, the former (especially $\widetilde{M}_{N,\theta}$ with $\theta = 0$) turn out to be better.

Among the methods $\widetilde{M}_{N,\theta}$ ($\theta = 0, \frac{1}{2}$) and $M_{1,3,N}$, the methods $\widetilde{M}_{N,\theta}$ with $\theta = 0$ (in which case the Backward Euler method underlies the iterative methods) appear to be more reliable than the methods $\widetilde{M}_{N,\theta}$ with $\theta = \frac{1}{2}$ (in which case the Trapezoidal rule underlies the iterative methods) and $M_{1,3,N}$ (in which case the Modified Euler method underlies the iterative methods). However, if small stepsizes are taken (i.e. N large) then the last two types of methods generate sequences that converge faster to the solution than the first ones.

If we restrict our attention to the methods of Classes 2 and 3, methods $M_{1,3,N}$ of Class 3 appear to be more reliable than the other ones. The methods $M_{\pi[N]}$ and the methods $M_{2,j,N}$, which are closely related to one another, appear to have about the same (non-) convergence behaviour.

If we compare the methods of Class 3 with one another with regard to the function g, the iterative methods that are based on g_1 (cf. (7.1.3a)) appear to be superior to the iterative methods based on g_2 (cf. (7.1.3b)). This indicates that the choice of the function g may significantly affect the convergence behaviour of an iterative method of the type (4.1.3b). It therefore seems worth while to investigate in future research the role of function g in the convergence behaviour of iterative processes [M,F] where

M is of the type (4.1.3b) and x_0 is remote from x^* . In conclusion, the methods $\widetilde{M}_{N,\theta}$ appear to be very reliable for solving the problem F(x) = 0. The investigation of these methods in future research would therefore seem to be desirable.

CHAPTER 8

PREDICTOR-CORRECTOR CONTINUATION ALGORITHMS

Let F ϵ F₁. So far, we have only been interested in *iterative processes* for solving

(8.1.0) F(x) = 0.

In [RHEINBOLDT, 1975] an *algorithm* for solving (8.1.0) is proposed which is based on discrete imbedding. It has an adaptive step strategy for determining the partition $\{t_0, t_1, \ldots, t_N\}$ (cf. section 1.1).

In section 8.1 we give an outline of this algorithm. In sections 8.2, 8.3 and 8.4 we go into more detail, and present some variants of the algorithm, in some of which Davidenko's method is used also (cf. section 1.1). These variants also use the results given in section 6.5. In some of them Newton's method is used, while in the others the iterative method $\widetilde{M}_{0,0}$ is used also (cf. (7.1.2)). In section 8.5 we present some numerical results obtained with the algorithms described here.

In this chapter we assume that E is finite-dimensional. Throughout the sections 8.1-4 the operator F denotes a given (fixed) element of F_1 .

8.1. INTRODUCTION

Let $u_0 \in D(F)$ and let

Q: $[0,1] \times D(F) \to E$,

(8.1.1)

 $Q(t,x) \equiv F(x) - (1-t)F(u_0)$

(cf. (1.1.4)). Suppose that for all t ϵ [0,1] the equation

(8.1.2) Q(t,x) = 0

has a unique solution x = U(t), which depends continuously on t.

We recall that the method of discrete imbedding consists of selecting a partition $P = \{t_0, t_1, \dots, t_N\}$ with $0 = t_0 < t_1 < \dots < t_N = 1$ and of solving (8.1.2) successively for $t = t_i$ (i = 1,2,...,N), (cf. (1.1.9)).

Let M be the iterative method (the *local method*) that is used for approximating the solution $U(t_i)$ of (8.1.2) (i = 1,2,...,N). By r(t) we denote the radius of convergence (see Definition 2.4.1) of the iterative process [M,Q(t, \cdot)] (t ϵ [0,1]).

We shall now describe how a partition P and starting points $y_{i,0}$ for $[M,Q(t_i,\cdot)]$ (the *i-th local process*) are selected in the algorithms to be described here.

Suppose the algorithm has progressed through t_1, t_2, \ldots, t_k with $t_k < 1$ and $k \ge 1$, so that close approximations u_i of $U(t_i)$ (i = 1,2,...,k) have been obtained. We indicate how t_{k+1} and $y_{k+1,0}$ are selected (we shall specify what follows further on).

1. An approximation ${\rm U}_{\rm k}$ of U is constructed. Thus

(8.1.3)
$$U_{t}(t) \approx U(t)$$
 (t $\in [0,1]$).

2. $\|U_{t}(t) - U(t)\|$ is estimated by $\varepsilon_{t}(t)$ (for all $t \ge t_{t}$). Thus

$$(8.1.4) \qquad \varepsilon_k(t) \approx \| \mathbf{U}_k(t) - \mathbf{U}(t) \| \qquad (t \ge t_k).$$

3. r(t) is estimated by $r_k(t)$ (for all $t \ge t_k$). Thus

(8.1.5)
$$r_k(t) \approx r(t)$$
 $(t \ge t_k)$.

4. Let h_k be the solution of

(8.1.6)
$$\varepsilon_k(t_k + h) = \gamma_1 r_k(t_k + h)$$
 (h > 0),

where $\gamma_1,\; 0 < \gamma_1 \leq 1,\; is a given number which reflects the uncertainties of the estimates. Then$

$$t_{k+1} = t_k + h_k$$
.

5. Set

 $y_{k+1,0} = U_k(t_{k+1})$. (8.1.7)

 $y_{k+1,0}$ is the starting point of the local process [M,Q(t_{k+1}, \cdot)].

Observe that, when $r_k(t) \equiv r(t)$ and $\varepsilon_k(t) \equiv \|U_k(t) - U(t)\|$, for any $\gamma_1 \in (0,1)$ the (k+1)-th local process yields a sequence $\{y_{k+1,j}\}_{j=0,1,2,...}$ that converges to $U(t_{k+1})$.

In the algorithms to be presented here, $r_k^{}(t)$ Ξ $\rho_k^{}$ for some constant $\rho_k > 0$ and either $\varepsilon_k = 0$, or ε_k is strictly isotone on $[t_k, \infty)$, $\varepsilon_k(t_k) = 0$ and $\lim_{t\to\infty} \varepsilon_k(t) = \infty$. In the first case we set $t_{k+1} = 1$, in the last case the equation (8.1.6) has a unique solution.

Following Rheinboldt ([RHEINBOLDT, 1976]) we call an algorithms of the type described above a predictor-corrector continuation (PCC) algorithm $(U_k \text{ is the predictor and } M \text{ is the corrector}).$

In the following survey of the algorithm, we also give a more detailed description of the way in which the stepsize is selected.

1. $k := 0; t_k := 0; h_k := h_{start}; go to 3.$ 2. h_k is the solution of (8.1.6).

 $\mathbf{t}_{k+1} := \begin{cases} \mathbf{t}_{k} + \mathbf{h}_{\min} & (\text{if } \mathbf{h}_{k} < \mathbf{h}_{\min}), \\ \mathbf{t}_{k} + \mathbf{h}_{k} & (\text{if } \mathbf{h}_{\min} \leq \mathbf{h}_{k} \leq \mathbf{h}_{\max}), \\ \mathbf{t}_{k} + \mathbf{h}_{\max} & (\text{if } \mathbf{h}_{k} > \mathbf{h}_{\max}). \end{cases}$

If $t_{k+1} > 1$ then $t_{k+1} := 1$; $y_{k+1,0} := U_k(t_{k+1})$. Perform [M,Q(t_{k+1},·)] with starting point y_{k+1},0

If convergence criterion is met for some y_{k+1}^{ℓ}, ℓ with $\ell \leq \ell_{\max}$,

then $(u_{k+1} := y_{k+1}, \ell;$ if $t_{k+1} = 1$ then STOP else (k := k+1; go to 2)) else if $t_{k+1} - t_k = h_{min}$ then FAILURE

З.

else $(h_k := \gamma_2 h_k; go to 3)$.

The constant $\gamma_2 \in (0,1)$ is given.

In sections 8.2-4 we specify how U $_k$ (k = 0,1,...), and ϵ_k and r_k (k = 1,2,...) are determined. We assume that for all t ϵ [0,1] the operator F'(x) exists and is invertible (for all x in some open neighbourhood of U(t)).

Hence, from Theorem 2.6.4(i) it follows that

$$\dot{U}(t) = -\Gamma(U(t))F(u_0)$$
 (t $\in [0,1]$),
(8.1.8)
 $U(0) = u_0$.

In sections 8.2-4 we suppose the algorithm has progressed through t_0, t_1, \ldots, t_k with $k \ge 0$. Hence close approximations u_0, u_1, \ldots, u_k of respectively $U(t_0), U(t_1), \ldots, U(t_k)$ are available.

8.2. DETERMINATION OF Uk

We shall describe three types of approximations ${\tt U}_k$ which we tested. We suppose $k\,\geq\,0\,.$

a. Lagrange interpolation. Let an integer p with $0 \le p \le k$ be given. Choose $U_k = U_{k,p}$, where

(8.2.1a)
$$U_{k,p}(t) \equiv \sum_{j=0}^{p} \pi_{k,p,j}(t) u_{k-j}$$

and

(8.2.1b)
$$\pi_{k,p,j}(t) \equiv \prod_{\substack{\ell=0\\ \ell\neq j}}^{p} \left[\frac{t-t_{k-\ell}}{t_{k-j}-t_{k-\ell}} \right] \qquad (j = 0, 1, \dots, p).$$

It follows that

$$(8.2.2) \qquad U_{k,p}(t_{k-j}) = u_{k-j} \qquad (j = 0, 1, \dots, p).$$

In [RHEINBOLDT, 1975] U_k is of this type. b. <u>Interpolation, using relation (8.1.8)</u>. Let an integer p with $0 \le p \le k$ be given. Set

(8.2.3)
$$\dot{u}_{k} = -\Gamma(u_{k})F(u_{0}).$$

Choose $U_k = \overline{U}_{k,p}$, where

(8.2.4a)
$$\bar{U}_{k,p}(t) \equiv u_0$$
 (if k = 0)
and

$$(8.2.4b) \quad \overline{v}_{k,p}(t) \equiv \sum_{j=1}^{p} \frac{t-t_{k}}{t_{k-j}-t_{k}} \pi_{k,p,j}(t) u_{k-j} + \pi_{k,p,0}(t) [(1 - \dot{\pi}_{k,p,0}(t_{k})(t-t_{k})) u_{k} + (t-t_{k}) \dot{u}_{k}]$$

$$(if \ k \ge 1).$$

Here $\pi_{k,p,j}$ is defined in (8.2.1b). It is easily verified that, when $k \ge 1$,

(8.2.5a)
$$\overline{U}_{k,p}(t_{k-j}) = u_{k-j}$$
 (j = 0,1,...,p)

and

(8.2.5b)
$$\overset{\bullet}{\overline{U}}_{k,p}(t_k) = \overset{\bullet}{u}_k.$$

This type of interpolation curve is, for example, used in [BYRNE & HINDMARSH, 1975] as a predictor for implicit backward differentiation formulae for the numerical solution of ordinary differential equations.

c. <u>Approximation</u>, using a Runge-Kutta method with operator coefficient. Consider the one-stage Runge-Kutta method with operator coefficient $R = (\rho_{i,j})$ where $\rho_{2,1}(z) \equiv [1-z]^{-1}$ (see section 3.3 and p.121); R is related to the Backward Euler method.

Let $\hat{U}_k(t)$ be the approximation through u_k of U(t), which is obtained by performing one integration step of R on problem (8.1.8). Thus

$$\hat{U}_{k}(t) \equiv u_{k} - (t - t_{k}) [I - (t - t_{k}) \Gamma(u_{k}) F''(u_{k}) \Gamma(u_{k}) F(u_{0})]^{-1} \Gamma(u_{k}) F(u_{0}).$$

As in (7.1.2) we approximate $F''(u_k)\Gamma(u_k)F(u_0)$ by

$$\frac{1}{\gamma_{3}} \{ F'(u_{k} + \gamma_{3}\Gamma(u_{k})F(u_{0})) - F'(u_{k}) \}.$$

Here, $\gamma_3 > 0$ is a given small number (cf. p.122). We then obtain the approximation \check{U}_k to U,

$$(8.2.6a) \qquad \breve{U}_{k}(t) \equiv u_{k} - (t-t_{k}) [F'(u_{k}) - \frac{(t-t_{k})}{\gamma_{3}} [F'(u_{k}+\gamma_{3}\Gamma(u_{k})F(u_{0})) - F'(u_{k})]^{-1} F(u_{0}).$$

Compared to the predictor $U_{k,p}$, the determination of $\bar{U}_{k,p}$ with $k \ge 1$, requires approximately one Newton-step extra work. For $k \ge 1$ we therefore approximate \dot{u}_k by $\Gamma(y_{k,\ell-1})F(u_0)$, where $y_{k,\ell-1}$ is the last but one iterate of the k-th local process (the L-U decomposition of F'($y_{k,\ell-1}) = \partial_2 Q(t_k,y_{k,\ell-1})$ is available from the local process).

Similarly, if $k \ge 1$, instead of \check{U}_k , defined in (8.2.6a), we use

(8.2.6b)
$$\widetilde{U}_{k}(t) \equiv u_{k} - (t-t_{k}) [F'(Y_{k,\ell-1}) - \frac{t-t_{k}}{\gamma_{3}} \{F'(Y_{k,\ell-1} + \gamma_{3}\Gamma(Y_{k,\ell-1})F(u_{0})) - F'(Y_{k,\ell-1})\}]^{-1} F(u_{0}).$$

When compared to $U_{k,p}$, the amount of extra work required for $\widetilde{U}_{k,p}$ is approximately equal to one Newton-step. For ease of reference, we put $\widetilde{U}_{0} = \breve{U}_{0}$.

8.3. ESTIMATION OF $\|U_k(t) - U(t)\|$

In this section we present the estimates of $||U_k(t)-U(t)||$ we use in the algorithms. We suppose $k \ge 1$. We shall use the following lemma.

LEMMA 8.3.1. Let p and q be integers with $-1 \le q \le p \le k$ and $p \ge 0$. Set p+q+1 = n. Let Y: $[0,1] \rightarrow E$ be (n+1)-times continuously differentiable on [0,1] and suppose

(8.3.1)
$$\max_{t \in [0,1]} \|\frac{d^{n+1}}{dt^{n+1}} Y(t)\| \leq \delta < \infty.$$

Let $\texttt{Y}_n: \texttt{[0,1]} \rightarrow \texttt{E}$ be a polynomial of legree \leq n satisfying

$$(8.3.2a) \qquad Y_n(t_{k-j}) = Y(t_{k-j}) \qquad (j = 0, 1, ..., p)$$

and

(8.3.2b)
$$\dot{y}_{n}(t_{k-j}) = \dot{y}(t_{k-j})$$
 $(j = 0, 1, ..., q)$.

Then

$$(8.3.3) \qquad \|Y_{n}(t) - Y(t)\| \leq \frac{\delta}{(n+1)!} \prod_{j=0}^{p} |t - t_{k-j}| \prod_{j=0}^{q} |t - t_{k-j}|$$

(for all t ϵ [0,1]).

PROOF. 1. If E = IR then this result can be found in e.g. [STUMMEL & HAINER, 1971; section 3.2.1, 3.2.2].

2. Suppose $E = \mathbb{R}^{\mathbb{M}}$ (with $\mathbb{m} > 1$). Let $\tau_0 \in [0,1]$. Suppose $\Delta(\tau_0) \neq 0$ where $\Delta(\tau_0) = \Upsilon_n(\tau_0) - \Upsilon(\tau_0)$. Then a linear functional L: $E \rightarrow \mathbb{R}$ exists with $\|L\| = 1$ such that $L\Delta(\tau_0) = \|\Delta(\tau_0)\|$.

Let LY = η and LY_n = η_n . It is easily verified that η is (n+1)-times continuously differentiable on [0,1] and that $\max_{t \in [0,1]} |d^{n+1}/dt^{n+1} \eta(t)| \le \delta$. Further $\eta_n(t_{k-j}) = \eta(t_{k-j})$ (j = 0,1,...,p) and $\dot{\eta}_n(t_{k-j}) = \dot{\eta}(t_{k-j})$ (j = 0,1, ...,q). Consequently (cf. part 1 of this proof)

$$|\eta_{n}(t)-\eta(t)| \leq \frac{\delta}{(n+1)!} \prod_{j=0}^{p} |t-t_{k-j}| \prod_{j=0}^{q} |t-t_{k-j}|$$
(for all $t \in [0,1]$).

In particular, since $\|\Delta(\tau_0)\| = |L\Delta(\tau_0)| = |\eta_n(\tau_0) - \eta(\tau_0)|$, relation (8.3.3) is true for $t = \tau_0$.

Consequently, (8.3.3) is true for all t ϵ [0,1]. This proves the lemma.

a. Let $p \ge 0$ (with $p \le k-1$) be given. Suppose that U is (p+2)-times continuously differentiable, so that a constant $\delta_1 > 0$ exists such that (8.3.1) holds with Y = U, n = p+1 and $\delta = \delta_1$. Let $U_k = U_{k,p}$ (cf. (8.2.1)). If $u_{k-j} = U(t_{k-j})$ (j = 0,1,...,p) then

(cf. (8.3.3)). It is easily verified that

(8.3.4)
$$U_{k,p}(t) - U_{k,p+1}(t) \equiv \begin{bmatrix} p \\ \Pi \\ j=0 \end{bmatrix} (t-t_{k-j})]v, \text{ where } v \in E.$$

In this case we therefore set $\boldsymbol{\epsilon}_k = \boldsymbol{\epsilon}_k$, where

(8.3.5)
$$\varepsilon_{k,p}^{(t)} \equiv ||U_{k,p}^{(t)} - U_{k,p+1}^{(t)}||$$

or, equivalently,

(8.3.6)

$$v = \{ \prod_{j=0}^{p} (t-t_{k-j}) \}^{-1} [U_{k,p}(t) - U_{k,p+1}(t)] \}$$

 $\boldsymbol{\varepsilon}_{k,p}^{}(t) \equiv \| \boldsymbol{v} \| \begin{array}{c} \boldsymbol{p} \\ \boldsymbol{j} = \boldsymbol{0} \end{array} | t - t_{k-j}^{} |, \quad \text{where}$

(for any t \in [0,1] with t \neq t_{k-j} (j = 0,1,...,p)).

This kind of estimate of $||U_k(t)-U(t)||$ is also used in [RHEINBOLDT, 1975].

<u>REMARK 8.3.1</u>. Determination of p. The - assumed to be unique - solution of (8.1.6) depends on p and we might choose $p = p_k$ such that the solution h of (8.1.6) is maximum. However, in general, too many changes in p are inadvisable. Following [RHEINBOLDT, 1975] we only tested on values of h corresponding to $p = p_{k-1} + \delta$ ($\delta = -1, 0, -1$). Moreover, whenever p was changed we kept it fixed for at least one further step.

b. Let $p \ge 0$ be given $(p \le k-1)$. Let $U_k = \overline{U}_{k,p}$ (cf. (8.2.4)). Like case a, we estimate $\|\overline{U}_{k,p}(t) - U(t)\|$ by $\|\overline{U}_{k,p}(t) - \overline{U}_{k,p+1}(t)\|$. As in (8.3.4) it can be shown that

$$\overline{\overline{U}}_{k,p}(t) - \overline{\overline{U}}_{k,p+1}(t) \equiv \begin{bmatrix} p \\ j \equiv 0 \end{bmatrix} (t-t_{k-j}) (t-t_k) \overline{v}, \text{ where } \overline{v} \in E.$$

We thus set $\varepsilon_k = \overline{\varepsilon}_{k,p}$, where

(8.3.7)
$$\overline{\varepsilon}_{k,p}(t) \equiv \|\overline{U}_{k,p}(t) - \overline{U}_{k,p^{-1}}(t)\|$$

or, equivalently,

$$\overline{\varepsilon}_{k,p}(t) \equiv \|\overline{v}\| \begin{bmatrix} p \\ \Pi \\ j \equiv 0 \end{bmatrix} |t-t_{k-j}| |t-t_k|, \text{ where}$$

(8.3.8)

$$\overline{\mathbf{v}} = \{ \begin{bmatrix} \mathbf{p} \\ \mathbf{\Pi} \\ \mathbf{j=0} \end{bmatrix} (\mathbf{t}-\mathbf{t}_{k-j})] (\mathbf{t}-\mathbf{t}_{k}) \}^{-1} [\overline{\mathbf{u}}_{k,p}(\mathbf{t}) - \overline{\mathbf{u}}_{k,p+1}(\mathbf{t})]$$

(for any t ϵ [0,1] with t \neq t_{k-j} (j = 0,1,...,p)).

The integer $p = p_k$ is determined in the way described in Remark 8.3.1.

c. Let $U_k = \hat{U}_k$. Suppose U is three-times continuously differentiable. Let $f(u) \equiv -\Gamma(u)F(u_0)$. Thus

(8.3.9)
$$U(t) = f(U(t))$$
 $(t \in [0,1]).$

It is easily verified that (cf. p.135)

$$\hat{U}_{k}(t) \equiv u_{k} + (t-t_{k})[I - (t-t_{k})f'(u_{k})]^{-1}f(u_{k}).$$

Define d_1 and d_2 such that

$$\hat{\mathbf{U}}_{k}(t) \equiv \mathbf{u}_{k} + (t-t_{k})[\mathbf{I} + (t-t_{k})f'(\mathbf{u}_{k})]f(\mathbf{u}_{k}) + \mathbf{d}_{1}(t)$$

and

$$U(t) \equiv U(t_k) + (t-t_k)\dot{U}(t_k) + \frac{1}{2}(t-t_k)^2 \frac{d^2}{dt^2} U(t_k) + d_2(t).$$

It is easily verified that a constant $\boldsymbol{\delta}_1 \in (0,\infty)$ exists such that, when |t-t_k| is sufficiently small,

$$\|d_{1}(t)\| \leq \delta_{1}|t-t_{k}|^{3}$$
.

Suppose

$$\max_{\substack{\text{max}\\ t\in[0,1]}} \frac{\|\frac{d^3}{dt^3} \upsilon(t)\| \leq \delta_2 < \infty.$$

Then

$$\|d_2(t)\| \leq \frac{\delta_2}{6} |t-t_k|^3$$
 (for all $t \in [0,1]$).

From (8.3.9) it follows that $d^2/dt^2 U(t) \equiv f'(U(t))f(U(t))$ (t $\in [0,1]$). Hence, if $u_k = U(t_k)$

$$\begin{aligned} \|\hat{\mathbf{U}}_{\mathbf{k}}(t) - \mathbf{U}(t)\| &\leq \|\mathbf{f}'(\mathbf{u}_{\mathbf{k}})\mathbf{f}(\mathbf{u}_{\mathbf{k}})\| \frac{|\mathbf{t} - \mathbf{t}_{\mathbf{k}}|^{2}}{2} + (\delta_{1} + \frac{\delta_{2}}{6})|\mathbf{t} - \mathbf{t}_{\mathbf{k}}|^{3} \\ &(|\mathbf{t} - \mathbf{t}_{\mathbf{k}}| \text{ sufficiently small}). \end{aligned}$$

We can therefore estimate $\|\hat{U}_{k}(t) - U(t)\|$ by $\hat{\varepsilon}_{k}(t)$, where

$$\hat{\varepsilon}_{k}(t) \equiv \|f'(u_{k})f(u_{k})\| \frac{|t-t_{k}|^{2}}{2}$$

or, equivalently,

$$\hat{\varepsilon}_{k}(t) \equiv \|\Gamma(u_{k})F''(u_{k})[\Gamma(u_{k})F(u_{0})]^{2}\| \frac{|t-t_{k}|^{2}}{2}$$

Similarly, we estimate $\|\check{\mathbf{U}}_{\mathbf{k}}(t) - \mathbf{U}(t)\|$ (cf. (8.2.6a)) by $\check{\mathbf{e}}_{\mathbf{k}}(t)$, where (8.3.10a) $\check{\mathbf{e}}_{\mathbf{k}}(t) \equiv \|\Gamma(\mathbf{u}_{\mathbf{k}})\frac{1}{\gamma_{3}}\{\mathbf{F}'(\mathbf{u}_{\mathbf{k}}+\gamma_{3}\Gamma(\mathbf{u}_{\mathbf{k}})\mathbf{F}(\mathbf{u}_{0})) - \mathbf{F}'(\mathbf{u}_{\mathbf{k}})\}\Gamma(\mathbf{u}_{\mathbf{k}})\mathbf{F}(\mathbf{u}_{0})\|\frac{|t-t_{\mathbf{k}}|^{2}}{2}$. Obviously, $\|\widetilde{\mathbf{U}}_{\mathbf{k}}(t) - \mathbf{U}(t)\|$ (cf. (8.2.6b)) is estimated by $\widetilde{\mathbf{e}}_{\mathbf{k}}(t)$, where (8.3.10b) $\widetilde{\mathbf{e}}_{\mathbf{k}}(t) \equiv \|\Gamma(\mathbf{y}_{\mathbf{k}}, \ell-1)\frac{1}{\gamma_{3}}\{\mathbf{F}'(\mathbf{y}_{\mathbf{k}}, \ell-1+\gamma_{3}\Gamma(\mathbf{y}_{\mathbf{k}}, \ell-1)\mathbf{F}(\mathbf{u}_{0})) - \mathbf{F}'(\mathbf{y}_{\mathbf{k}}, \ell-1)\}$ $\Gamma(\mathbf{y}_{\mathbf{k}}, \ell-1)\mathbf{F}(\mathbf{u}_{0})\|\frac{|t-t_{\mathbf{k}}|^{2}}{2}$.

8.4. LOCAL PROCESSES; ESTIMATION OF THE RADII OF CONVERGENCE

In this section we present the two types of local processes that are used in the algorithms whose test results are presented in section 8.5.

The first type uses Newton's method, which requires the least amount of work per iteration step of all the methods considered in chapter 7. Since Newton's method is quadratically convergent, this local process will, in general, only require a few steps. The second type of local process uses a combination of Newton's method and the method $\widetilde{M}_{0,0}$ (cf. (7.1.2)), which appears to have very good convergence behaviour (cf. Tables 7.2.1, 7.2.5).

We also tested algorithms in which the methods $M_{\pi[N]}$ (cf. (6.3.3), (6.6.13)) are used as local method. (The radius of convergence of these methods was estimated by using the theory of chapter 6, Part II.) However, in contradistinction to the iterative methods in chapter 7, almost all PCC algorithms which were tested, managed to deliver the desired solution (within the required accuracy). Consequently, it is only sensible to compare the PCC algorithms with respect to the total amount of work, which is roughly the total number of local steps required to solve the problem. It appeared that, when using a method $M_{\pi[N]}$ or a combination of $M_{\pi[N]}$ and Newton's method, the total amount of work was significantly higher than in the case in which Newton's method only was used.

We shall now specify the local process that uses Newton's method, and the one that uses $\widetilde{M}_{0,0}$ also. We assume $k \ge 1$. Both local processes contain the following control mechanism (for the k-th local process):

ITER:
$$y_{k,j+1} = y_{k,j} - d_{k,j};$$

$$(8.4.1a) \quad \text{if } (\|d_{k,j}\| < (1 + \|y_{k,j}\|) \delta_1 \text{ or } \|Q(t_k, y_{k,j})\| < \delta_2)$$

then "convergence criterion is met"

(8.4.1b) else if
$$\|d_{k,j}\| > \gamma_4 r_{k-1} (t_k)$$

then "convergence criterion is not met"

(8.4.1c) else if
$$((j \ge 7 \text{ and } \| \mathbf{d}_{k,j} \| < \delta_3 \text{ and } \| \mathbf{d}_{k,j} \| < \delta_4 \| \mathbf{d}_{k,j-1} \|) \text{ or } j < 7)$$

then (j := j+1; go to ITER)

(8.4.1d) else

"convergence criterion is not met".

Three remarks on this control mechanism.

1. We shall use $\delta_1 = \delta_2 = 10^{-6}$ in (8.4.1a). Other types of stopping criteria than requirement (8.4.1a) are possible (see e.g. [SCHWETLICK, 1975]). However, it is not yet clear which stopping criterion is optimal (in the sense that it minimizes the total number of local iteration steps), (see also [RIBARIC & SELISKAR, 1974] and [WACKER, 1977b]). 2. In requirement (8.4.1b), the constant $\gamma_4 \ge 2$ is given. If $r_{k-1}(t_k) = r(t_k)$ and if $\|d_{k,j}\| > \gamma_4 r_{k-1}(t_k)$ (for some $j \ge 0$) it seems reasonable to assume that $y_{k,j} \notin B(U(t_k), r(t_k))$. In general $r_{k-1}(t_k) \approx r(t_k)$. In practice it appeared that in cases where requirement (8.4.1b) was met, in general no convergence occurred (we chose $\gamma_4 = 10$). 3. We shall use $\delta_3 = \delta_4 = 10^{-2}$ in (8.4.1c). The requirements (8.4.1a,c) then guarantee that the number of steps in a local process is always less than 10. If after the 7 step no convergence is to be expected, the process is

broken off.

In the subsections 8.4.1 and 8.4.2 we assume that the k-th local process has been performed successfully, so that a sequence $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ has been generated where $y_{k,\ell} = u_k$. In these subsections we show how $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ has been obtained and how $r_k(t) \ (\approx r(t))$ is constructed. r_k will be of the type

(8.4.2)
$$r_k(t) \equiv \rho_k'$$

where ρ_k is an estimate of $r(t_k)$. Set

(8.4.3)
$$P(x) \equiv Q(t_k, x)$$
.

8.4.1. Newton's method

If Newton's method (subsequently denoted by ${\rm M}_{\underset{\mbox{\scriptsize 0}}{0}})$ is used as local method, then

(8.4.4a)
$$y_{k,j+1} = y_{k,j} - d_{k,j}^{0}$$

where

(8.4.4b)
$$d_{k,j}^{0} = [P'(y_{k,j})]^{-1}P(y_{k,j})$$
 $(j = 0,1,...,\ell-1).$

It is easily verified that $P \in F_1$ (cf. pp. 131, 133, and (2.6.1)).

Consider the function a, defined in (6.4.16). For all $\sigma \in (0,\infty]$ we have

$$(8.4.5a) \quad a(P;\sigma) = \begin{cases} \inf_{\substack{0 < \|U(t_{k}) - x\| < \sigma}} \frac{(x - U(t_{k}), [P'(x)]^{-1}P(x))}{\|x - U(t_{k})\|^{2} + \|[P'(x)]^{-1}P(x)\|^{2}} \\ (\text{if } B(U(t_{k}), \sigma) \subset D(P) \text{ and } P'(x) \text{ is invertible on } B(U(t_{k}), \sigma)), \\ -\frac{1}{2} \quad (\text{otherwise}). \end{cases}$$

We set

(8.4.5b)
$$a(P;0) = \frac{1}{2}$$
.

It follows that (cf. (6.4.18)) $P \in F < \sigma, a(P; \sigma) > (for all \sigma \in (0, \infty])$. Let

(8.4.6) $s_k = \sup\{\sigma \mid a(P;\sigma) - \frac{2}{5} > 0\}.$





By virtue of Theorem 6.5.1 we have $r(M_0; F < \sigma, a(P; \sigma) >) = \sigma$ whenever σ satisfies $a(P; \sigma) - \frac{2}{5} > 0$. Consequently

$$(8.4.7)$$
 $s_k \leq r(t_k)$.

 ρ_k will be an estimate of s_k ,

(8.4.8)
$$\rho_k \approx s_k$$
.

 s_k is determined by $a(P; \cdot)$. However, in general, we do not know $a(P; \cdot)$ explicitly. We therefore estimate $a(P; \cdot)$, using $y_{k,0}, y_{k,1}, \cdots, y_{k,\ell}$.

From Lemma 6.4.1, (6.4.17) and (8.4.5b) it follows that $a(P; \cdot)$ is antitone on $[0,\infty]$, the right-derivative of $a(P; \cdot)$ exists at $\sigma = 0$ and

(8.4.9)
$$\frac{\mathrm{d}}{\mathrm{d}\sigma} a(P;\sigma) = 0$$
 (for $\sigma = 0$).

We therefore estimate s_k as follows. For $j = 0, 1, \dots, l-1$ let

(8.4.10a)
$$\sigma_{k,j} = \| y_{k,j} - u_k \|$$

and

(8.4.10b)
$$\alpha_{k,j} = \frac{(y_{k,j} - u_k, d_{k,j})}{\|y_{k,j} - u_k\|^2 + \|d_{k,j}\|^2}$$



Fig. 8.4.2.

Define

(8.4.11a)
$$a_{k,j}(\sigma) \equiv \frac{1}{2} + c_{k,j}\sigma^2$$

where c is such that

(8.4.11b)
$$a_{k,j}(\sigma_{k,j}) = \alpha_{k,j}$$
 $(j = 0, 1, ..., l-1).$

Set

$$(8.4.12) \qquad a_{k}(\sigma) \equiv a_{k,j_{0}}(\sigma), \text{ where } c_{k,j_{0}} \leq c_{k,j} \qquad (j = 0,1,\ldots,\ell-1).$$

Then ρ_k is the solution of $a_k(\sigma) = \frac{2}{5} (\sigma > 0)$. Thus

(8.4.13)
$$\rho_k = [\sqrt{-10c_k, j_0}]^{-1}$$

(see also Fig. 8.4.2).

We notice that in [RHEINBOLDT, 1975] also, Newton's method is used as local method. However, in that paper, the estimate $r_k(t)$ of r(t) is based on a result similar to Theorem 6.2.1.

<u>REMARK 8.4.1</u>. It is easily verified that, when $\ell = \ell_k = 1$ and $\|y_{k,0} - y_{k,1}\| \neq 0$, then $c_{k,j0} = 0$. Hence ρ_k cannot be determined. (In fact $\rho_k = \infty$ so that $t_{k+1} - t_k = h_{max}$ (cf. p.133).) In such a case, either $r(t_k)$ is large (so that $\rho_k = \infty$ is a "good" estimate of $r(t_k)$) or $r(t_k)$ is small and it is only by chance that the k-th local process required only one step. In this last case, the stepsize $h_k = h_{max}$ may be too large. Unfortunately too little information is available to show us which is the case. Consequently, in order to prevent the (k+1)-th local process from failing we put (if $\ell_k = 1$)

(8.4.14)
$$h_k = \sqrt{(t_k - t_{k-1}) \times h_{max}}$$

so that the new step h_k will not differ too much from $t_k - t_{k-1}$.

<u>REMARK 8.4.2</u>. For k = 1, the control mechanism of the k-th local process should contain an estimate $r_0(t)$ of r(t), (cf. (8.4.1b)). However, r_0 cannot be determined in the way described above without making extra computations. For k = 1 we therefore drop the requirement (8.4.1b) (or, equivalently, we put $r_0(t) = \infty$).

8.4.2. A mixed method

The method $\widetilde{M}_{0,0}$ (cf. (7.1.2) with N = θ = 0) exhibited good convergence behaviour on the test problems of section 7.2. Suppose $\widetilde{M}_{0,0}$ is the local method, then $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ satisfies

(8.4.15a)
$$y_{k,j+1} = y_{k,j} - \tilde{d}_{k,j}$$

where

(8.4.15b)
$$\widetilde{d}_{k,j} = [P'(y_{k,j}) - \frac{1}{\tau} \{P'(y_{k,j} + \tau d_{k,j}^{0}) - P'(y_{k,j})\}]^{-1} P(y_{k,j})$$

(j = 0,1,...,l-1).

In (8.4.15b) the number $\tau > 0$ is a given (small) constant ($\tau = 10^{-4}$) and $d_{k,j}^0$ is defined by (8.4.4b). However, the work per step required for $\widetilde{M}_{0,0}$ is roughly twice the work required for Newton's method. In order to reduce the amount of work, we therefore use a mixed method which only uses $\widetilde{M}_{0,0}$ when the Newton-step may be expected to be bad. Using this mixed method

 $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ satisfies

(8.4.16a)
$$y_{k,j+1} = y_{k,j} - \bar{d}_{k,j}$$

where

$$(8.4.16b) \quad \bar{d}_{k,j} = \begin{cases} d_{k,j}^{0} & (cf. (8.4.4b)) & (if \|d_{k,j}^{0}\| < \gamma_{5} 2r_{k-1}^{0}(t_{k})), \\ \tilde{d}_{k,j} & (cf. (8.4.15b) & (otherwise) \\ & (j = 0, 1, \dots, \ell-1) \end{cases}$$

In our test examples we took for the constant γ_5 the values 0, $\frac{1}{2}$ and 1 (cf. section 8.5). If $\gamma_5 = 0$, then the mixed method reduces to $\widetilde{M}_{0,0}$. In (8.4.16b) $r_{k-1}^0(t_k)$ is an estimate of the radius of convergence of Newton's method for the k-th local problem. We have assumed the k-th local process has already been performed. In order to perform (8.4.16) for k+1 we should estimate the radius of convergence of Newton's method for the (k+1)-th local problem $(r_k^0(t_{k+1}))$. Since in the k-th local process $d_{k,j}^0$ is evaluated in each iteration step, we set $r_k^0(t_{k+1}) = \rho_k$, where ρ_k is defined by (8.4.13).

Let $\overline{r}(t_k)$ denote the radius of convergence of the mixed process (solving the k-th local problem). We shall only use the mixed method if $r(M_0, P) < r(\widetilde{M}_{0,0}, P)$ is to be expected (cf. (8.4.21)). If this is the case and if γ_5 is not large (cf. (8.4.16b) and (8.5.3)), it may be expected that $\overline{r}(t_k) = r(\widetilde{M}_{0,0}, P)$. We therefore use an estimate of $r(\widetilde{M}_{0,0}, P)$ as estimate of $\overline{r}(t_k)$.

We notice that $\widetilde{M}_{0,0}$ has been derived from $M_{0,0}$ (cf. (7.1.1)). $[M_{0,0},P]$ is equivalent to $[M_{0,0},P]$, where

$$\hat{\mathbf{P}}(\mathbf{x}) \equiv [\mathbf{P'}(\mathbf{x})]^{-1} \mathbf{P}(\mathbf{x}).$$

(i) Suppose $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ has been obtained by $[M_{0,0},P]$. Thus

$$y_{k,j+1} = y_{k,j} - \hat{a}_{k,j}$$

where

$$\hat{\hat{A}}_{k,j} = [\hat{P}'(y_{k,j})]^{-1} \hat{P}(y_{k,j})$$
 (j = 0,1,...,*l*-1).

Similarly to (8.4.13) we can estimate $r(M_0, \hat{P})$, which is equal to $r(M_0, 0, P)$, by

$$\hat{\rho}_{k} = [\sqrt{-10\hat{c}_{k,j_0}}]^{-1},$$

where $\hat{c}_{k,j_0} = c_{k,j_0}$, and j_0 and c_{k,j_0} are defined by (8.4.10-12) with $d_{k,j} = \hat{d}_{k,j}$ $(j = 0,1,\ldots,\ell-1)$. (ii) Suppose $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ has been obtained by $[\widetilde{M}_{0,0},P]$ (cf. (8.4.15)). Then, similarly, we estimate $r(\widetilde{M}_{0,0},P)$ by $\widetilde{\rho}_k$, defined by

(8.4.17)
$$\tilde{\rho}_{k} = \left[\sqrt{-10\tilde{c}_{k,j}}\right]^{-1}$$
.

Here, $\widetilde{c}_{k,j0} = c_{k,j0}$, and j_0 and $c_{k,j0}$ are defined by (8.4.10-12) with $d_{k,j} = \widetilde{d}_{k,j}$ (j = 0,1,...,l-1).

(iii) Suppose $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ has been obtained by the mixed method. Hence $\{y_{k,j}\}_{j=0,1,\ldots,\ell}$ satisfies (8.4.16). We would like to estimate the radius of convergence of the (k+1)-th local process by $\tilde{\rho}_k$ (cf. (8.4.17)). However, it may occur that $\bar{d}_{k,j} = d_{k,j}^0$ for some j, so that $\tilde{d}_{k,j}$ is not available. In general it is too expensive to calculate $\tilde{d}_{k,j}$ only for the purpose of estimating $\tilde{\rho}_k$. However, if $\|d_{k,j}^0\|$ is not large, then

$$\begin{aligned} d_{k,j}^{0} + d_{k,j+1}^{0} &= \hat{P}(Y_{k,j}) + \hat{P}(Y_{k,j} - \hat{P}(Y_{k,j})) \approx \\ &\approx [I + (I - \hat{P}'(Y_{k,j}))] \hat{P}(Y_{k,j}) \approx [I - (I - \hat{P}'(Y_{k,j}))]^{-1} \hat{P}(Y_{k,j}) \\ &= \hat{d}_{k,j} \approx \tilde{d}_{k,j}. \end{aligned}$$

Hence if $\overline{d}_{k,j} \neq \widetilde{d}_{k,j}$ and $0 \le j \le \ell-2$ we approximate $\widetilde{d}_{k,j}$ by $d_{k,j}^0 + d_{k,j+1}^0$. Thus, similarly to (8.4.17) we estimate $r(\widetilde{M}_{0,0}, P)$ by $\overline{\rho}_k$, defined by

(8.4.18)
$$\bar{\rho}_{k} = \gamma_{6} [\sqrt{-10\bar{c}_{k,j_{0}}}]^{-1}$$
.

Here, $\overline{c}_{k,j_0} = c_{k,j_0}$, and j_0 and c_{k,j_0} are defined by (8.4.10-12) with

$$(8.4.19) \qquad \mathbf{d}_{k,j} = \begin{cases} \widetilde{\mathbf{d}}_{k,j} & (\text{if } \overline{\mathbf{d}}_{k,j} = \mathbf{d}_{k,j}), \\ \mathbf{d}_{k,j}^{0} + \mathbf{d}_{k,j+1}^{0} & (\text{if } \overline{\mathbf{d}}_{k,j} = \mathbf{d}_{k,j}^{0}). \end{cases}$$

The factor γ_6 reflects the uncertainties in the estimate.

Let \overline{M} denote the mixed method. Suppose the estimate $\varepsilon_k(t)$ of $||U(t)-U_k(t)||$ satisfies $\varepsilon_k(t_k+h) = c|h|^2$ (with c > 0), (see e.g. (8.3.8) with p = 0, and (8.3.10)). Suppose furthermore that $\overline{M} = \widetilde{M}_{0,0}$. The method $\widetilde{M}_{0,0}$ requires twice as much work per iteration step as Newton's method. It therefore seems reasonable to use \overline{M} ($=\widetilde{M}_{0,0}$) only if this implies that the stepsize h_k is at least twice as large as it would be if Newton's method was used. Consequently, in cases where

(8.4.20)
$$r_k^0(t_{k+1}) = \rho_k \ge \gamma_7 \bar{\rho}_k = \gamma_7 \bar{r}_k(t_{k+1}),$$

where $\gamma_7 = \frac{1}{4}$, it seems advisable to use Newton's method instead of method \overline{M} (cf. (8.1.6)). Obviously, in general $\overline{M} \neq \widetilde{M}_{0,0}$. We therefore use Newton's method instead of \overline{M} whenever the (weaker) requirement (8.4.20) with $\gamma_7 = \frac{1}{2}$ (cf. (8.5.3)) is satisfied. We shall denote this combination of Newton's method and the mixed method by \overline{M}_0 . Hence

$$(8.4.21) \qquad \bar{M}_0 = \begin{cases} \text{Newton's method} & \text{(if (8.4.20) is true),} \\ \\ \text{the mixed method (cf. (8.4.16))} & \text{(otherwise).} \end{cases}$$

<u>REMARK 8.4.3</u>. When $\ell \le 2$ and Newton's method only has been used in the k-th local process, $\bar{\rho}_k$ cannot be determined (when $\ell = 2$, then $\bar{c}_{k,j_0} = 0$). In that case we use Newton's method only in the (k+1)-th local process.

REMARK 8.4.4. In the first local process we use Newton's method only. See also Remark 8.4.2.

8.5. NUMERICAL RESULTS WITH THE PCC ALGORITHMS

In all the test examples that are to be presented, we use

(8.5.1)

$$\delta_1 = \delta_2 = 10^{-6} \text{ (cf. (8.4.1a),}$$

 $\delta_3 = \delta_4 = 10^{-2} \text{ (cf. (8.4.1c)),}$

in the control mechanisms for the local methods.

1. As a first example, we consider the behaviour of several PCC algorithms when applied to Problem 1 (cf. p.125). The control parameters were chosen as follows:

(8.5.2)

$$\gamma_1 = 1$$
 (cf. (8.1.6)), $\gamma_2 = \frac{1}{2}$ (cf. p.133),
 $\gamma_3 = 10^{-4}$ (cf. (8.2.6b)), $\gamma_4 = 10$ (cf. (8.4.1b))

Furthermore, we set

$$h_{start} = h_{min} = 10^{-2}$$
, $h_{max} = 1$ (cf. p.133).

In Table 8.5.1 results are given where Newton's method is used as local method.

TABLE 8.5.1.

PCC algorithm.

Local method: M_0 .

Problem 1 ($\lambda = 0.1$, m = 99).

ſ	ε	predictor	total number of t-steps	total number of local steps	total amount of work
F	0.01	U	6	25	25
	0.01	ī,p	5	20	20
	0.01	, k,p	4	16	21
	0.01		8	37	37
	0.001	^k ,p	6	27	27
	0.001	~ ^k ,p		23	30
	0.001		0	23	

The total amount of work has been expressed in the number of Newton-steps required. (We assumed that the evaluation of $\widetilde{U}_k(t_{k+1})$ ($k \ge 1$) was equivalent to one Newton-step.) As an example, we specify the performance of one of the algorithms.

TABLE 8.5.2.

PCC algorithm.

Predictor: U_{k,p}.

Local method: M_0 .

Problem 1 ($\epsilon = 0.001$, $\lambda = 0.1$, m = 99).

k	t _k	p	$r_{k-1}(t)$ (estimate of the radius of conv.)	number of local steps
1	0.010	0	-	3
2	0.143	1	0.278 ₁₀ 1	6
3	0.227	1	0.111_{10}^{1}	5
4	0.364	2	0.88610	5
5	0.524	2	0.57610	5
6	0.715	2	0.40010	5
7	0.944	2	0.27810	5
8	1	2	0.191100	3

We also used the combination of the mixed method and Newton's method as local method. Its control parameters were chosen as follows:

$$\begin{split} \gamma_5 &= \frac{1}{2} \quad (\text{cf. (8.4.16b)})\,, \\ (8.5.3) \qquad \gamma_6 &= 1 \quad (\text{cf. (8.4.18)})\,, \\ \gamma_7 &= \frac{1}{2} \quad (\text{cf. (8.4.20)})\,. \end{split}$$

TABLE 8.5.3.

PCC algorithm. Predictor: $\widetilde{U}_{k}^{}$. Local method: $\overline{M}_{0}^{}$ ($\tau = 10^{-4}$, cf. (8.4.21), (8.4.15b)). Problem 1 ($\lambda = 0.1$, m = 99).

ε	total number of t-steps	total nur M ₀ -steps	$\widetilde{M}_{0,0}^{\text{mber of}}$	total amount of work
0.01	3	12	0	16
0.001	4	15	1	22

When both an iterative method and a PCC algorithm manage to solve a problem F(x) = 0, both starting from the same $u_0 = x_0$, the latter, in general, requires much more work than the former (see e.g. Table 7.2.1 (for $\widetilde{M}_{0,0}$) and Table 8.5.3). This phenomenon is due to the fact that PCC algorithms contain many control mechanisms which prevent divergence. They are therefore reliable but also rather expensive.

2. Secondly, we consider the behaviour of PCC algorithms when applied to Problem 3, a problem given in [RHEINBOLDT, 1975].

PROBLEM 3. This problem arises from a finite element approach to the twodimensional boundary value problem

$$(8.5.4a) \frac{\partial}{\partial v} [f(T;v,w)\frac{\partial}{\partial v} T(v,w)] + \frac{\partial}{\partial w} [f(T;v,w)\frac{\partial}{\partial w} T(v,w)] = c \\ ((v,w) \ \epsilon \ \Theta = [0,1] \times [0,1]), \\ T(v,w) = 0 \quad (if \ (v,w) \ \epsilon \ \partial \Theta, \ the \ boundary \ of \ \Theta).$$

In (8.5.4a)

(8.5.4b)
$$f(T;v,w) \equiv q\left(\left[\frac{\partial}{\partial v}T(v,w)\right]^2 + \left[\frac{\partial}{\partial w}T(v,w)\right]^2\right)$$

and

$$(8.5.4c) \qquad q(s) = \begin{cases} q_0 & \text{(if } s \le 0.15), \\ \frac{1}{2}(q_0 + q_1) + \frac{1}{4}(q_1 - q_0)(3\overline{s} - \overline{s}^3) & \text{(if } 0.15 < s < 0.5, \ \overline{s} = \frac{40s - 13}{7}), \\ q_1 & \text{(if } s \ge 0.5). \end{cases}$$

Let $m \geq$ 1. Consider the $m \times m - dimensional problem F(x) = 0$

(8.5.5a)
$$\phi_{i,j}(x) = 0$$

where

$$(8.5.5b) \qquad \phi_{i,j}(\mathbf{x}) = Q_{i,j}^{C}(\mathbf{x})\xi_{i,j} - Q_{i,j}^{N}(\mathbf{x})\xi_{i,j+1} - Q_{i,j}^{W}(\mathbf{x})\xi_{i-1,j} - Q_{i,j}^{S}(\mathbf{x})\xi_{i,j-1} - Q_{i,j}^{E}(\mathbf{x})\xi_{i+1,j} + \Delta^{2}c$$

$$(i,j = 1, 2, \dots, m)$$

Here, $\xi_{i,j}$ and $\phi_{i,j}(x)$ are the ((i-1)m+j)-th components of respectively x and F(x). Moreover,

(8.5.5c) $\Delta = \frac{1}{m+1}$,

 $(8.5.5d) \qquad \xi_{\mathbf{k},\boldsymbol{\ell}} = 0 \quad (\text{if } (\mathbf{k}\Delta,\boldsymbol{\ell}\Delta) \in \partial\Theta \ (\mathbf{k},\boldsymbol{\ell} = 0,1,\ldots,m+1)).$

Further,

$$\begin{aligned} Q_{i,j}^{N}(\mathbf{x}) &= \frac{1}{2} \{ q(\delta_{v}(i,j+1)^{2} + \delta_{w}(i,j)^{2}) + q(\delta_{v}(i-1,j)^{2} + \delta_{w}(i,j)^{2}) \}, \\ Q_{i,j}^{W}(\mathbf{x}) &= \frac{1}{2} \{ q(\delta_{v}(i-1,j)^{2} + \delta_{w}(i,j)^{2}) + q(\delta_{v}(i-1,j)^{2} + \delta_{w}(i-1,j-1)^{2}) \}, \\ Q_{i,j}^{S}(\mathbf{x}) &= \frac{1}{2} \{ q(\delta_{v}(i-1,j-1)^{2} + \delta_{w}(i,j-1)^{2}) \\ &+ q(\delta_{v}(i,j)^{2} + \delta_{w}(i,j-1)^{2}) \}, \end{aligned}$$

(8.5.5e)

$$Q_{i,j}^{E}(x) = \frac{1}{2} \{q(\delta_{v}(i,j)^{2} + \delta_{w}(i,j-1)^{2}) + q(\delta_{v}(i,j)^{2} + \delta_{w}(i+1,j)^{2})\},$$

$$Q_{i,j}^{C}(x) = Q_{i,j}^{N}(x) + Q_{i,j}^{W}(x) + Q_{i,j}^{S}(x) + Q_{i,j}^{E}(x) \quad (i,j = 1,2,...,m)$$

and

(8.5.5f)
$$\delta_{v}(k,\ell) = \frac{1}{\Delta} (\xi_{k+1}, \ell - \xi_{k}, \ell), \quad \delta_{w}(\ell,k) = \frac{1}{\Delta} (\xi_{\ell,k+1} - \xi_{\ell,k})$$

 $(k = 0, 1, \dots, m; \ell = 0, 1, \dots, m+1),$

The starting point is $u_0 = 0$. For the problems we solved, (8.1.2) (or, equivalently, (8.1.8)) has a unique continuous solution. Setting $x^* = U(1)$, we define D(F) in a similar way as in Problem 2 (cf. (7.2.6)). Although F $\notin F_1$ (F need not be twice differentiable on D(F) if $q_0 \neq q_1$) we tested PCC algorithms on (8.5.5). The computations were performed for m = 5 and 11, c = 15 and $q_0 = 1$; for q_1 we took $q_1 = 10$, 25 and 50.

In the examples to be given, the control parameters $\gamma_1,~\gamma_2,~\gamma_3$ and γ_4 satisfy (8.5.2). We chose

(8.5.6)
$$h_{\text{start}} = 7.10^{-2}$$
, $h_{\text{min}} = 2.10^{-2}$, $h_{\text{max}} = 5.10^{-1}$ (cf. p. 133).

Furthermore, following Rheinboldt, we let $h_1 = h_{start}$.

In Table 8.5.4 results are given where Newton's method is used as local method.

TABLE 8.5.4.

PCC algorithm. Local method: M_0 . Problem 3 (c = 15, m = 5).

Γ	d ⁰	9 ₁	predictor	total number	total number of local steps	total amount of work
	Ŭ	-		OI L-SCEPS		
T	1	10	TI	11	53	53
	T	10	k,p		27	37
	1	10	Ū,	8	57	
	-		~ ^{ĸ,p}	7	28	36
	1	10	UL	/	20	
+			<u>N</u>		72	72
	1	50	U, U,	14	12	
	-		_ ^ĸ ' ^p	11	56	56
	1	50	U U K D	11		42
			~~ [*] ′ ^P	8	34	43
	1	50	ll ^U k	Ŭ		
		1				

We also used the combination of the mixed method and Newton's method as local method. Its control parameters were chosen as follows:

(8.5.7)
$$\gamma_6 = \frac{1}{4}$$
 (cf. (8.4.18)), $\gamma_7 = \frac{1}{2}$ (cf. (8.4.20)).

For γ_5 (cf. (8.4.16b)) we took $\gamma_5 = 0$ and $\gamma_5 = 1$.

TABLE 8.5.5.

PCC algorithm.

Predictor: \tilde{U}_{k} . Local method: \bar{M}_{0} ($\tau = 10^{-4}$, cf. (8.4.21), (8.4.15b)). Problem 3 (c = 15, m = 5).

total number of						total amount	
	Problem		olem	of t-steps	Msteps	Msteps	of work
	^γ 5	Ч ₀	⁴ 1	01 0 1 1	0	0,0	5.0
	0	1	10	8	16	14	53
	0			6	25	1	34
	1	1	10	6	23		50
	0	1	50	8	18	16	59
	U	1			29	1	38
	1	1	50	6	25		

As an example, we specify the performance of a PCC algorithm, using ${ar{ extsf{M}}}_{ extsf{O}}$ as local method.

TABLE 8.5.6.

PCC algorithm.

Predictor: \tilde{U}_{k} . Local method: \bar{M}_{0} ($\gamma_{5} = 1, \tau = 10^{-4}, \text{ cf. (8.4.21), (8.4.15b)}$).

Problem 3 (c = 15, $q_0 = 1$, $q_1 = 50$, m = 5).

		estimate r	number of	
		the radius	of conv. of	local steps
k	t _k	M ₀ M _{0,0}		
1	0.070	-	_	1
2	0.140	-	-	7
3	0.279	0.16510-1	0.54410-1	7*
4	0.366	0.102_{10}^{-1}	0.21710-1	5
5	0.588	0.53810-2	0.13310-1	6
6	1	0.39410-2	0.14210-1	4

* The local process included one $\widetilde{M}_{0,0}^{-\text{step}}$.

We also tested PCC algorithms on Problem 3 where $q_0 = 1$, $q_1 = 25$ and m = 11. This case was also considered in [RHEINBOLDT, 1975]. We present results for the following two PCC algorithms:

PCCI, which uses \tilde{U}_k as predictor and \bar{M}_0 (with control parameters $\gamma_5 = 1$, $\gamma_6 = \frac{1}{4}$ and $\gamma_7 = \frac{1}{2}$) as local method.

PCCII, which uses $U_{k,p}$ as predictor and Newton's method as local method. In both algorithms the control parameters γ_1 , γ_2 , γ_3 and γ_4 satisfy (8.5.2), and h_{start} , h_{min} and h_{max} satisfy (8.5.6). Further, $h_1 = h_{\text{start}}$.

$\frac{\text{TABLE 8.5.7}}{\text{PCC algorithm.}}$ Problem 3 (c = 15, q₀ = 1, q₁ = 25, m = 11).

Algorithm	total number of t-steps	total nu M _o -steps	mber of M _{0.0} -steps	total amount of work
*	19	82	-	82
RHEINBOLDT	10	47	1	60
PCCI	10	4/	۱ ⁻	I
PCCII		FAILUI	RE	

* The numbers have been obtained from [RHEINBOLDT, 1975].

** The 5th local process broke off at $t_5 \approx 0.17$ while $t_5 - t_4 = h_{min}$.

<u>CONCLUSION</u>. From tables 8.5.1 and 8.5.4 it appears that the predictors $\overline{U}_{k,p}$ (see (8.2.4)) and \widetilde{U}_{k} (see (8.2.6b)) allow greater t-steps than the predictor $U_{k,p}$ (see (8.2.1)). Furthermore, the total amount of work required to solve the problem is significantly less when one of the first two predictors is used rather than the last one.

The predictor \tilde{U}_k requires the least t-steps to solve the problem. As an evaluation of $\tilde{U}_k(t)$ is more expensive than an evaluation of $\tilde{U}_{k,p}(t)$, algorithms using $\tilde{U}_{k,p}$ or \tilde{U}_k require about the same amount of work.

The algorithms using the method \overline{M}_0 in which the mixed method with $\gamma_5 > 0$ is used (see (8.4.21) and (8.4.16b)) required less work than the ones using Newton's method only as local method.

With regard to the total amount of work, from Table 8.5.5 it appears that the method \overline{M}_0 in which the mixed method with $\gamma_5 = 0$ is used is inferior to the method \overline{M}_0 in which the mixed method with $\gamma_5 = 1$ is used. (We recall that the mixed method with $\gamma_5 = 0$ is equivalent to the method $\widetilde{M}_{0,0}$ (cf. (7.1.2)).)

Obviously, more numerical experiments are required in order to assess this combination of the mixed method and Newton's method.

We note that, although the algorithm in [RHEINBOLDT, 1975] and the algorithm PCCII both use the predictor $U_{k,p}$ and Newton's method, the latter

algorithm PCCII both use the predictor k , p failed to solve the problem with which Table 8.5.7 is concerned. This may be due to a better choice of control parameters, which are unknown to us, in

the former. It may also be due to the different ways in which the radii of convergence of the local processes are estimated.

We finally notice that the differential equation (8.1.8) is of the type (2.6.11) where H = Q and g = 0. In view of the computational results of chapter 7 it also seems worth while to investigate PCC algorithms that use the differential equation (2.6.11) with H = Q and $g \neq 0$.

In conclusion, PCC algorithms appear to be very reliable for solving the problem F(x) = 0. Among the algorithms tested, the ones in which Davidenko's method is used, require the least amount of work. With suitably chosen control parameters, the combination of the mixed method and Newton's method appears to require less work as local method than Newton's method alone. It is clear, however, that many questions remain open for future research.

REFERENCES

- AVILA, J.H. JR., 1974, The feasibility of continuation methods for nonlinear equations, SIAM J. Num. Anal. 11, 102-122.
- BERGER, M.S., 1977, "Nonlinearity and Functional Analysis", Academic Press, New York.
- BITTNER, L., 1967, Einige kontinuierliche Analogien von Iterationsverfahren, in: "Funktionalanalysis, Approximationstheorie, Numerische Mathematik", ISNM 7, pp. 114-135, Birkhauser, Basel.
- BOGGS, P.T., 1971, The solution of nonlinear systems of equations by A-Stable integration techniques, SIAM J. Num. Anal. 8, 767-785.
- BOGGS, P.T. & J.E. DENNIS JR., 1974, A continuous analogue analysis of nonlinear iterative methods, Cornell University, Dept. of Computer Science, Techn. Rep. TR 74-200.
- BOSARGE, W.E., 1971, Iterative continuation and the solution of nonlinear two-point boundary-value problems, Num. Math. 17, 268-283.
- BROWN, A.L. & A. PAGE, 1970, "Elements of Functional Analysis", Van Nostrand Reinhold Company, London.
- BROYDEN, C., 1969, A new method of solving nonlinear simultaneous equations, Comp. J. 12, 94-99.
- BUS, J.C.P., 1975, A comparitive study of programs for solving nonlinear equations, Mathematisch Centrum, Amsterdam, Rep. NW 25-75.
- BYRNE, G.D. & A.C. HINDMARSH, 1975, A polyalgorithm for the numerical solution of ordinary differential equations, A.C.M. Trans. on Math. Software 1, 71-96.
- CARNAHAN, B., H.A. LUTHER & J.O. WILKES, 1969, "Applied Numerical Methods", Wiley, New York.
- COURANT, R., 1961, "Vorlesungen über Differential- und Integralrechnung", Erster Band; third (improved) print, Springer, Berlin.
- DAHLQUIST, G., 1963, A special stability problem for linear multistep methods, BIT 3, 27-43.

- DAVIDENKO, D.F., 1953, On a new method of numerically integrating a system of nonlinear equations, (Russian), Dokl. Akad. Nauk. SSSR 88, 601-604.
- DAVIDENKO, D.F., 1965a, An application of the method of variation of parameters to the construction of iterative formulas of higher accuracy for the determination of the elements of the inverse matrix, Dokl. Akad. Nauk. SSSR 162, 743-746; Soviet Math. Dokl. 6, 738-742.
- DAVIDENKO, D.F., 1965b, An application of the method of variation of paramaters to the construction of iterative formulas of increased accuracy for numerical solutions of nonlinear integral equations, Dokl. Akad. Nauk. SSSR 162, 499-502; Soviet Math. Dokl. 6, 702-706.
- DAVIDENKO, D.F., 1966, On the construction of iterative processes of increased accuracy by the method of variation of a parameter, (Russian), Abstracts of Session 14, p. 31, International Congress of Mathematicians, Moscow, 1966.
- DAVIDENKO, D.F., 1975, An iterative method of parameter variation for inverting linear operators, USSR Comp. Math. Phys. 15(1), 27-43.
- DEIST, F. & L. SEFOR, 1967, Solution of systems of nonlinear equations by parameter variation, Comp. J. 10, 78-82.
- DEUFLHARD, P., 1974, A modified Newton method for the solution of illconditioned systems of nonlinear equations with application to multiple shooting, Num. Math. 22, 289-315.
- DEUFLHARD, P., 1976, A stepsize control for continuation methods with special application to multiple shooting techniques, Technische Universität München, Institut für Mathematik, Techn. Rep. TUM-MATH 7627.
- DEUFLHARD, P., H.J. PESCH & P. RENTROP, 1976, A modified continuation method for the numerical solution of nonlinear two-point boundary value problems by shooting techniques, Num. Math. 26, 327-343.
- DI LENA, G. & D. TRIGIANTE, 1976, Metodo di Euler e ricerca delle radici di una equazione, Calcollo 13, 377-396.

FEILMEIER, M., 1972, Numerische Aspekte bei der Einbettung nichtlineare Probleme, Computing 9, 355-364.

- FICKEN, F., 1951, The continuation method for functional equations, Comm. Pure Appl. Math. 4, 435-456.
- GAVURIN, M., 1958, Nonlinear functional equations and continuous analogues of iterative methods, (Russian), Izv. Vysš. Učebn. Zaved. Matematica 6, 18-31.
- GEAR, C.W., 1971, "Numerical Initial Value Problems in Ordinary Differential Equations", Prentice-Hall, Englewood Cliffs, New Jersey.
- GRAGG, W.B. & R.A. TAPIA, 1974, Optimal error bounds for the Newton-Kantorovich theorem, SIAM J. Num. Anal. 11, 10-13.
- HOUWEN, P.J. VAN DER, 1977, "Construction of integration formulas for initial value problems", North-Holland, Amsterdam.
- KANTOROWITSCH, L.W. & G.P. AKILOW, 1964, "Funktional Analysis in Normierten Raümen", Akademie-Verlag, Berlin.
- KELLER, H.B., 1968, "Numerical Methods for Two-Point Boundary-Value Problems", Blaisdell, London.
- KITCHEN, J.W., 1966, Concerning the convergence of iterates to fixed points, Studia Math. 27, 247-249.
- KIZNER, W., 1964, A numerical method for finding solutions of nonlinear equations, SIAM J. Appl. Math. 12, 424-428.
- KLEINMICHEL, H., 1968, Stetige Analoga und Iterationsverfahren für nichtlineare Gleichungen in Banachraüme, Math. Nachr. 37, 313-343.
- KOGAN, T.I., 1967, An iteration process for functional equations, (Russian), Sibirsk. Mat. Z. 8, 958-960.
- KUBIČEK, M., 1976, Dependence of solution of nonlinear systems on a parameter, A.C.M. Trans. on Math. Software 2, 98-107.
- LAASONEN, P., 1970, An imbedding method of iteration with global convergence, Computing 5, 253-258.
- LAHAYE, E., 1934, Une méthode de résolution d'une catégorie d'équations transcendantes, C.R. Acad. Sci. Paris 198, 1840-1842.

- LAHAYE, E., 1935, Sur la représentation des racines systèmes d'équations transcendantes, Deuxième Congrès National des Sciences 1, 141-146.
- LAMBERT, J.D., 1973, "Computational Methods in Ordinary Differential Equations", Wiley, London.
- LEDER, D., 1974, Automatische Schrittweitensteuerung bei global konvergenten Einbettungsmethoden, ZAMM 54, 319-324.
- MENZEL, R. & H. SCHWETLICK, 1976, Über einen Ordnungsbegriff bei Einbettungsalgorithmen zur Lösung nichtlinearer Gleichungen, Computing 16, 187-199.
- MEYER, G.H., 1968, On solving nonlinear equations with a one-parameter operator imbedding, SIAM J. Num. Anal. 5, 739-752.
- MURRAY, J.D., 1968, A simple method for obtaining approximate solutions for a class of diffusion-kinetics enzyme problems, Math. Biosciences 2, 379-411.
- ORTEGA, J.M. & W.C. RHEINBOLDT, 1970, "Iterative Solution of Nonlinear Equations in Several Variables", Academic Press, New York.
- OSTROWSKI, A.M., 1960, "Solution of Equations and Systems of Equations", Academic Press, New York; third edition 1973.
- PETRY, W., 1971, Eine Verallgemeinerung des Newtonsche Iterationsverfahren, Computing 7, 25-45.
- POMENTALE, T., 1974, Homotopy iterative methods for polynomial equations, J. Inst. Math. Applic. 13, 201-213.
- RALL, L.B., 1968, Davidenko's method for the solution of nonlinear operator equations, University of Wisconsin, Mathematics Research Center, MRC Techn. Summary Rep. 948.
- RALL, L.B., 1969, "Computational Solution of Nonlinear Operator Equations", Wiley, New York.
- RHEINBOLDT, W.C., 1975, An adaptive continuation process for solving systems of nonlinear equations, University of Maryland, Computer Science Techn. Rep., TR - 393.
- RHEINBOLDT, W.C., 1976, Numerical continuation methods for finite element applications, University of Maryland, Computer Science Techn. Rep., TR - 454.

RIBARIČ, M. & M. SELIŠKAR, 1974, An optimization of stepsize in the continuation method, Mat. Balkanica 4, 517-521.

- ROSENBLOOM, P., 1956, The method of steepest descent, Sixth Symp. Appl. Math., pp. 126-176, Amer. Math. Soc., Providence, Rhode Island.
- RUDIN, W., 1953, "Principles of Mathematical Analysis", McGraw-Hill, New York.
- RUDIN, W., 1973, "Functional Analysis", McGraw-Hill, New York.
- SCHRÖDER, E., 1870, Über unendlich viele Algorithmen zur Auflosung der Gleichungen, Math. Ann. 2, 317-365.
- SCHWETLICK, H., 1975, Ein neues Prinzip zur Konstruktion implementierbarer, global konvergenter Einbettungsverfahren, Beitr. Num. Math. 4, 215-228.
- SCHWETLICK, H., 1976, Ein neues Prinzip zur Konstruktion implementierbarer, global konvergenter Einbettungsalgorithmen (Testbeispiele), Beitr. Num. Math. 5, 201-206.
- SPIJKER, M.N., 1972, Equivalence theorems for nonlinear finite-difference methods, in: "Numerisch Lösung nichtlinearer partieller Differential- und Integrodifferentialgleichungen", Lecture Notes 267, pp. 233-264, Springer, Berlin.
- STETTER, H.J., 1973, "Analysis of Discretization Methods for Ordinary Differential Equations", Springer, Berlin.
- STUMMEL, F. & K. HAINER, 1971, "Praktische Mathematik", Teubner, Stuttgart.
- TRAUB, J.F., 1964, "Iterative Methods for the Solution of Equations", Pentrice-Hall, Englewood Cliffs, New Jersey.
- UEBERHUBER, C.W., 1976, Optimierung mittels kontinuierlicher Quasi-Newton-Verfahren und deren Diskretisierungen, Technische Universität Wien, Institut für Numerische Mathematik, Bericht nr. 19/76.
- WACKER, H., 1971, Eine Lösungsmethode zur Behandlung nichtlinearer Randwertprobleme, in: "Iterationsverfahren, Numerische Mathematik, Approxmationstheorie", ISNM 15, pp. 245-257, Birkhauser, Basel.
- WACKER, H., 1974, Übergangsmöglichkeiten zwischen verschiedene Iterationsverfahren, ZAMM 54, T236-T237.

- WACKER, H., 1977a, Globalisierung und Aufwandminimierung für ein Iterationsverfahren von Kivistik, ZAMM 57, T306-T308.
- WACKER, H., 1977b, Minimierung des Rechenaufwands bei Globalisierungen spezieller Iterationsverfahren vom Typ Minimales Residuum, Computing 18, 209-224.

WASSERSTROM, E., 1973, Numerical solution by the continuation method, SIAM Review 15, 89-119.

SUMMARY

In this monograph we are concerned with the numerical solution of the equation

$$(*)$$
 F(x) = 0.

Here F is a nonlinear operator from D into E, where E is a (real) Hilbert space and D a subset of E. Let x^* be the (unknown) solution of (*).

A well-known iterative method for approximating x^* is Newton's method. However, if the starting point $x_0 \in D$ is remote from x^* , then in general x^* cannot be approximated by this method. In many such cases, so-called imbedding methods succeed in approximating x^* numerically. We restrict our attention to the imbedding

$$H(t,x) = (1-t)K(x,x_0) + tF(x)$$
 ($0 \le t \le 1, x \in D$)

where K is an operator from D×D into E for which K(x,x) = 0 (for all $x \in D$). Let X be the solution of the initial value problem

$$\dot{\mathbf{X}}(t) = -\left[\frac{\partial}{\partial \mathbf{x}} H(t, \mathbf{X}(t))\right]^{-1} \frac{\partial}{\partial t} H(t, \mathbf{X}(t)) \qquad (0 \le t \le 1),$$

$$\begin{pmatrix} * \\ * \end{pmatrix}$$

$$\mathbf{X}(0) = \mathbf{x}_{0}.$$

With certain restrictions on K and F it follows that x = X(t) is the solution of

$$(*^{*}*)$$
 H(t,x) = 0 (t $\in [0,1]$),

so that $X(1) = x^*$.

In this thesis iterative methods are investigated that are based on the successive numerical integration of $\binom{*}{*}$ with changing initial value x_0 . The numerical integration is performed by means of (generalized) Runge-Kutta methods. Hence these iterative methods depend on K and the Runge-Kutta method used.

The investigations are performed using the concept "radius of convergence of an iterative method" (cf. section 2.4).

In chapter 5 the restrictions on K and the Runge-Kutta method which are sufficient for the iterative process to have a positive radius of convergence are given.

In chapter 6 the radii of convergence of some iterative methods of this type are determined. Iterative methods are presented that have a greater radius of convergence than Newton's method.

In chapter 7 numerical results are presented. In particular, an iterative method based on a generalized Runge-Kutta method related to the Backward Euler method appears to be very reliable when x_0 is a remote from x^* .

In chapter 8 algorithms are described that solve successively (***) for t = t_1, t_2, \ldots, t_N (with 0 = $t_0 < t_1 < \ldots < t_N = 1$) using an iterative method. In some of these algorithms the differential equation $\binom{*}{*}$ is also used. The numbers t_1, t_2, \ldots, t_N are determined during the process itself. Some numerical results are presented. The algorithms appear to be very reliable. The algorithms in which differential equation $\binom{*}{*}$ is used, appear to require the least amount of work in order to solve (*).

OTHER TITLES IN THE SERIES MATHEMATICAL CENTRE TRACTS

A leaflet containing an order-form and abstracts of all publications mentioned below is available at the Mathematisch Centrum, Tweede Boerhaavestraat 49, Amsterdam-1005, The Netherlands. Orders should be sent to the same address.

MCT	1	T. VAN DER WALT, Fixed and almost fixed points, 1963. ISBN 90 6196 002 9.
мст	2	A.R. BLOEMENA, Sampling from a graph, 1964. ISBN 90 6196 003 7.
MCT	3	G. DE LEVE, Generalized Markovian decision processes, part I: Model and method, 1964. ISBN 90 6196 004 5.
MCT	4	G. DE LEVE, Generalized Markovian decision processes, part II: Pro- babilistic background, 1964. ISBN 90 6196 005 3.
MCT	5	G. DE LEVE, H.C. TIJMS & P.J. WEEDA, Generalized Markovian decision processes, Applications, 1970. ISBN 90 6196 051 7.
мст	6	M A MAURICE, Compact ordered spaces, 1964. ISBN 90 6196 006 1.
MCT	7	W.R. VAN ZWET, Convex transformations of random variables, 1964. ISBN 90 6196 007 X.
MCT	8	J.A. ZONNEVELD, Automatic numerical integration, 1964. ISBN 90 6196 008 8.
NOT	0	P.C. BAAVEN, Universal morphisms, 1964. ISBN 90 6196 009 6.
MCI	• 1C	E.M. DE JAGER, Applications of distributions in mathematical physics, 1964. ISBN 90 6196 010 X.
МСЛ	11	A.B. PAALMAN-DE MIRANDA, Topological semigroups, 1964. ISBN 90 6196 011 8.
MCT	c 12	2 J.A.TH.M. VAN BERCKEL, H. BRANDT CORSTIUS, R.J. MOKKEN & A. VAN WIJNGAARDEN, Formal properties of newspaper Dutch, 1965. ISBN 90 6196 013 4.
MC	г 1	3 H.A. LAUWERIER, Asymptotic expansions, 1966, out of print; replaced by MCT 54 and 67.
MC	т 1	4 H.A. LAUWERIER, Calculus of variations in mathematical physics, 1966. ISBN 90 6196 020 7.
ма	rr 1	5 R. DOORNBOS, Slippage tests, 1966. ISBN 90 6196 021 5.
MC	т 1 т 1	6 J.W. DE BAKKER, Formal definition of programming languages with an application to the definition of ALGOL 60, 1967. ISBN 90 6196 022 3.
М	ст 1	17 R.P. VAN DE RIET, Formula manipulation in ALGOL 60, part 1, 1968. ISBN 90 6196 025 8.
M	T	18 R.P. VAN DE RIET, Formula manipulation in ALGOL 60, part 2, 1968. ISBN 90 6196 038 X.
M	ст	19 J. VAN DER SLOT, Some properties related to compactness, 1968. ISBN 90 6196 026 6.
M	СТ	20 P.J. VAN DER HOUWEN, Finite difference methods for solving partial differential equations, 1968. ISBN 90 6196 027 4.

- MCT 21 E. WATTEL, The compactness operator in set theory and topology, 1968. ISBN 90 6196 028 2.
- MCT 22 T.J. DEKKER, ALGOL 60 procedures in numerical algebra, part 1, 1968. ISBN 90 6196 029 0.
- MCT 23 T.J. DEKKER & W. HOFFMANN, ALGOL 60 procedures in numerical algebra, part 2, 1968. ISBN 90 6196 030 4.
- MCT 24 J.W. DE BAKKER, Recursive procedures, 1971. ISBN 90 6196 060 6.
- MCT 25 E.R. PAERL, Representations of the Lorentz group and projective geometry, 1969. ISBN 90 6196 039 8.
- MCT 26 EUROPEAN MEETING 1968, Selected statistical papers, part I, 1968. ISBN 90 6196 031 2.
- MCT 27 EUROPEAN MEETING 1968, Selected statistical papers, part II, 1969. ISBN 90 6196 040 1.
- MCT 28 J. OOSTERHOFF, Combination of one-sided statistical tests, 1969. ISBN 90 6196 041 X.
- MCT 29 J. VERHOEFF, Error detecting decimal codes, 1969. ISBN 90 6196 042 8.
- MCT 30 H. BRANDT CORSTIUS, Excercises in computational linguistics, 1970. ISBN 90 6196 052 5.
- MCT 31 W. MOLENAAR, Approximations to the Poisson, binomial and hypergeometric distribution functions, 1970. ISBN 90 6196 053 3.
- MCT 32 L. DE HAAN, On regular variation and its application to the weak convergence of sample extremes, 1970. ISBN 90 6196 054 1.
- MCT 33 F.W. STEUTEL, Preservation of infinite divisibility under mixing and related topics, 1970. ISBN 90 6196 061 4.
- MCT 34 I. JUHASZ, A. VERBEEK & N.S. KROONENBERG, Cardinal functions in topology, 1971. ISBN 90 6196 062 2.
- MCT 35 M.H. VAN EMDEN, An analysis of complexity, 1971. ISBN 90 6196 063 0.
- MCT 36 J. GRASMAN, On the birth of boundary layers, 1971. ISBN 9061960649.
- MCT 37 J.W. DE BAKKER, G.A. BLAAUW, A.J.W. DUIJVESTIJN, E.W. DIJKSTRA, P.J. VAN DER HOUWEN, G.A.M. KAMSTEEG-KEMPER, F.E.J. KRUSEMAN ARETZ, W.L. VAN DER POEL, J.P. SCHAAP-KRUSEMAN, M.V. WILKES & G. ZOUTENDIJK, MC-25 Informatica Symposium, 1971. ISBN 90 6196 065 7.
- MCT 38 W.A. VERLOREN VAN THEMAAT, Automatic analysis of Dutch compound words, 1971. ISBN 90 6196 073 8.
- MCT 39 H. BAVINCK, Jacobi series and approximation, 1972. ISBN 90 6196 074 6.
- MCT 40 H.C. TIJMS, Analysis of (s,S) inventory models, 1972. ISBN 90 6196 075 4.
- MCT 41 A. VERBEEK, Superextensions of topological spaces, 1972. ISBN 90 6196 076 2.
- MCT 42 W. VERVAAT, Success epochs in Bernoulli trials (with applications in number theory), 1972. ISBN 90 6196 077 0.
- MCT 43 F.H. RUYMGAART, Asymptotic theory of rank tests for independence, 1973. ISBN 90 6196 081 9.
- MCT 44 H. BART, Meromorphic operator valued functions, 1973. ISBN 9061960827.

MCT 45 A.A. BALKEMA, Monotone transformations and limit laws, 1973. ISBN 90 6196 083 5.

- MCT 46 R.P. VAN DE RIET, ABC ALGOL, A portable language for formula manipulation systems, part 1: The language, 1973. ISBN 90 6196 0843.
- MCT 47 R.P. VAN DE RIET, ABC ALGOL, A portable language for formula manipulation systems, part 2: The compiler, 1973. ISBN 90 6196 0851.
- F.E.J. KRUSEMAN ARETZ, P.J.W. TEN HAGEN & H.L. OUDSHOORN, An ALGOL 60 compiler in ALGOL 60, Text of the MC-compiler for the мст 48 EL-X8, 1973. ISBN 90 6196 086 X.
- MCT 49 H. KOK, Connected orderable spaces, 1974. ISBN 90 6196 088 6.
- MCT 50 A. VAN WIJNGAARDEN, B.J. MAILLOUX, J.E.L. PECK, C.H.A. KOSTER,

M. SINTZOFF, C.H. LINDSEY, L.G.L.T. MEERTENS & R.G. FISKER (Eds), Revised report on the algorithmic language ALGOL 68, 1976. ISBN 90 6196 089 4.

- MCT 51 A. HORDIJK, Dynamic programming and Markov potential theory, 1974. ISBN 90 6196 095 9.
- MCT 52 P.C. BAAYEN (ed.), Topological structures, 1974. ISBN 90 6196 096 7.
- MCT 53 M.J. FABER, Metrizability in generalized ordered spaces, 1974. ISBN 90 6196 097 5.
- MCT 54 H.A. LAUWERIER, Asymptotic analysis, part 1, 1974. ISBN 90 6196 098 3.
- MCT 55 M. HALL JR. & J.H. VAN LINT (Eds), Combinatorics, part 1: Theory of designs, finite geometry and coding theory, 1974. ISBN 90 6196 099 1.
- MCT 56 M. HALL JR. & J.H. VAN LINT (Eds), Combinatorics, part 2: graph theory, foundations, partitions and combinatorial geometry, 1974. ISBN 90 6196 100 9.
- M. HALL JR. & J.H. VAN LINT (Eds), Combinatorics, part 3: Combina-torial group theory, 1974. ISBN 90 6196 101 7. MCT 57
- W. ALBERS, Asymptotic expansions and the deficiency concept in staмст 58 tistics, 1975. ISBN 90 6196 102 5.
- J.L. MIJNHEER, Sample path properties of stable processes, 1975. мст 59 ISBN 90 6196 107 6.
- MCT 60 F. GOBEL, Queueing models involving buffers, 1975. ISBN 90 6196 108 4.
- * MCT 61 P. VAN EMDE BOAS, Abstract resource-bound classes, part 1. ISBN 90 6196 109 2.
- * MCT 62 P. VAN EMDE BOAS, Abstract resource-bound classes, part 2. ISBN 90 6196 110 6.
 - MCT 63 J.W. DE BAKKER (ed.), Foundations of computer science, 1975. ISBN 90 6196 111 4.
 - W.J. DE SCHIPPER, Symmetric closed categories, 1975. ISBN 90 6196 MCT 64 112 2.
 - J. DE VRIES, Topological transformation groups 1 A categorical apмст 65 proach, 1975. ISBN 90 6196 113 0.
 - H.G.J. PIJLS, Locally convex algebras in spectral theory and eigen-MCT 66 function expansions, 1976. ISBN 90 6196 114 9.

- * MCT 67 H.A. LAUWERIER, Asymptotic analysis, part 2. ISBN 90 6196 119 X.
 - MCT 68 P.P.N. DE GROEN, Singularly perturbed differential operators of second order, 1976. ISBN 90 6196 120 3.
 - MCT 69 J.K. LENSTRA, Sequencing by enumerative methods, 1977. ISBN 90 6196 125 4.
 - MCT 70 W.P. DE ROEVER JR., Recursive program schemes: semantics and proof theory, 1976. ISBN 90 6196 127 0.
 - MCT 71 J.A.E.E. VAN NUNEN, Contracting Markov decision processes, 1976. ISBN 90 6196 129 7.
 - MCT 72 J.K.M. JANSEN, Simple periodic and nonperiodic Lamé functions and their applications in the theory of conical waveguides,1977. ISBN 90 6196 130 0.
 - MCT 73 D.M.R. LEIVANT, Absoluteness of intuitionistic logic, 1979. ISBN 90 6196 122 x.
 - MCT 74 H.J.J. TE RIELE, A theoretical and computational study of generalized aliquot sequences, 1976. ISBN 90 6196 131 9.
 - MCT 75 A.E. BROUWER, Treelike spaces and related connected topological spaces, 1977. ISBN 90 6196 132 7.
 - MCT 76 M. REM, Associons and the closure statement, 1976. ISBN 90 6196 135 1.
 - MCT 77 W.C.M. KALLENBERG, Asymptotic optimality of likelihood ratio tests in exponential families, 1977 ISBN 90 6196 134 3.
 - MCT 78 E. DE JONGE, A.C.M. VAN ROOIJ, Introduction to Riesz spaces, 1977. ISBN 90 6196 133 5.
 - MCT 79 M.C.A. VAN ZUIJLEN, Empirical distributions and rankstatistics, 1977. ISBN 90 6196 145 9.
 - MCT 80 P.W. HEMKER, A numerical study of stiff two-point boundary problems, 1977. ISBN 90 6196 146 7.
 - MCT 81 K.R. APT & J.W. DE BAKKER (eds), Foundations of computer science II, part I, 1976. ISBN 90 6196 140 8.
 - MCT 82 K.R. APT & J.W. DE BAKKER (eds), Foundations of computer science II, part II, 1976. ISBN 90 6196 141 6.
- * MCT 83 L.S. VAN BENTEM JUTTING, Checking Landau's "Grundlagen" in the AUTOMATH system, ISBN 90 6196 147 5.
 - MCT 84 H.L.L. BUSARD, The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?) books vii-xii, 1977. ISBN 90 6196 148 3.
 - MCT 85 J. VAN MILL, Supercompactness and Wallman spaces, 1977. ISBN 90 6196 151 3.
- MCT 86 S.G. VAN DER MEULEN & M. VELDHORST, Torrix I, 1978. ISBN 90 6196 152 1.
- * MCT 87 S.G. VAN DER MEULEN & M. VELDHORST, Torrix II, ISBN 90 6196 153 x.
 - MCT 88 A. SCHRIJVER, Matroids and linking systems, 1977. ISBN 90 6196 154 8.

MCT 89 J.W. DE ROEVER, Complex Fourier transformation and analytic functionals with unbounded carriers, 1978. ISBN 90 6196 155 6.
* MCT 90 L.P.J. GROENEWEGEN, Characterization of optimal strategies in ay- namic games,
* MCT 91 J.M. GEYSEL, Transcendence in fields of positive characteristic, . ISBN 90 6196 157 2.
* MCT 92 P.J. WEEDA, Finite generalized Markov programming, ISBN 90 6196 158 0.
MCT 93 H.C. TIJMS (ed.) & J. WESSELS (ed.), Markov decision theory, 1977. ISBN 90 6196 160 2.
MCT 94 A. BIJLSMA, Simultaneous approximations in transcendental number theory, 1978. ISBN 90 6196 162 9.
MCT 95 K.M. VAN HEE, Bayesian control of Markov chains, 1978. ISBN 90 6196 163 7.
* MCT 96 P.M.B. VITANYI, Lindenmayer systems: structure, languages, and arowth functions, . ISBN 90 6196 164 5.
* MCT 97 A. FEDERGRUEN, Markovian control problems; functional equations and algorithms,
MCT 98 R. GEEL, Singular perturbations of hyperbolic type, 1978. ISBN 90 6196 166 1
MCT 99 J.K. LENSTRA, A.H.G. RINNOOY KAN & P. VAN EMDE BOAS, Interfaces between computer science and operations research, 1978. ISBN 90 6196 170 X.
MCT 100 P.C. BAAYEN, D. VAN DULST & J. OOSTERHOFF (Eds), Proceedings bicenten- nial congress of the Wiskundig Genootschap, part 1,1979. ISBN 90 6196 168 8.
MCT 101 P.C. BAAYEN, D. VAN DULST & J. OOSTERHOFF (Eds), Proceedings bicenten- nial congress of the Wiskundig Genootschap, part 2,1979. ISBN 90 9196 169 6.
MCT 102 D. VAN DULST, Reflexive and superreflexive Banach spaces, 1978. ISBN 90 6196 171 8.
MCT 103 K. VAN HARN, Classifying infinitely divisible distributions by functional equations, 1978. ISBN 90 6196 172 6.
* MCT 104 J.M. VAN WOUWE, Go-spaces and generalizations of metrizability, ISBN 90 6196 173 4.
* MCT 105 R. HELMERS, Edgeworth expansions for linear combinations of order statistics, . ISBN 90 6196 174 2.

-

•

AN ASTERISK BEFORE THE NUMBER MEANS "TO APPEAR"